Consumers' Saving Behaviour : An Empirical Investigation of Consumers' Attitudes Towards Commercial Banking Services with Special Reference to Savings Accounts in Iraq.

> By Salim Karim Hussain

B.Sc. in Accountancy & Business Administration, Iraq, 1979. Diploma in Bank Management, Iraq, 1982. Diploma in Management Sciences, Wales, 1987. Master in Public Administration, Liverpool, 1988.

A Thesis Submitted to the University of Sheffield, School of Management and Economic Studies for the Fulfilment of the Requirements for the Degree of Doctor of Philosophy.

> University of Sheffield The United Kingdom 1993

To My Mother

.

May your soul rest in peace in my troubled country.

-

Acknowledgements

During my studies at Sheffield School of Management and Economic Studies I was fortunate to receive valuable advice from many people.

My sincere thanks should go to Professor Arthur Meidan for accepting the opportunity of supervising my studies and for his constant encouragement, advice, comments and criticisms. In fact, without his help the completion of this thesis would have been impossible.

I would like to thank Professor Sean Holly, the Director of Research at Sheffield School of Management and Economic Studies, who encouraged me and participated in the process of supervising my studies and its completion.

I would like to thank Dr. F. A. Fretwell-Downing, Chairman of Sheffield School of Management and Economic Studies, for providing all the facilities available at the school to finish my studies.

My thanks should go to Professor Roy Wilkinson for his special help and encouragement. Also, my thanks should go to Mr. Robert Sedgwick for his kind help.

I would like to thank EM. Professor G. Clayton and Dr. C. Easingwood for their constructive comments which helped in improving this research.

In addition, I would like to thank the following people for their kind help: Dr. E. Jacobs; Dr. A. E. Tomes; and Dr. I. Moosa of Sheffield School of Management and Economic Studies. Mr. M. Savas and the late Roger Richards of the Computing Services, Sheffield University. Mrs. B. King, Voluntary worker at the British Red Cross. Mrs. D. Green of the Registration Office, Sheffield University. All the staff at Sheffield School of Management. All the staff at the Main and Crookesmoor Libraries. All the staff at the Computing Services, Sheffield University.

Finally, my thanks should go to my family, my wife and my three children for their sacrifices during the course of my studies.

Ι

Abstract

This thesis is the result of a study of consumers' saving behaviour in Iraq. It is an empirical investigation of consumers' attitudes towards commercial banking services with special reference to savings accounts in Iraq.

A number of research objectives and related hypotheses were stated to be achieved through carrying out this investigation. The main focus of the research objectives was to develop a theoretical model and an empirical method of investigating consumers' saving behaviour (decisions) in Iraq. This investigation was carried out in order to achieve the following benefits: (a) providing better understanding of consumers' banking behaviour in Iraq, as one of the less-advanced countries, where little attention has previously been paid to consumers and their needs, wants and satisfaction; (b) contributing to the literature of bank marketing in the Third World which is very limited; (c) establishing the extent to which models of consumer behaviour are applicable in investigating consumers' banking behaviour in a less-advanced country such as Iraq; (d) providing better understanding of consumers' saving decisions in Iraq, which may help banks to gain more savers and enhance the process of mobilisation of finance, and that will have positive effect on the national economy; (e) relating consumers' saving decisions to their attitudes, socio-economic, demographic and some psychographic financial characteristics; and (f) the findings of the research may encourage banks to provide better services for consumers in Iraq by providing banks with some insights about Iraqi consumers in terms of their wants, needs, desires and opinions.

The theoretical part of the research was devoted to covering the following: (a) a background review of the status of the Iraqi banking system; (b) a review of the literature related to the problem under investigation; (c) a discussion of the theories of consumer behaviour in general and consumer banking behaviour in particular; and (d) a presentation of the research methodologies of research design, data collection, sampling and the statistical methods of data analysis.

The empirical part of the research was designed according to the Faceted Design which helps in defining and selecting the most

II

important and relevant variables to the research problem. Also, the faceted design was of a great help in preparing the research questionnaire for this investigation which was used for the purpose of primary data collection from respondents.

Data was gathered in Iraq by questionnaires distributed to 800 respondents. Two methods of probability random sampling, area and systematic sampling, were used for the purpose of selecting a representative sample of the population.

The collected primary data consisted of attitudinal and categorical variables. In other words, the data was multivariate in type and of ordinal and nominal scales. The variables were dependent and independent in nature. Therefore, three statistical methods of data analysis were used to analyse the primary data. Factor Analysis and Two-Group Discriminant Analysis were used to analyse the attitudinal data and achieve the relevant research objectives, while Multiple Classification Analysis was used to analyse the categorical data and achieve the relevant objectives. In addition, the Kolmogorov-Smirnov Goodness-of-Fit test, the F-test and the T-test were used as statistical tests for the purpose of testing the research hypotheses.

With regard to consumers' saving behaviour (decisions) in Iraq, the findings of the empirical part of the investigation revealed, in brief, the following:

The results of Factor Analysis indicate that there are thirteen attitudinal factors which are perceived by SAVERS in Iraq as of significant importance in influencing their decisions to open savings accounts. These thirteen factors summarise the 60 attitudinal variables related to SAVERS' attitudes towards savings accounts.

The results of Two-Group Discriminant Analysis and T-test confirm that there are dissimilarities between the Iraqi consumers' attitudes towards the idea of saving and having savings accounts, if compared according to their: (1) Saving decisions; (2) Saving habits; (3) Age group; (4) Family size; (5) Income; and (6) Educational level.

The results of the Multiple Classification Analysis suggest that consumers' socio-economic and demographic characteristics have an influence on Iraqi consumers' saving decisions (hold / not hold a

III

bank savings account). In other words, the results of the analysis confirm that the seven consumers' socio-economic and demographic characteristics used in this investigation, taken together, have significant influence on their saving decisions. However, when the relationship between each of the seven consumers' socio-economic and demographic characteristics and consumers' saving decisions are examined separately, the findings confirm that each of them, except sex of respondents, has a significant influence on consumers' saving decisions in Iraq. Thus, the findings confirm that sex of respondent in Iraq had no significant influence on his / her saving decision, while, marital status, age, family size, income, educational level and occupation of respondents were of significant importance in influencing consumers' saving decisions. The main reason for this finding can be attributed to the fact that in the Iraqi society women have a less prominent role in general, and have less financial independence in particular.

Finally, the main contributions of the present study can be summarised as follows: (a) a model for investigating consumer's banking behaviour in Iraq, as one of the underdeveloped countries, was developed, as presented in figures 4.4, 4.5, 4.6 and 5.1; (b) Facet Theory was extended to be used in bank marketing research in general, and in investigating consumer banking behaviour in Iraq, as one of the underdeveloped countries, in particular; (c) this study is the first study of consumer banking behaviour in Iraq and its model can be generalised and modified to be used in investigating consumer banking behaviour in the underdeveloped countries; and (d) in comparison with previous comparative studies this study is more comprehensive in terms of its model, methodology and findings.

IV

Table of Contents

Title	Pag	<u>e No.</u>
List of Tabl	es.	6
List of Figu	ires.	8
Chapter One:	apter One: Introduction.	
	1.1 Preface.	10
	1.2 Statement of the Problem and Its Scope.	12
	1.3 Research Objectives.	13
	1.4 Research Hypotheses.	13
	1.5 Research Plan and Organisation.	15
Chapter Two:	The Historical Developments of the Banking	20
	System in Iraq.	
	2.1 Introduction.	21
	2.2 The Iraqi Banking System Before	28
	Nationalisation: Structure and Problems.	
	2.3 Nationalisation of the Commercial banks:	30
	Reasons and Effects.	
	2.3.1 The Main Motives for Nationalisation	32
	of the Commercial Banks in Iraq.	
	2.3.2 The Effects of Nationalising the	34
	Commercial Banks in Iraq.	
	2.3.3 The Advantages and Disadvantages of	35
	Nationalisation of the Commercial	
	Banks in Iraq.	
	2.4 The Iraqi Banking System in Present	39
	Time: Structure, Functions and Problems.	
	2.4.1 Structure and Functions.	3 9
	2.4.2 Problems of the Nationalised Banking	45

System.

2.5 Conclusions.	50
Chapter Three: Review of Literature Related to Commercial	52
Bank Marketing with Special Reference to th	e
Service of Savings Accounts.	
3.1 Introduction.	53
3.2 Commercial Banking: An Overview.	53
3.3 Marketing of Banking Services.	55
3.3.1 The Concept of Bank Marketing.	55
3.3.2 Development of Bank Marketing.	58
3.3.3 Marketing Research in Banking.	63
3.4 Review of Previous Marketing Research	68
Studies Related to Consumers' Saving	
Behaviour.	
3.5 Conclusions.	92
Chapter Four: Consumer Behaviour.	96
4.1 Introduction.	97
4.2 Models of Consumer Behaviour:	97
The Comprehensive Models.	
4.2.1 The Howard - Sheth Model.	99
4.2.2 The Howard Model.	101
4.2.3 The Engel, Kollat and Blackwell Model.	101
4.2.4 Comparison of the Models.	105
4.3 Factors Affecting Consumer Banking	107
Behaviour.	
4.3.1 The External Factors.	107
4.3.2 The Internal Factors.	109

.

Title	Page No.
4.3.3 The Consumer Buying Process.	114
4.4 Conclusions.	120
Chapter Five: The Faceted Design of the Empirical Study	. 123
5.1 Introduction.	124
5.2 The Faceted Approach for the Research	124
Design.	
5.3 The Mapping Sentence of the Definition	1 28
by Facets of the research Problem.	
5.4 The Context of Consumers' Saving	134
Behaviour in Iraq.	
Facet D1: Consumers' Attitudes Towards Savings	134
Accounts.	
Facet D2: Bank Attributes.	136
Facet D3: Consumers' Financial Psychographic	139
Characteristics.	
5.5 Conclusions.	144
Chapter Six: Data Collection and Sampling Methods.	146
6.1 Introduction.	147
6.2 Data Measurements.	147
6.2.1 Measurements of Respondents Attitude	es. 150
6.2.2 Measurements of Respondents	152
Characteristics.	
6.3 Data Collection.	153
6.3.1 Methods of Data Collection.	154
6.3.2 Development of the Questionnaire.	157
6.4 Sampling.	162
6.4.1 Methods of Sampling.	162

.

Title	Page No.
6.4.2 Sample Size.	170
6.5 Summary.	171
Chapter Seven: Methods of Data Analysis.	173
7.1 Introduction.	174
7.2 Factor Analysis.	176
7.3 Two-Group Discriminant Analysis.	180
7.4 Multiple Classification Analysis.	182
7.5 The Kolmogorov-Smirnov Goodness-of-Fit	184
Test.	
7.6 The F-Test.	185
7.7 The T-Test.	186
7.8 Summary.	187
Chapter Eight: Factor Analysis Findings.	188
8.1 Introduction.	189
8.2 Interpretation of Factor Analysis	190
Findings.	
8.3 Testing Research Hypothesis.	206
8.4 Conclusions.	209
Chapter Nine: Discriminant Analysis and T-Test Findings	. 211
9.1 Introduction.	212
9.2 Interpretation of the Findings.	212
9.2.1 The Findings of Discriminant Analysis	s: 212
Discriminating Between Respondents Accordin	lg
to their Saving Decisions. (Banked with	
Savings Accounts vs. Non-banked).	
9.2.2 The Findings of the T-Test.	237
9.2.2.1 Respondents' Saving Habits.	237

•

TILE		Page No.
	9.2.2.2 Respondents' Age Group.	242
	9.2.2.3 Respondents' Family Size.	244
	9.2.2.4 Respondents' Income Group.	245
	9.2.2.5 Respondents' Educational Level.	246
	9.3 Comparison with Other Studies of	248
	Consumers' Saving Behaviour.	
	9.4 Conclusions.	254
Chapter Ten:	Multiple Classification Analysis Findings	. 256
	10.1 Introduction.	257
	10.2 Multiple Classification Analysis	259
	Findings For:	
	10.2.1 Consumers' Demographic	267
	Characteristics.	
	10.2.2 Consumers' Socio-economic	274
	Characteristics.	
	10.3 Testing Research Hypotheses.	281
	10.4 Conclusions.	286
Chapter Elev	ven: Conclusions.	290
	11.1 Research Conclusions.	291
	11.2 Research Contributions.	300
	11.3 Research Limitations.	306
	11.4 Recommendations for Further Research	ı . 308
List of Refe	erences.	311
Bibliography	ý.	321
Appendix A:	The Research Questionnaire.	337
Appendix B:	The D-Values of the Kolmogorov-Smirnov Tes	st. 355
Appendix C:	The Complete Rotated Factor Matrix.	363

mia 1.

List of Tables

<u>Title of Table</u>	Page No.
2.1 Number of savings accounts at commercial banks	26
in Iraq.	
2.2 Types of banks in Iraq.	41
2.3 Comparison between the Iraqi and the Developed	49
Countries banking systems.	
3.1 Summary of previous research publications.	88
5.1 The context of consumers' saving behaviour in Iraq	. 141
6.1 Rate of response.	169
7.1 Justifications for using Factor Analysis.	178
7.2 Justifications for using Discriminant Analysis.	182
7.3 Justifications for using Multiple Classification	184
Analysis.	
8.1 Factor one: Customers' satisfaction.	194
8.2 Factor two: Advertising.	195
8.3 Factor three: Benefits of savings.	196
8.4 Factor four: Bank's image.	197
8.5 Factor five: Location convenience.	198
8.6 Factor six: Quality of transaction.	199
8.7 Factor seven: Building design.	200
8.8 Factor eight: Customers' benefit.	201
8.9 Factor nine: Reasons for saving.	202
8.10 Factor ten: Influence of relationship.	203
8.11 Factor eleven: Bank loyalty.	204
8.12 Factor twelve: Access convenience.	205
8.13 Factor Thirteen: Waiting time.	206
8.14 D-values of the Kolmogorov-Smirnov test.	208

<u>Title of Table</u>	Page No.
9.1 Number of cases by group (saving decisions).	214
9.2 The canonical discriminant function (saving decisions	5). 215
9.3 Standardized coefficients and group mean values	216
(saving decisions).	
9.4 Summary of dissimilarities between "banked" and	232
"non-banked" respondents.	
9.5 Groups classification results (saving decisions).	234
9.6 The results of t-test for respondents' saving habits	. 239
9.7 The results of t-test for respondents age group.	243
9.8 The results of t-test for respondents family size.	244
9.9 The results of t-test for respondents income.	245
9.10 The results of t-test for respondents educational	247
level.	
10.1 The Multiple Classification Analysis results.	261
10.2 Rate of occurrence of the consumers' profile	265
characteristics.	
10.3 Main groups and characteristics of consumers.	289

.

List of Figures

Title of Figure	Page No.
1.1 The plan of the research.	17
2.1 Development of the Iraqi banking system.	23
2.2 Development of savings accounts in Iraq.	27
2.3 The Iraqi financial system in present time.	40
3.1 Recent developments in bank marketing research.	66
4.1 The Howard - Sheth model.	102
4.2 The Howard model.	103
4.3 The complete EKB model.	104
4.4 Stages of consumers' saving decisions.	115
4.5 A method to investigate consumers' saving decisions.	117
4.6 A model of consumer behaviour in banking.	119
5.1 The mapping sentence for the definition by facets of	130
consumers' saving behaviour in Iraq.	
6.1 The sampling procedures used in this investigation.	167
7.1 Methods of data analysis.	175
9.1 Each and all-groups histogram (saving decisions).	235
10.1 Sex of respondents.	268
10.2 Marital status of respondents.	270
10.3 Age group of respondents.	272
10.4 Family size of respondents.	275
10.5 Income group of respondents.	276
10.6 Educational level of respondents.	279
10.7 Occupations of respondents.	280
11.1 The money circulation between banks and customers.	291

CHAPTER ONE

•

.

INTRODUCTION

.

.

1.1 PREFACE.

1.2 STATEMENT OF THE PROBLEM AND ITS SCOPE.

1.3 RESEARCH OBJECTIVES.

1.4 RESEARCH HYPOTHESES.

1.5 RESEARCH PLAN AND ORGANISATION.

1.1 Preface:

A lot of work has been done to provide an understanding of consumer behaviour in matters related to goods (products), which are tangible. Relatively, less has been done to provide such an understanding for consumer behaviour in matters related to services which are intangible, especially in the developing countries. However, given the importance of the banking sector in any economy, it is not clear if the models of consumer behaviour, which are developed to deal with consumption of products (goods) and other services, are applicable in the case of banking, because banks provide distinct types of services. In addition, it is not clear if the methods of bank marketing, marketing research and models of consumers' behaviour developed in the Western Developed World are applicable in a developing country like Iraq.

Shankleman (1979) pointed-out that in banking, formal marketing only really started in 1968 in Britain and even earlier in the USA. Most of the previous studies dealing with bank marketing, especially those related to consumers' attitudes, were conducted in the United States of America, where a very different banking system exists. Very few studies were found dealing with these topics in countries where a relatively small number of chartered banks dominate the market through a strong network of branches nation-wide such as in Britain, Canada, France, etc.

Earl and Rose (1976) concluded that in commercial banking, only a limited amount of careful research has been directed toward marketing of commercial banking services. Commercial banks are the dominant institutions in the financial system of most countries, for

example, they control about 40 per cent of all financial institutions transactions in the U.S.A.

Relatively fewer researches have been conducted in the less advanced countries, i.e., the Third World countries, where the situation is extremely different. Therefore, the main reasons and motivations for conducting this study can be summarized in the following:

Firstly, there is an urgent demand to provide an understanding of consumers' behaviour and their needs and wants in matters related to banking in Iraq, as one of the less-advanced countries, in order to develop the banking services and satisfy consumers.

Secondly, it is hoped that this study will contribute to the literature of studies on bank marketing, which is very limited in the Third World countries.

Thirdly, the whole purpose of this study is to secure the quality of banking services through effective marketing. That means reaching and satisfying customers' needs and wants in order to survive and provide better services.

Fourthly, one of the overall objectives of this study is to establish the extent to which models of consumer behaviour are applicable in investigating consumers' banking behaviour, especially, in one of the third world countries, i.e., Iraq.

Fifthly, the main focus of this investigation is on one of the most importance banking services, that are provided by commercial (retail) banks. That is, the service of opening savings accounts for customers.

Finally and in addition to the above, this study is an attempt to relate consumers' attitudes, socio-economic, demographic

and psychographic characteristics to their saving behaviour (decisions) in Iraq.

1.2 Statement of the Problem and its Scope:

Understanding consumers' behaviour in the banking sector is a very important factor for any bank in establishing its policies and strategies. Furthermore, consumers' needs and wants are increased because of increasing competition between banks to gain customers, changes in technology, life style and life sophistication.

The foundation of banks was on their deposit taking system. Thus, deposits or savings accounts are the most important source in financing any bank to operate in the market. However, in recenttimes, commercial (retail) banks found that in order to encourage consumers to deposit their money in the bank, they should adopt a new marketing strategy, which should be more consumer-oriented than product-oriented as it used to be. Therefore, many commercial banks started to carry-out marketing research investigations in order to reach the opinions of their customers and discover their needs and wants.

Initially, most market research was based on qualitative methods. More recently, attempts have been made to establish quantified approaches to studies in bank marketing research. Thus, the examination of the subject of consumers' attitudes toward commercial banking services within the context of consumer behaviour in banking has remained limited.

Therefore, this study is an attempt to explore the relevant factors and variables which affect consumers and their attitudes toward commercial banking services with particular reference to savings accounts. The service of savings accounts is chosen because

of its importance for the commercial banking business and its wide usage by consumers. In other words, the main research problem is to define what factors and variables may affect consumers in their attitudes toward commercial banking services, and what are the methods that can be applied by the commercial banks in order to satisfy their customers and increase their share in the market, especially the savings market.

1.3 Research Objectives:

In this study the following research objectives were stated to guide the empirical part of the investigation:

 To determine those universal attitudinal factors which are perceived by the Iraqi savers as important in their decision to open savings accounts at a commercial bank.
To establish the extent to which dissimilarities exist between consumers who are "banked with savings accounts vs. non-banked" in terms of the rate of importance they attach to the variables related to the service of savings accounts provided by commercial banks in Iraq.

3. To examine the relationship between Iraqi consumers' saving decisions (saving behaviour) and their profile characteristics, i.e., socio-economic and demographic characteristics such as: a. sex; b. marital status; c. age; d. family size; e. income; f. educational level; and g. occupation.

1.4 Research Hypotheses:

Churchill (1988) defines hypothesis as being a statement that specifies how two or more measurable variables are related. Also, it

has been defined by Moutinho (1982) as an assumption, which on the basis of some experimental findings, may be rejected or allowed to stand. However, in this study three assumptions (hypotheses) were made which are justified on the following:

a. The literature analysis, as mentioned in chapter three, showed that comparative assumptions were made to investigate consumers of banking services in other environments (countries). Therefore, it was thought that such hypotheses can be applied to investigate consumers of banking services in Iraq.

b. Consumers, in general, behave differently according to the rate of importance they may attach to a number of variables which may influence their behaviour. As it is known, different segments of consumers attach different rates of importance to one or more variables related to their behaviour. For example, consumers who are holders of savings accounts with banks may consider the rate of interest on their savings to be more important than nonholders. In other words, there must be reasons for the difference in the behaviour between holders of savings accounts and non-holders.

c. This study's theoretical model, as presented in figures 4.4, 4.5, 4.6 and 5.1, indicated that there are variables that may be able to explain the difference in consumers' saving behaviour in Iraq.

d. The pilot study carried out before embarking on this investigation indicated that there are differences between different segments of Iraqi consumers in terms of their

wants, needs, desires, attitudes, socio-economic and demographic characteristics.

Therefore, the following research hypotheses were formulated to guide the empirical part of this research.

<u>Hypothesis (A)</u>: There are significant similarities between respondents who are "banked with savings accounts" in terms of their attitudes towards the service of savings accounts provided by commercial banks in Iraq.

<u>Hypothesis (B)</u>: There are significant dissimilarities between respondents who are "banked with savings accounts" and those "non-banked" in terms of their attitudes towards the service of savings accounts provided by commercial banks in Iraq.

<u>Hypothesis (C)</u>: Consumers' profile characteristics, i.e., the seven socio-economic and demographic characteristics taken together or separately, have a significant influence on consumers' saving decisions (hold / not hold a savings account) in Iraq.

1.5 Research Plan and Organisation:

The plan of this investigation was based on achieving the study's chief objective of developing a method of investigating consumers' saving behaviour (decisions) in Iraq. The plan of the research, as shown in figure 1.1, was divided into four main stages, these are: The first stage was devoted to define the research problem, objectives, and hypotheses. The second stage was designed to investigate the following:

(a) the concepts of bank marketing and bank marketingresearch; (b) the relevant previous bank marketing research

studies; (c) the background of the Iraqi banking system, because it is the case study in this investigation; and (d) the relevant models of consumer behaviour.

The third stage dealt with the research methodology of data collection, sampling, design and statistical analyses. The fourth was devoted to the reporting of the empirical findings of the research and their implications. Also, it was used to report how the research plan had participated in achieving the chief research objective mentioned above, both theoretically and empirically. The following is a brief description of the research organisation, chapter by chapter:

The purpose of this first chapter is to briefly highlight the study topic area, identify the research problem and its scope, state the research objectives, and specify the research hypotheses relevant to the objectives.

Chapter two is an attempt to provide wider background to the study in terms of the importance of this topic for the Iraqi banking system, which is the field of the investigation. Also, the chapter was an attempt to describe the historical developments of the Iraqi banking system, and its present structure with reference to its functions and drawbacks.

16



Figure 1.1: The plan of the research.

Chapter three is a review of the relevant literature in the field and aims to acquaint the reader with existing knowledge on the subject, together with a discussion of the present study in relation to the current literature. Chapter four contains a description of consumer buying behaviour models with special reference to their application in banking and consumers' banking behaviour. This approach was adopted because the investigation specifically concentrates on consumers' saving behaviour.

Chapter five describes the research design of the investigation. The research design was based on employing facet design in order to present systematically the universe of the content of research area, underline the relations between the variables used in the investigation, and produce a mapping sentence of the whole universe of context under investigation.

Chapter six discusses in detail, data requirements, questionnaire design, the process of sampling the population of respondents and other methods of data collection used to gather the required data for the research.

Chapter seven describes the research methodology of the analysis. It contains a description of the analytical techniques used in this study which consisted of: Factor Analysis, Discriminant Analysis, and Multiple Classification Analysis (MCA). Also, a description of the statistical tests used to test research hypotheses such as: Kolmogorov-Smirnov One-sample Goodness-of-Fit Test, F-Test, T-Test and other statistical tests used was provided in this chapter.

Chapters eight, nine, and ten were devoted to reporting the research findings for each research objective and its hypothesis(es). Therefore, chapter eight reports the findings of Factor Analysis, which was used to determine those attitudinal factors which are perceived by savers as important in their decision to open savings accounts at a commercial bank. Kolmogorov-Smirnov

One-sample Goodness-of-Fit Test was used for testing the relevant hypotheses to this chapter.

Chapter nine reports the findings of Discriminant Analysis, which was used to achieve the objective of establishing the extent to which differences exist between consumers, who are banked with savings accounts and non-banked, in terms of the difference in the rate of importance they attached to the variables related to the service of savings accounts provided by commercial banks in Iraq. savers and non-savers in terms of their attitudes towards the variables under investigation. Also, it includes the findings of the T-Test regarding establishing the differences between respondents' attitudes according to their: (a) Age; (b) Family size; (c) Income level; (d) Educational level; and (e) Saving habits.

Chapter ten reports the findings of Multiple Classification Analysis (MCA), which was used to achieve the objective of examining the relationship between consumers' saving decisions (saving behaviour) and consumers' profile characteristics (socio-economic and demographic characteristics of consumers) such as: income, qualification, occupation, sex, age, marital status, and family size. The programme of (MCA) provides the basic statistics from which F-test easily calculated to test the hypotheses relevant to this objective.

Finally, chapter eleven was devoted to report the research conclusions, limitations, contributions in terms of theoretical and practical aspects, and recommendations regarding further research in the area of bank marketing.

CHAPTER TWO

THE HISTORICAL DEVELOPMENTS OF THE BANKING SYSTEM IN IRAQ

2.1 INTRODUCTION.

2.2 THE IRAQI BANKING SYSTEM BEFORE NATIONALISATION: STRUCTURE AND PROBLEMS.

2.3 NATIONALISATION OF THE COMMERCIAL BANKS: REASONS AND EFFECTS.

2.3.1 THE MAIN MOTIVES FOR NATIONALISATION OF THE

COMMERCIAL BANKS IN IRAQ.

2.3.2 THE EFFECTS OF NATIONALISING THE

COMMERCIAL

BANKS IN IRAQ.

2.3.3 THE ADVANTAGES AND DISADVANTAGES OF

NATIONALISATION OF THE COMMERCIAL BANKS IN

IRAQ.

2.4 THE IRAQI BANKING SYSTEM IN RECENT-TIMES:

STRUCTURE, FUNCTIONS, AND PROBLEMS.

2.4.1 STRUCTURE AND FUNCTIONS.

2.4.2 PROBLEMS OF THE NATIONALISED BANKING SYSTEM.

2.5 CONCLUSIONS.

2.1 INTRODUCTION:

This chapter will review the historical development and structure of the banking system in Iraq. This review will make special reference to the main problems of a totally state-owned banking system, and how this type of ownership has affected the performance of banks and the satisfaction of customers. Also, this review will help to provide some sort of background for the investigation of consumers' attitudes toward commercial banking services in Iraq, with particular reference to the service of savings accounts.

Providing such a review will help the investigation in later stages, especially in the process of interpreting research findings regarding banking consumers and their banking behaviour in Iraq. Also, it should be mentioned here that some of the contents of this chapter will be descriptive. In fact, the reason for that is to provide the reader with some insights about the state of the Iraqi banking system. Thus, because it was found that information about the Iraqi banking system available to the reader in the English language are very rare, and, as is clear, the empirical part of the present study is about consumers of banking services in Iraq. Therefore, readers should be given sufficient information about the state of the Iraqi banking system in terms of its structure, development, and problems. However, it should be mentioned here also that some of the contents of this chapter are based on previous researches carried out by the researcher [see S. Hussain (1982) and (1988)].

The origin of the commercial banking in Iraq can be traced to 1891 when the Ottoman Bank opened a branch, mainly because Iraq at the time was part of the Ottoman Empire. After that, many other

foreign banks opened branches in Iraq. Development of domestic commercial banking in Iraq, however, has a relatively recent history. In fact, it was not until 1941 that the first national commercial bank (called the Rafidain Bank) was established. The pace of banking activity and bank holdings expanded especially after the 1950s, mainly because of economic developments. However, the most important event which significantly affected the commercial banking system in Iraq was the government's programme of nationalisation of private and foreign commercial banks in 1964.

The nationalisation was followed by a number of mergers between nationalised commercial banks until 1974, when the stateowned Rafidain Bank became the only commercial bank in the country, providing all the commercial banking services.

In 1987, the government acknowledged that having only one commercial bank led to a lack of competition, poor performance, and consumer dissatisfaction within the commercial banking system. This unacceptable situation had led to a decline in the quality of commercial banking services and to the consumer having no choice but to deal with this bank whatever the conditions of the services.

Therefore, the government decided to establish another state-owned commercial bank, called Al-Rasheed Bank, to provide commercial banking services for consumers and to create some sort of competitive environment between the two commercial banks.

Figure 2.1 explains the development of the Iraqi banking system before and after nationalisation in terms of the number of banks. Figure 2.1 confirms that before nationalisation the number of banks had increased from one in 1891 to 17 at the end of 1963. After nationalisation, which occurred in 1964, the number of banks

FIGURE 2.1: DEVELOPMENT OF THE IRAQI BANKING SYSTEM



BANKS

decreased from 17 to only 6 in 1991 (see also figure 2.3, the Iraqi banking system as it was in 1991).

In fact, in Iraq, as is the case in many developing countries, in late 1950s and throughout the 1960s and 1970s the notion of nationalisation dominated the economic strategies of most developing countries. Ramanadham (1984) indicated that large-scale nationalisations have been carried out in countries like Zambia. Tanzania, Somalia, Ethiopia, Sudan, Egypt, Iraq, Syria, Pakistan and Bangladesh on some form of socialist ground. Thus, Iraq, as one of the developing countries, practiced the same concept of nationalisation adopted by many developing countries at that time, because nationalisation for them provided the means to escape from foreign economic domination. Also, nationalisation provided them with spiritual forces for national identification. In most developing countries, such as those mentioned above, the banking system was the first in their list of nationalisation, as was the case in Iraq, where the decision was taken to nationalise banks in Iraq in 1964. Therefore, it can be said that the Iraqi banking system shares the same characteristics of most banking systems of the developing countries in terms of being state-owned and centrally controlled by the central government.

However, throughout the 1980s many developing countries started the process of revising their nationalisation policies and even carrying out some privatisation programmes which are expected to expand and influence most developing countries in the 1990s including Iraq.

In the savings market, most banks in Iraq, accept deposits. Therefore, the competition for savings is high and every bank tries to increase its share in the savings market. In addition, the

National Post Office Enterprise provides an alternative source of savings accounts through a network of Savings Funds. However, most savers prefer to deposit their savings at the commercial banks for the following reasons: (a) Commercial banks have more branches than other deposit-taker institutions (for example, the Rafidain Bank has about 250 branches nationwide); and (b) Commercial banks provide a wider range of other banking services than any other financial institutions including insurance companies.

The focus of this study will be on commercial banking with particular reference to savings. Therefore, this study will attempt to highlight the major factors and variables which have affected consumers' attitudes toward savings with particular reference to the role of commercial banks in Iraq in providing this type of banking services.

In addition, the importance of this study for the commercial banks in Iraq rests on the fact that savings accounts is the banking service most used by the general public in Iraq. This is so for a number of reasons:

1. Savers are a large segment of banking savings services users in Iraq. Savers are the kind of bank customers who wish to keep their surplus of income in a safe and secure place, and gain some return on it. Table 2.1 and figure 2.2 show the development in the size of savings accounts in Iraq in terms of number of accounts holders and balances for a number of years.

2. The service of checking accounts in Iraq is restricted to companies and businessmen. Checking accounts are not available to other segments of consumers because of some limitations on their use introduced by Iraqi Commercial Law, and practiced by the Iraqi commercial banks.

3. Although the savers have money which they are ready to invest (save), in Iraq, most such customers do not like to be engaged in the sophistication of banking services unless they need it. In fact, this is the nature of customers in most less-developed countries, where banking services are seen by people as something difficult to deal with, and they prefer dealing in cash.

Year	Total Savings (I. D.)	Number of Savings Accounts	Average Balance per Account (I. D.)
1970	64,300,000	NA ·	NA
1971	73,700,000	NA	NA
1972	84,000,000	NA	NA
1973	104,300,000	NA	NA
1974	134,800,000	NA	NA
1975	176,700,000	NA	NA
1976	236,700,000	NA	NA
1977	281,200,000	NA	NA
1978	364,600,000	NA	NA
1979	428,851,000	407,053	1,054
1980	522,789,000	436,949	1,196
1981	837,387,000	504,604	1,659
1982	1,024,629,000	507,683	2,018
1983	1,180,225,000	539,947	2,186
1984	1,490,304,000	566,323	2,632
1985	1,760,080,000	559,162	3,148
1986	2,006,318,000	558,905	3,590
1987	2,417,761,000	591,258	4,089
1988	2,975,953,000	653,285	4,555

Table (2.1): Number of Savings Accounts and its total balances at the Commercial Banks in Iraq.

- 1. One Iraqi Dinar (I. D.) equal to about £2.00 (This is the official rate of exchange as it was in 1991).
- 2. Balances are at the end of the financial period (31 December) of each year.

Source: Central Bank of Iraq, Yearly Bulletins, 1971-1989.

Figure 2.2: Development of Savings Accounts in Iraq.





2.2 THE IRAQI BANKING SYSTEM BEFORE NATIONALISATION:

STRUCTURE AND PROBLEMS.

The banking system in Iraq has developed gradually since the establishment of the first commercial bank in 1891. Its structure and functions have been directly affected by the political, economic and social changes which have affected Iraq as a developing country.

At the time of establishment of the first commercial bank in Iraq, and until the end of the First World War, Iraq was a colony of the Ottoman Empire. After the war, Iraq became a British protectorate. Until independence in 1921, political unrest affected all the economic sectors in Iraq. However, the effect on the banking sector was greater than other sectors because, as is known, investors in financial projects such as banks do not want to invest in countries suffering instability. The political situation caused a slump in the economy, which badly affected the desire of the domestic private sector to invest in any economic sector, especially in establishing banks. Meanwhile, the government at that time did not have enough resources to do so.

This situation led Iraq to remain as a less-developed country with no economic centrally determined plans, policies or strategies. All economic activities tended to depend on the decisions of individuals or foreign or domestic companies to invest in Iraq or not.

One important part of any economy is its financial system, especially central and commercial banking, which in Iraq did not develop until the 1960s, and that affected the Iraqi economy deeply.

The late development of commercial banking in Iraq was mirrored by the absence of a central bank until after independence.

According to the Central Bank of Iraq (1972), there was no central banking organisation in Iraq because Iraq had been a British protectorate. The desire to establish a central bank started after the establishment of the Iraqi government in 1921.

However, there were attempts to issue a national currency to be used instead of the Indian Rupee. For example, Article 108 of the Iraqi Constitution in 1924 referred to the need to issue a national currency. In 1930, the government established a Currency Board (Committee) with its main office in London and decided to issue Iraqi currency based on the British Pound.

The National Bank of Iraq was established according to Act 43 of 1947, with capital of 5 million Iraqi Dinar. The duties of the Currency Board (Committee) were then transferred to the new bank. The National Bank of Iraq acted as a central bank for the Iraqi government, and its success played an important role in developing the central banking system in Iraq.

In order to develop central banking in Iraq, the Central Bank of Iraq was established in place of the National Bank of Iraq according to Act 72 of 1956. Its capital increased to 15 million Iraqi Dinar, with reserves of 30 million Iraqi Dinar. The Act of 1956 further defined the main aims and duties of the bank as follows:

a. Issue and manage the currency;

b. Ensure the stability of the national currency;

c. Play an important role in the banking credit status;

d. Act as the bank of the government and the bank of the banks;

e. Organise foreign payments with foreign central banks; and

f. Provide other central banking services for the economy.

However, the role of the central bank of Iraq has been redefined through various legal amendments which will be discussed later in this chapter.

The status of the commercial banking and other financial institutions before nationalisation was characterised as ineffective. Many foreign commercial banks opened branches in Iraq concentrated on financing only commercial after 1891, but activities. At that time, commerce was an active part in the Iraqi economy, mainly because Iraq depended heavily on importing her needs for goods and other services from other countries. Little foreign banking assistance was given to non-commercial sectors such as industry, agriculture and construction.

Furthermore, these foreign commercial banks concentrated their efforts on maximizing their profits and stabilising their liquidity. On the other hand, the government had no financial institution to control and supervise their activities at that time. However, in 1947 when the National Bank of Iraq was established, it was given a limited authority to control foreign and national banks activities.

2.3 Nationalisation of the Commercial Banks:

Reasons and Effects.

The government's attempt to establish a national banking system before nationalisation can be summarised as follows:

The first step was to establish a national central monetary authority responsible for controlling foreign currencies and banking credit in the country, represented by the central bank. That was
achieved by the establishment of the National Bank in 1947, which then changed to become the Central Bank of Iraq in 1956.

The second step was to create some type of re-distribution of banking credit in order to finance the industrial, agricultural, and construction sectors as well as the commercial sector. Commerce had been over-financed by the commercial banks while the other sectors had found difficulty in obtaining sufficient banking credit. In fact, this step was achieved in the 1950s by the establishment of a number of specialist banks to assist other economic sectors such as the Industrial bank, the Agricultural bank, and the Building bank.

Finally, the third step was the attempt to change the ownership of the foreign banks to the public and / or the domestic private sector which was carried out throughout the 1950s and 1960s.

Nevertheless, these attempts failed to achieve the desired shifts in Iraqi banking. Moreover, the fragmented nature of the banking system went against the notion, then fashionable among many less-developed countries, that it was beneficial to give central direction to the economy and to nationalise banking. The Iraqi government took the decision of bank nationalisation in 1964. According to Act 100 in 1964, the ownership of all non-state-owned commercial banks (foreign, Arabic and domestic private) was transferred to the Iraqi government.

The nationalisation of the commercial banks in Iraq required changes in the banking system in terms of structure and functions. For example, item 5 of the same Act provided for the establishment of a Public Corporation for Banks, which became the head office of all the nationalised banks. That was followed by a number of mergers between banks until the banking system developed into its present arrangement (see figure 2.3).

2.3.1 The main motives for the nationalisation of the commercial banks in Iraq:

Khalaf (1977) reports that F. Al-Kaysee, known as the architect of the nationalisation decision, summarised the main motives for the nationalisation of the commercial banks in Iraq as follows:

"It has become clear that the resources of the private commercial banks are not used in a suitable way to satisfy the needs of the economic development process in the country. And, the owners of the foreign capital invest it in the areas which provide profits for them without any care for the country's needs. Furthermore, most of the private commercial banks attempted to gain maximum and fast profits, without caring for other factors in the economy. Such a banking strategy was very risky for the banks and forced some of them out of the market, affecting the financial system in general. In addition, there appeared new economic strata which monopolised the banks, insurance companies and other industries, with good ability to interfere in the national economic and social matters in order to gain maximum profits".

Al-Kaysee (1967) added the following additional reasons for the nationalisation of commercial banks:

1. The commercial banks' negligence of the requirements of economic development process when they invested their money.

2. The domination of the foreign capital over the commercial banks, and its efforts to direct the investments toward very profitable sectors only.

3. The highly competitive situation between the commercial banks had led them to maximise their profits without caring for banking conventions and principles, which created a very risky environment and as a result weakened some of them.

4. The concentration of wealth in the hands of some segments of society which dominated over banks, insurance companies, industrial and agriculture sectors. Moreover, this concentration of wealth produced a very strong social class which attempted to influence political and economic policies and to direct them toward their wishes. This led to unequal distribution of the national income.

Al-Shamaa (1978) mentioned other aspects, which he described as some technical reasons, for the nationalisation of the commercial banks in Iraq. These included the following:

(a) In first place, nationalisation attempted to organise banking facilities and services, and put them under good control and better management system;

(b) It also increased the control of the Central Bank of Iraq over the banking system activities, especially those related to banking credit and foreign currencies;

(c) Nationalisation was also an attempt to avoid the situation whereby private banks concentrated on providing banking facilities for only some segments of private businesses;

(d) Another reason for nationalisation was due to the fact that some of the commercial banks issued false annual accounts showing losses in order to hide their true profits and expropriate them; and

(e) Another important technical motive for nationalisation was the unequal distribution of banking services and the concentration of private banks branches only in main cities, which disadvantaged other parts of the country.

All the above-mentioned reasons played an important role in increasing the demand for nationalisation at that time, and led the government decision in 1964.

2.3.2 The effects of nationalising the commercial banks in Iraq:

Nationalisation of the commercial banks in Iraq affected the national economy as well as the banking system itself. The most important effects have been described by Khalaf (1977) as follows:

Nationalisation of the commercial banks meant transferring ownership from the private to the public sector. The public sector, he states, is the one which must be responsible for economic developments, especially in a developing country like Iraq. The change will facilitate the participation of the banking system more effectively in these developments. Moreover, he claims, private banks can not provide such an opportunity for the national economy, especially in the cases of less-developed and developing countries.

The transformation of the commercial banks to state-owned banks will change the banking policies of these banks. In other words, the nationalised banks will pay less attention to maximising their profits and more attention to financing the economic requirements. Also, they will create some kind of co-operative relations between different financial and banking institutions in place of a profit seeking competitive situation.

Furthermore, nationalisation can ease the control system of the central bank over the banking system by ensuring that all banks

are subordinated to the central bank. Also, nationalisation will change the relationship between the State and the banking system by making all banks state-owned organisations, which should be complemented with government plans.

Another important effect of nationalisation of commercial banks related to money and its movement within the national economy. Nationalisation of the commercial banks in Iraq put the control of all such banking firmly in the hand of the government and other national institutions.

This type of system means that all the national savings (private and public) now accumulated in the nationalised banks. Then, the state-owned banks can re-direct these savings by providing banking credit for specific sectors in the economy, as required by the national central development plans. Also, this sort of system provided the government with a good chance to benefit from the national savings to the full, in order to finance projects which require a huge amount of money.

2.3.3 The Advantages and Disadvantages of Nationalisation of Commercial Banks in Iraq:

In addition to the motivations of nationalisation discussed earlier in this chapter, the move achieved some advantages for both the banks and consumers, as well as the government. However, the main advantages provided by the nationalisation can be summarised in the following:

1. For the central government, the nationalisation of commercial banks helped in increasing its control on the activities of the commercial banks. Also, it helped to give the government access to additional financial resources,

and redirected them towards centrally planned economic activities.

2. For the banks and their managements, nationalisation led to a market environment that can be characterised in the following: (a) less competition; (b) more opportunities in the market; and (c) in fact, easy profit. Such an environment pleased the top managements of the nationalised banks, because their only measurable indicator of success was in terms of profit.

3. For consumers in general, the first years after nationalisation can be described as good for consumers of banking services in Iraq. In fact, the obvious reason for that was that the newly nationalised banks had expanded their roles by providing financial assistance, i.e., loans and other banking facilities, to wider segments of consumers. Before nationalisation commercial banks had concentrated their efforts on assisting some wealthy consumers and businessmen, especially when it came to providing banking credit. However, after nationalisation banking credit had been given to wider segments of bank customers eligible to get credit according to bank regulations and criteria.

However, a nationalised banking system (i.e., state-owned banks) tends to suffer from a number of disadvantages, and the Iraqi experiment is no exception. Thus, the idea of a state-owned bank has elements of a major contradiction between the personal service expected from a bank and impersonal service expected from a government institution. A bank is seen as distinct and different type of institution mainly for the following reasons:

(a) It provides financial services, and services, as is known, are intangible;

(b) The personal relationship between staff and customers is very important in the case of banks; and

(c) Consumers want respect, better services, and satisfaction. In fact, real experiments, such as the Iraqi experiment, proved that state-owned organisations, especially in terms of their employees, are insufficiently concerned about consumers and their perceptions.

The problem is illustrated by the following example. An employee at a state-owned bank tends to think he / she is only a civil servant, and his/her salary is equal to the salaries of other civil servants working in other government departments. Then, the employee asks, why should he/she be more concerned about consumers than other civil servants working, for example, at the Birth and Death Registration Office?

In other words, the nationalisation of commercial banks in Iraq created a situation described in the following:

(a) Consumers come to the bank because it is the only one in the market: in Iraq from 1974 until 1989 there was only one state-owned commercial bank;

(b) Bank personnel tend to minimise their concern for customers and their job since they think they are public employees (civil servants), their salary does not differ whatever their performance, and their job is necessary; and (c) Banks' top managements face no problems in regard to profits or competition because their success in the market, after all, is guaranteed.

Such a situation led to deterioration in terms of the banks' efficiency, the quality of the banking services, and the bankcustomer relationship. In the eyes of both consumers and banks' employees, banks became indistinguishable from any other government offices such as, for example, a property registration office.

A second disadvantage is the lack of competition. In Iraq, after bank nationalisation, some government decisions led to a number of mergers between the nationalised banks. These mergers led to a reduction in the number of banks, from 17 before nationalisation to only six in 1991 (see figure 2.1). This meant that in 1991 only two commercial (retail) banks operated in Iraq and both were state-owned. Therefore, competition between banks has been reduced to its minimal level. And it is known in the developed world, competition in banking is the key element to achieve effective, better, developed, and satisfactory banking services.

In addition to the above, nationalisation and merger of the commercial banks created what is called the bank concentration problem. In other words, one large bank in the country had a very large number of customers, a very large number of employees, and huge amounts of bank transactions to process each day. Adding to that, communications between the bank's headquarter and its branches are difficult because advanced communication technology was not employed properly.

Finally, for the central government, in spite of the huge financial advantages it gained from a nationalised banking system, it found that the task of controlling a nationalised banking system is a difficult, if not on occasion an impossible task. In addition, the government, as the body responsible in any country for the country's affairs, found that:

Firstly, bank personnel are unhappy about their salaries and wages. Thus, they want the government to treat them distinctly in terms of payment, i.e., they want to be paid in line with the distinction of their job. Because, the government can not pay different salaries and wages for public employees in different departments without severe objections from the others, the performance of banks' employees reduced dramatically.

Secondly, consumers reflected a great dissatisfaction with the quality of banking services, and their treatment by the bank staff.

Finally, a nationalised and highly concentrated banking system had created managerial difficulties for the government in terms of controlling and supervising the bank's management, staff, and activities.

2.4 The Iraqi Banking System in Recent-Times:

Structure, Functions, and Problems.

2.4.1 Structure and Functions:

During the past three decades, the government has made many reforms of the banking system in terms of its structure and functions. These reforms led to re-organisation of the system until it became as it is in recent-times.

Figure 2.3 shows the structure of the Iraqi banking system in 1991, after the reorganisation of the system. Table 2.2 describes the main types of banks in Iraq in terms of their functions, types of customers, and types of banking activities.



Figure 2.3: The Iraqi Banking System in Present Time (As it was in 1991).

Source: Prepared by the researcher according to some publications of the Ministry of Finance in Iraq.

Table	2.2	: Types	of	Banks	and	Their	Characteristics	in	Iraq.
-------	-----	---------	----	-------	-----	-------	-----------------	----	-------

Bank	Function	Types of Customers	Types of bank activities
The Central Bank of Iraq	central banking	 government. banks. other financial institutions. foreign banks. other consumers. 	 issuing and managing the currency. controlling and holding foreign currencies. controlling and supervising banks. act as the bank of the government.
The Rafidain Bank and Al-Rasheed Bank	commercial banking	 companies. businessmen. individuals. foreign banks. 	 accepting deposits. opening banking lending money to customers. providing all types of retail banking. investing in projects.
The Agricultural Bank	providing banking services for the agricultu- ral sector	 farmers. farming companies. other banks. 	 lending farmers and farming companies. investing in farming projects. accepting deposits.
The Industrial Bank	providing banking services for the industrial sector	 industrial companies. individuals. other banks. 	 lending industrial projects. investing in industrial projects accepting deposits.
The Building Bank	providing banking services for the building sector	 individuals. building companies, and societies. other banks. 	 lending companies, individuals, and building societies. investing in building projects. accepting deposits.

As shown in figure 2.3 and table 2.2, the Iraqi Banking System in recent-times consists of the following banks and other financial institutions:

Central Banking:

The central banking services provided by the Central Bank of Iraq, which is the first bank for the government and through which the government practice its monetary and banking credit policies. The bank has the following main functions according to Act 64, 1964:

a. issuing and managing the Iraqi Currency;

 managing the national reserve of gold and foreign currencies;

c. shaping foreign exchange policy and controlling foreign currencies;

d. organising and planning banking credit and controlling
all the banks and other financial mediation institutions;
e. participating in solving monetary and economical crises;
and

f. participating effectively in economic developments by providing the financial needs for such developments.

Commercial Banking:

Currently there are only two commercial banks in Iraq, both state-owned. The first bank, called Al-Rafidain Bank. was established in 1941 and has about 250 branches nationwide; the second, called A1-Rasheed Bank, is newly established (1987) and has about 150 branches nationwide. Both banks provide all the commercial banking services required for different economic sectors in terms of ownership (public, mixed, and private). Their functions are identical, and can be summarised according to the government's Banking Acts and Regulations as follows:

a. acceptance of monetary deposits and opening of banking accounts for its customers (personal and business);

b. lending to its customers by providing banking facilitiesfor them and any other types of banking loans;c. investing its resources in different types of

investments available to the bank;

d. borrowing money from other financial institutions; and
e. providing all other types of commercial banking services for its customers.

The Specialist Banks:

There are three specialist banks in Iraq, providing different types of banking services for different economic sectors. They are as follows:

The Co-operative Agricultural Bank:

This bank is state-owned and provides banking credit and services for the agricultural sector. The main functions of the banks are as follows:

a. providing monetary loans for the co-operative peasants' societies and state farming organisations;

b. providing loans for farmers and other farming companies for several purposes to facilitate their farming activities;

c. investing money and participating in farming companies
with an aim to develop the agricultural sector; and
d. providing any other banking services for the agricultural sector.

The Industrial Bank:

This bank is state-owned and provides banking credit and services for the industrial sector. The main functions of the banks are as follows:

a. providing loans for establishing new industrial projects and for expanding existing ones, in order to develop the industrial sector;

b. investing and participating in private, public and mixed industrial projects;

c. providing banking services for industrial projects; and d. acting as supervisor and consultant for industrial projects.

The Building Bank:

This bank is state-owned also and provides banking credit and services for the construction sector. The main functions of the bank are as follows:

a. lending to the general public in order to build houses, flats and other types of buildings;

 b. lending to co-operative societies to build houses for its members; and

c. lending to the private sector to build such tourist facilities as hotels, restaurants, camps, etc.

Other Financial Institutions:

In addition to the previously mentioned banks, certain other financial institutions provide some types of financial and banking services. These institutions are as follows:

The Insurance Companies:

There are some companies which are state-owned and provide insurance services for the public and businesses such as:

- _ The Life Insurance Company.
- _ The Iraqi Company for Re-Insurance.
- The Iraqi National Insurance Company.

The Pension Fund and Social Security:

This fund is state-owned as a government department which provides services for pensioners after their retirement, such as by paying their monthly salaries.

The Post-Office Savings Fund:

This fund is state-owned as part of the Post-Office Enterprise and accepts savings from the general public by opening savings accounts for them. It has a nationwide network of branches. <u>The Co-operative Fund</u>:

This state-owned fund provides the co-operative societies with loans and other banking services.

2.4.2 Problems of the Nationalised Banking System:

Nationalisation of the commercial banks in Iraq provided several advantages for the Iraqi economy, especially because Iraq is a developing country in which the public sector consists of a high per cent of its structure (about 98% in 1975).

Furthermore, the Iraqi economy was subject to a difficult period of economic transitions, especially after 1972 when Iraq nationalised its oil industry and the oil prices increased sharply in 1974. In such an economic situation, the banking system should play an important role in organising the financial aspects of the economic and social changes. However, the Iraqi banking system in its present structure failed to do so in terms of providing all the financial needs for different economic sectors, and it was discovered that it was not able to handle the sharp increase in the number of banking transactions and providing the necessary banking services.

In summary, the present banking system in Iraq suffers from the following main problems:

Banking Concentration Problem:

The problem of banking concentration is an important factor which affected the performance of the different banks within the system in Iraq. After the nationalisation of the commercial banks in Iraq in 1964, there were a number of nationalised commercial banks, but soon after that, these banks were merged into only one commercial bank, called the "Rafidain Bank". This bank provided banking services for many years as the sole commercial bank, but its job became very difficult, especially after 1974, because of the rapid growth in demand for banking credit and services, and only one commercial bank with one headquarter could not do it all. Therefore, another state-owned commercial bank, called Al-Rasheed Bank, was established in 1987. However, this did little to help, and the problem of banking concentration still overshadows commercial banking in Iraq.

The same pattern emerges for other banking institutions. Currently there is only one agricultural bank, one industrial bank, one building bank, and one of each of other financial institutions. These, too, are unable to satisfy the increasing demand for banking services.

Lack of competition:

The Iraqi banking system consists mainly of one bank for each economic sector. This has led to a banking market without any sort of competition, no choices for consumers, and low efficiency. It is well known that at the heart of banking is competition for customers. The means to win them is to provide better and competitive services in many terms. Also, banking is the job of creating money and opportunities, gaining customers and providing the best services for them, and it requires distinct management and

staff to achieve that. Such requirements can be satisfied only under a high level of competition.

Managerial Problems:

Because the nationalised banks are state-owned, many of their staff feel that they are working in enterprises which are similar to any other public department, and that their job and performance should be regulated by this fact. This has made bank management and staff forget the proper role of banks and their personnel. As a result of that, nationalised banks suffer from too much regulation, poor performance, and low consumer satisfaction. In order to make them more effective, they should be freed from unnecessary regulations and should be treated as independent commercial organisations.

Lack of Marketing Strategies:

The lack of marketing strategy is another type of managerial difficulty, but it is of special importance in the case of banking. All commercial banks and other financial institutions operating in Iraq do not have acceptable use of marketing strategies and methods to market their services. Even the use of advertising is still very poor: only a few advertising campaigns have appeared in some national newspapers or magazines to advertise for existing services. Furthermore, not one market research investigation has been carried out by any of the operating banks to investigate any of the marketing problems they face. Therefore, it is appropriate to mention that this study is the first of its type to be undertaken about one aspect of the banking services in Iraq.

That no market research has been undertaken was because: (a) Banks are state-owned, and therefore their management and staff behave like civil servants doing departmental jobs. (b) The instability of the banking system in terms of its structure, as mentioned in this chapter. (c) Lack of specialist staff required to implement and carry out marketing techniques. (d) Until very recently, most consumers in Iraq usually dealt with banks only when their needs for some financial assistance were very desperate. Otherwise, they did not like to engage themselves in any banking activities. This type of situation is called by bankers in Iraq "The Lack of Banking Education", and the banks are responsible for this because their poor marketing strategies participated in creating it.

Table 2.3 explains the main differences between the Iraqi banking system and the banking systems of other developed countries in terms of some important bank characteristics.

Table 2.3: Comparison Between the Iraqi and the Developed Countries Banking Systems.

Characteristic	Developed Countries	Iraq	Implications
1. Ownership	mostly private-owned banks	state-owned banks	heavily regulated by government
2. Competition	very high	very low	poor service & no choice for consumers
3. Types of services	a large number of various types of services	a limited number of services	less choices of services for consumers
4. Number of banks, and branches	many banks, with a lot of branches	fewer banks & branches	less choices of banks to select from by consumers
5. Technology	has been used heavily by banks	has not been used in a proper manner by banks	poor service in all terms
6. Bank staff	highly trained and qualified	less trained & less qualified	poor service in all terms
7. Marketing	the marketing concept has been employed effectively by banks	marketing neglected by banks	consumers wants and needs are ignored
8. Consumers	consumers and their satisfaction are very important issues	consumers and their satisfactions are not important	consumers dissatisfied
9. Use of banking services	used heavily by consumers	used occasionally by consumers, i.e., when it is required	less desire to deal with bank by consumers and loses of many potential marketing segment of consumers

2.5 Conclusions:

The review of past and present developments of the Iraqi banking system in terms of its structure, functions, and problems showed that the financial system in Iraq is still limited in terms of its structure and performance. The purpose of this study is an attempt to investigate consumers' attitudes toward commercial banking services in Iraq. This review will help in understanding consumers' responses when asked to evaluate commercial banking services in Iraq.

One of the most important conclusions highlighted by this chapter is the adverse effect of state ownership of the commercial banks, particularly how it affects the performance of banks as well as the satisfaction of consumers.

Commercial banks in Iraq are totally state-owned, and this has two different dimensions. The first dimension is that at one time, there were private commercial banks in Iraq which were then nationalised, mainly for reasons: (a) their supposed two inefficiency to provide the required banking services for the national economy, and (b) their alleged interest in maximising their share of profits only, without caring for the national interests. In fact, the nationalisation decision was taken by the government at a when the concept of nationalisation was dominant in the time developing world. By the late 1970s and throughout the 1980s until recent times, however, another concept was and still is dominant: that of "privatisation" or denationalisation.

As present thinking, in general, is towards privatisation, some sort of privatisation programme will be of substantial help towards improving the performance of commercial banks in Iraq. But, such a step depends on the second dimension which is in fact a

dilemma for the banking sector in Iraq. As such, it requires hard work and probably a long time to achieve an acceptable solution. Such a dilemma can be summarised into three points:

Firstly, there is no private sector in Iraq which can invest in establishing or managing commercial banks on a large scale. Secondly, the public sector is still dominant in the Iraqi economy. Therefore, private commercial banks will find that it is very difficult to operate in such an economic environment. Thirdly, neither bank management (in terms of staff and other managerial skills) nor current consumers can cope with major changes in the near future.

It will be possible to achieve, however, if available time is dedicated to this end. Privatisation is essential for improvement in the performance of commercial banks and the creation of a new era for the banking sector in Iraq.

CHAPTER THREE

REVIEW OF LITERATURE RELATED TO COMMERCIAL BANK MARKETING WITH SPECIAL REFERENCE TO THE SERVICE OF SAVINGS ACCOUNTS PROVIDED BY COMMERCIAL BANKS

3.1 INTRODUCTION.

3.2 COMMERCIAL BANKING: AN OVERVIEW.

3.3 MARKETING OF BANKING SERVICES.

3.3.1 THE CONCEPT OF BANK MARKETING.3.3.2 DEVELOPMENT OF BANK MARKETING.3.3.3 MARKETING RESEARCH IN BANKING.

3.4 REVIEW OF PREVIOUS MARKETING RESEARCH STUDIES RELATED TO CONSUMERS' SAVING BEHAVIOUR.

3.5 CONCLUSIONS.

3.1 Introduction:

The main purpose of this chapter is to present the theoretical framework of bank marketing in general, with particular reference to commercial banking. In addition, a review of previous similar and comparative studies is to be presented.

Investigation into the theoretical framework of bank marketing and the relevant prior studies is essential in order to: (a) provide a better understanding for the problem under investigation; and (b) help the researcher in formulating the research model.

With so much material covered in an effort to build a theoretical framework for bank marketing, it will be useful at this point to be specific and mention aspects of literature review which are related to this study's main topic, i.e., the service of savings accounts provided by commercial banks, and draw pertinent conclusions. Also, because this study is an empirical investigation of consumers' saving behaviour in Iraq, the literature review will be limited to similar and comparative studies dealing with the subject of marketing of commercial banking services with special reference to savings accounts.

3.2 Commercial Banking: An Overview:

Hindle (1985) defines commercial banks as those banks that are most visible to the general public. Their main business is taking deposits and making loans to individuals and industry - a business that they conduct through a network of branches. They are commercial in two senses: they make loans to commerce (i.e., for the

financing of trade), and they are themselves commercial profitoriented businesses.

Commercial banks are sometimes called retail or deposittaking banks depending on the banking system and the country they operate in. However, as relates to this study, the term commercial bank in Iraq is used to describe what might be called in Britain or elsewhere a retail / clearing / or deposit-taking bank.

According to Crosse and Hempel (1980) commercial banks have a number of specific tasks. They perform a number of interrelated functions, many of which are necessary parts of our private, freeenterprise system. Commercial banks bring into being the most important ingredient of the money supply-demand deposits through the creation of credit in the form of loans and investment. Banks are the custodians of the community's money as well as the suppliers of its liquidity. Commercial banks also participate with other institutions in the process of accumulating and investing savings, and perform a number of other services.

These functions explain that commercial banks provide several types of retail banking services to a broad range of consumers. However, in this investigation, the service of opening savings accounts by commercial banks is the service of concern.

Reed, Cotter, Gill, and Smith (1984) mention that commercial banks perform a very important service to all sectors of the economy by providing facilities for the pooling of savings and making them available for economically and socially desirable purposes. The saver is rewarded by the payment of interest on his or her savings, which are safe and in a highly liquid form.

This type of commercial bank activity is called the process of mobilisation of savings, which is very important in any national

economy. On the other hand, savers are able to obtain return on their savings and at the same time achieve the degree of safety, liquidity, and other terms compatible with their objectives.

3.3 Marketing of Banking Services:

This section is designed to discuss three important aspects in marketing of banking services. The first part is devoted to highlight the concept of bank marketing. The second part reviews the development of bank marketing. The third part is an overview of the use of marketing research and its applications in banking.

3.3.1 The Concept of Bank Marketing:

Marketing in banking has been defined by Crosse and Hempel (1980) as the creation and delivery of consumer - satisfying products (services) at a profit to the bank.

This definition highlights three important dimensions in bank marketing. Firstly, banks should work hard to create new services and to deliver existing services in such a way that satisfies consumers' needs and wants. Secondly, the concept of marketing in banks should be consumer - oriented rather than product-oriented. Thirdly, as mentioned by Reekie (1972), bank marketing is that part of management activity which seeks to direct the flow of banking services profitably to selected customers. Thus, marketing of banking services seeks, in addition to serving and satisfying customers, to achieve an acceptable level of profitability.

However, in any business the marketing concept has two coextensive dimensions as identified by Brien and Stafford (1967). The first dimension of the marketing concept is operational. Thus, at the operating level the marketing concept includes many functional decision areas whose end objective is the development of the optimum

method of serving profitable opportunities in the market-place. Thus, marketing management typically involves the formulation of programmes that cover four major strategic areas: Product, Pricing, Place (Distribution), and Promotion. The second dimension of the marketing concept is philosophical. It argues that the needs, wants, and value - attitudes of consumers should be the common focus of all marketing decisions.

Meidan (1984) mentions that the adoption of the marketing concept will recognise that bank customers are changing in terms of their wants, needs, desires, expectations, and problems. Banks must define these in explicit terms and then evaluate their offering in customers' terms, that is, satisfying customers' needs at a profit to the bank.

Therefore, bank marketing management should consider the importance of the following activities in bank marketing as outlined by Crosse and Hempel (1980):

 Identification of consumers' wants and needs through marketing research.

2. Development and management of products fulfilling the identified consumers' wants and needs.

3. Promotion of the products developed for customers.

4. Establishment and maintenance of meaningful relationships with the community and other special groups.
5. Co-ordination of marketing as an integral part of most banking activities.

However, in order to provide better understanding of the concept of marketing in banking, it is important to refer to the marketing function or the concept of marketing mix in banking. Baker (1977) defines the nature and scope of the marketing function in

terms of what have been called the four P's of marketing: Product, Price, Place and Promotion.

In the following a listing for the four elements of the bank marketing function:

1. Product

In banking product means service; thus, banks provide services for the consumers which they need. Therefore, the product (service) is the central element of bank marketing function. Consumers want a variety of services that they can choose from provided under one roof. Products in banking can be banking services - various categories of deposits and loans, and other services (travel agency, leasing, trust services etc.).

2. Price

The term price in banking refers to interest rate, service charges, discount, trust fees, and other charges and fees. This marketing element is very important in banking because of its importance as a competitive factor. Thus, as relates to price, consumers tend to seek high interest returns with low costs when they deal with banks.

3. Place

Bank customers consider the convenience to home, place of work, and market-place, of bank location as one of the major priorities in selecting a particular bank. Also, whether the bank has many branches located conveniently or has Automatic Teller Machines (ATMs) located in convenience places, are among factors considered as important by customers when making their selection decisions between banks.

Thus, the element place refers to the distributional channels of banking services easily available to consumers and consists of

location of banking offices, design and decor of banking premises, and location and convenience of banking facilities.

4. Promotion

Promotion is one of the visible elements in bank marketing. Promotion consists of different ways to inform consumers about the services provided by the bank to them.

Advertising is one of the most common methods used in promotion. Advertising campaigns can be carried out through the media, brochures, bulletin boards, leaflets, direct mail, etc.

Also, non-advertising methods can be used in promotion, such as economic and other education programmes which do not carry explicit advertising material, financial libraries, and participation of banks in social or sporting activities.

In addition, personal selling is another method which can be used by banks to promote bank services.

3.3.2 Development of Bank Marketing:

In the 1950s and before banks believed that as suppliers of needed services, the public would come to them, therefore, the concept of marketing, as it has been defined in this investigation, was largely ignored by banks. Many of them thought that marketing meant merely advertising, public relations, and sometimes personal selling.

This failure to apply the marketing concept in banking was due to several factors. Person (1964) pointed out the following reasons behind the ignorance of the marketing concept by banks at that time:

a. Banks are profitable;

b. The direction of a bank is complex, requiring close attention to financial matters, and the two-way process of marketing banking services differs from the usual marketing process as it is performed by the sellers of products;
c. Bankers do not consider aggressive marketing practices to be compatible with the traditions and ethics of banking; and

d. The lack of qualified personnel in marketing and marketing research in the labour markets from which the banks can draw caused the dearth of marketing research in banks.

However, during the late 1950s and throughout the 1960s, banks discovered that they had to adopt the marketing concept. Donnelly and Ivancevich (1969) point out that due to increasing competition between banks and other financial institutions on one hand, and increasing customers' demands for better and wider banking services on the other, banks were forced to acknowledge that marketing strategies based only on advertising, public relations, and personal selling were no longer sufficient to keep their share of the market.

What they needed was a new marketing strategy based on the marketing concept. In other words, a consumer (market) - oriented, instead of product - oriented, strategy was required to deal with the rapid changes in the market-place. All the same, in comparison with manufacturing industries, the application of the marketing concept in the banking industry was relatively slow to emerge. Clowe and Diina (1978) have shown that development of marketing and marketing research as formal disciplines in the financial industry followed progress in industry by a significant margin. For example,

according to Meidan (1984), the possible application of the marketing concept in banking was mentioned for the first time in the USA, at the American Banking Association (ABA) Conference in 1958. However, in the later 1950s and throughout the 1960s, professional marketing and research activities in financial institutions significantly closed the gap with other industries.

Donnelly and Ivancevich (1969) outlined four essential characteristics which best describe the marketing concept adopted by banks by the late 1960s:

 Discovering customer wants and needs through marketing research before customer - oriented programs are developed.
 An emphasis on profitability of marketing operations, rather than on their ability to simply generate increased revenues.

3. Establishment of a marketing department and utilization of trained personnel.

4. An emphasis on the planning of marketing operations.

Perez (1983) mentions that in addition, banks were beginning to understand that convenience was the key to the future of retail banking. The term convenience here means mainly convenience of bank location, and time of bank operations in terms of opening hours and length of waiting time. Also, banks adopted heavy advertising schedules and sales promotion programs to help market banking services.

Capaldini (1974) said that well-informed observers cite the early 1960s as the period when the first real evidence appeared that marketing was being recognized in banking. As the early 1960s became the mid - and late 1960s, businessmen and consumers wanted more convenience, more services, more personal (or technical) assistance.

Commercial banks therefore became increasingly interested in the retail side of the business.

In short, the very nature of banking changed in 1960s. And the change, the response, was essentially a marketing response, i.e., doing what the customer wanted. Thus, the response by banks in the 1960s as described by Perez (1983) amounted to a fundamental change in focus.

According to Perez (1983), as competition continued to increase in the 1960s, banks began the banking innovation. For example, Bank of America offered customers 350 separate financial services. Also, banks throughout the 1960s began to realize that marketing was a lot more than smiling friendly tellers. The banks started to ask their customers in surveys, and responded to customers' answers by improving services. In addition, bankers were beginning to understand the concept of market segmentation and its importance in understanding and analysing consumers.

Perez (1983) describes the changes in banking throughout the 1970s as follows:

1. The bank marketing profession changed dramatically in the 1970s, as for example: (a) marketing positions were increasingly filled by people with formal education in marketing; and (b) specialities and new positions emerged such as: advertising manager, public relations officer, research director, and sales promotion manager.

2. The major influence on bank marketing in the 1970s was the advent of electronic funds transfer systems, which marked a turning point in terms of service speed and efficiency, e.g., Automated teller machines, Direct deposit of pay-roll, Pay-by-phone system, Point-ofsale system, Credit and debit cards, Pre-authorised funds transfer, and Automated clearing house.

As described by Capaldini (1974), the 1970s was characterised as an era of moving from an age of things (products) to an age of people. Thus, bank marketing became an increasingly important bank activity, and was recognised as a job for everyone in the bank, not just the so-called marketing specialists. In other words, the marketing concept in banking became more oriented towards consumers and satisfying their wants and needs, i.e., consumeroriented, rather than product-oriented.

Lewis (1984) emphasised that throughout the 1970s and 1980s, the nature of banking had altered and progressed as a result of a combination of changes in the market-place. The changes in the market-place were due to the legislative, economic, business, technological, and social developments which affected both personal and corporate customers. As a result, the marketing concept became increasingly relevant and important for the banking industry.

In the 1980s, customers' needs and wants were increasingly expanded. Bank marketing throughout the 1980s had become more consumer-oriented by attempting to achieve the following marketing aims:

1. Providing better, competitive, advanced, and more numerous services to customers.

2. Conducting more marketing research surveys in order to reach a better understanding of consumers' behaviour, attitudes, wants, and other characteristics.

3. Responding rapidly and efficiently to market, social, economic, and technological changes.

4. Bank public relations must be marked by more planning, more positioning of the distinctive image of each bank, and

a higher awareness of the value of public relations to the banking organisation.

5. Devoting a large share of marketing budgets on internal programs to improve the skills of employees.

6. Developing advertising campaigns to be in line with modern social changes and new consumer thinking.

7. Improving personal relationships between bank personnel and customers through building a friendly bank- customer relationship.

Beveloping the personal selling to became a consultative selling relationship between bank personnel and customers.
 Establishing a marketing information system in order to have access to the latest information about products, consumers, competitors, and other variables.

Thus, by the end of the 1980s the concept of bank marketing had been developed and adopted comprehensively by most banks and other financial institutions. In fact, it can be concluded that marketing in banking is expected to receive more attention and importance in the financial industry during the 1990s.

3.3.3 Marketing Research in Banking:

Hodges and Tillman (1968) defined marketing research as the gathering, recording, and analysing of all facts about problems relating to the transfer and sale of goods and services from producers to consumer. Ross (1968) defined marketing research as the systematic, objective, and exhaustive search for and study of facts relevant to any problem in the field of marketing.

These two selected definitions spotlight the fact that marketing research, in general, is the process of collecting and

analysing facts about problems which are related to present market customers and potential market opportunities.

Meidan (1984) mentions that the concept of marketing research in banking is relatively new. It was not until the early 1960s that the first few banks realised the importance of marketing research for both their future planning and current activities. By 1973, all the major UK clearing banks had organised and staffed a marketing department and most had a formal market research function.

In the banking industry, two sources of data (facts) relating to a particular problem are available: internal and external. The early attempts to apply quantitative marketing research to banking depended heavily on the internal sources of data available to the banks. Thus, bank records contain a huge amount of data about their customers such as socio-economic, demographic, and other financial information about customers. Such information can be about: (a) existing customers, (b) new customers, (c) corporate customers, and (d) in some cases, prospective customers.

As time progressed, especially by the end of the 1970s, the need for more comprehensive information drove banks to seek additional sources of information available outside the bank. By reaching this stage, real marketing research investigations have been carried out by many banks using all the available internal and external data for the purpose of solving different marketing problems.

Once banks implemented the marketing concept, i.e., bank marketing became customer-oriented, especially in the 1980s, marketing research in banking became more sophisticated in the manner in which it collected, reviewed, and analysed data and recommended specific action.

Meidan (1984) mentions that marketing research in banking is a tool for problem solving. It refers essentially to two categories of activities: (a) problem solving, and (b) forecasting results of alternative marketing and business decisions.

Also, the same author mentions five major areas where marketing research techniques have been successfully applied. These are: (1) market segmentation; (2) customer behaviour and bank selection criteria; (3) customer loyalty; (4) bank product / service studies; and (5) bank branch research.

In addition to the above-mentioned areas, new trends have been developed in bank marketing research, especially in the last decade. As mentioned by Meidan (1984), computer technology offers opportunities for cross-evaluating bank marketing services. The research in this field has recently reached quite an advanced stage both in quantity and in quality terms, attempting to supply quantitative answers to some of the most pressing marketing problems facing the banking industry. Figure (3.1) depicts the recent developments in bank marketing research and its applications in the banking industry.



Figure 3.1: Recent development in bank marketing research.

However, in a more recent discussion, as described in figure (3.1), Pezzullo (1989) outlines the recent developments in the use of marketing research in banking. According to Pezzullo (1989) marketing research enables the marketer to identify and analyse the differences between various target markets. It provides information on how the members of various markets are motivated, how they
perceive the world, and how they behave in response to various stimuli (for example, small differences in interest rates paid, or cash bonuses to attract deposits).

The same author also mentions that marketing research is invaluable in the design of product mix strategies. New product ideas can come from market research; new product concepts can be tested in marketing research, and alternative pricing schemes can be tested among the target market. Research on the attitudes and behaviour of the target market can provide input for the people who design the advertising. Then, research can be used to test the reactions to advertising's creative strategy and the effectiveness of the ad copy.

Also, Pezzullo (1989) indicated that marketing research can identify the geographical areas into which the bank might expand its distribution network. It can be used to measure the sales effectiveness of the retail and commercial customer-contact staff. Completing the planning loop, research is an integral tool for evaluating and controlling the marketing management process. It can measure the success of the marketing strategy and track its effect over time.

Finally, according to Morgan and Piercy (1990), for financial service organisations the 1990s wi11 be dominated by the increasingly competitive nature of the financial services marketunlikely to be restricted to place. Competition is those organisations that have historically serviced the market for financial services: new competition is likely to emerge from the retail sector and also, for example, from post- 1992 Europe. The key problem that needs to be addressed is that of achieving competitive differentiation in offering largely generic financial services. It

was the emerging competitive environment of the 1980s that led to the introduction of a marketing function in many, for example the UK financial services organisations. The continuing competition of the 1990s is likely to lead to a reappraisal of the role of marketing in this sector. In fact, marketing research in the financial services sector will play important role as a key function in bank marketing. <u>3.4 Review of Previous Marketing Research Studies Related to</u> Consumers' Saving Behaviour:

As shown earlier in this chapter, early marketing research studies in banking were carried out by some banks in the early 1960s. By the end of the 1960s and later, more banks and researchers applied marketing research techniques for solving different marketing problems in the banking industry. Having discussed this process, it is possible now to turn to a review of previous marketing research studies related to consumers' savings behaviour and attitudes towards savings accounts (see Exhibit (1) for a summary review of the below mentioned studies).

One of the earliest consumer savings behaviour studies was carried out by Claycamp in 1965. Claycamp (1965) conducted an exploratory investigation of consumer savings behaviour in order to: (1) identify characteristics which discriminate between three customer groups, (2) determine the relative importance of different types of variables as discriminating variables between the three groups of consumers, and (3) investigate implications for marketing research of commercial banks and savings and loan associations.

Claycamp utilized three categories of variables in his study: (a) socio-economic / demographic variables; (b) asset variables; and (c) personality variables.

Sample members were classified into one of three categories: savers who held accounts at commercial banks (group c), savers who held accounts at savings and loan associations (group s), and savers who held accounts in both of them (group b). A Two-Way Discriminate Analysis was used in Claycamp's study.

Claycamp tested four differences between group means for each variable included in the analysis. Comparing two groups at a time, he reported that Group C was significantly younger, had more dependents living at home, less investment in stocks, and lower occupational status than Group S. Other two group comparisons showed little significant differences.

By utilizing a two-way step-wise Discriminate analysis to identify deferences between group C and group S, Claycamp derived a Discriminate function containing eight final variables: five were psychological, two were socio-economic / demographic, and one was from the asset category. Furthermore, Claycamp noted that psychological variables were better discriminating variables than either socio-economic / demographic or asset characteristics.

By analysing Claycamp's findings, the following conclusions can be reported: (1) Socio-economic variables (excepting age) were of little value in discriminating between different consumers groups; (2) It seems apparent that differences existed between the groups in terms of psychological variables that can be related to marketing strategies; (3) Savings motives variables did not prove to be important as discriminating variables between the groups; and (4) Claycamp's findings indicated that only psychological variables proved to be important discriminating variables between consumer groups.

as discussed below, much research relating to However, consumers' saving behaviour carried out later indicated that both consumers' socio-economic variables and motivations were of great importance in discriminating between different groups of consumers behaviour. in regarding their savings addition to other psychological variables. Beside this contrary evidence, Claycamp's study was subject to the following limitations:

1. The sample was taken from one point in time, one geographic region, and households who were known holders of savings accounts. 2. Results found at this point in time, i.e., at the time when the study was carried out, may have been caused in part by prior promotion (effect). Thus, variables which distinguish between the groups may not have generality in situations where other types of promotion (effect) have been employed.

3. Since a step-wise search of the data was made in fitting the best model, statistical tests of significance are not strictly applicable. Consequently, validation of these results on a new body of data should be done before making generalizations.

Martin (1967) investigated the service of Savings and Loan Institutions (S&Ls) in new sub-markets searching for predictive factors related to savings inflow in new Savings and Loan Offices. One dependent and nineteen independent variable were used in the analysis. Data were analysed by applying a Step-wise Multiple Regression Program that designated the variable "annual savings inflow" as the dependent variable. The findings of regression equation revealed that three independent variable were positively related factors, namely: the number of competitors, the proportion of population age (65 and over), and population growth. Also, the findings showed more significant relationship between the dependent

variable (mean annual savings inflow) and four other economic and demographic variables, which were: percentage of population age (65 and over), social rank, percentage of minority households, and mean household size.

Regarding the competition factors, Martin's findings revealed that the two competitive factors - the number of S&L offices within a two mile radius of subject location, and the number of aggressive competitors within a two mile radius - did affect the savings inflow of S&L offices.

Martin in his study attempted to search for predictive factors which can help S&L offices in predicting their savings inflow. His model built on offices operating area attributes only, while other factors which can affect savings inflow were neglected. This sort of thinking suffers from the neglect of important variables which might effect savings inflow, such as: consumers' perceptions, motivations, and other socio-economic and demographic characteristics.

Therefore, it can be concluded that Martin's model lacks a comprehensive approach to study the problem of predicting savings inflow. However, it did explain the relationship between the dependent variable (i.e., annual savings inflow) and some offices' area economic and demographic attributes.

Allaire (1972) conducted a study based on competition between banks and so called near-banks, by establishing the reasons behind the selection of a particular financial institution by consumers. He used the raw data obtained by the Royal Commission on Banking and Finance in Canada from 1221 respondents who were asked to answer the following question: "How do you determine the choice of financial organizations (bank, trust company, credit union, etc.) in which to

open savings account - that is what were the reasons for your choice of particular financial organization?".

Allaire concluded from the raw data he obtained that the decision to select a particular financial institution is not a very traumatic one. Respondents ranked convenience of location in first place (26.7%), while they ranked higher rate of interest in fifth place (6.5%). Also, the findings showed other important reasons behind selecting a particular financial institution such as: simply preferred it - logical place (17.5%), security of institution (12.3%), familiarity with personnel or methods (8.2%), and other important reasons with lower rank.

These findings provide clear evidence that convenience is the prime reason for the selection of a bank, and it is probably true in many circumstances. In Allaire's study, the term convenience contained three factors: convenience of location, logical place, and convenient hours of operation, which in total counted for (48.8%) in terms of importance.

One can conclude from Allaire's study findings that convenience has become one of the most important dimensions in the marketing of banking services. Thus, consumers' perception of a bank depends largely on the convenience it provides to consumers in terms of location, time, and facilities. In fact, many bank marketing research studies had concluded that convenience was the most frequently given reason for selecting a particular bank by consumers.

Mason and Mayer (1974) analysed the effects of some factors in banking patronage. The main focus of the analysis was on determining consumer's reasons for maintaining checking accounts

image perceptions in the minds of the respondents. In addition, the findings indicated that consumers want to achieve more than financial gains by chosing a particular bank. They want convenience, good treatment and respect.

Finally, in spite of the tentative generalisations derived from the findings of Mason and Mayer's study, which can be of value in strategy formulation by various types of financial institutions, the investigation lacks a more sophisticated and advanced approach to tackle the problem of analysing patronage factors.

Reynolds and Wells (1978) conducted a study on life style analysis as a dimension for bank research, based on a nationwide survey of adult males. For the purpose of analysis, respondents were assigned to one of three groups, as follows:

1. Non-savers: those indicating no deposits during the past year.

- Infrequent savers: those reporting from 1-11 deposits during the past year.
- 3. Frequent savers: those reporting 12 or more deposits in the past year.

Their findings indicated that frequent savers demonstrate a relatively high participation in buying government bonds and common stocks, heavy use of bank credit cards for its convenience with less commitment to borrow money.

Also, frequent savers were reported to be more engaged in a wide range of community activities. When compared to the other two groups, frequent savers are most apt to be pleased with their current financial situation, to spread their investment (savings, bonds, and stocks) and to be near-term financial optimists. They feel financially secure, they are most likely to continue to save, to spend pragmatically, to shop carefully and to be value conscious

and to demand satisfaction. They are influential in their neighbourhoods and are opinion leaders.

Frequent savers are most apt to be appearance conscious and to demonstrate this consciousness in the way they dress. They are likely to be highly conscious of nutrition and health, to be cosmopolitan urbanites, active, on-the-go, and confident travellers who read Sunday and evening newspapers, weekly news magazines, sports and home magazines. They are likely to enjoy music, football, comedy and talk shows. Frequent savers are hard working people, dedicated to success, and adventurous about leisure time pursuits. They are apt to try new things, but only after obtaining information about them. They are satisfied with life and living, not overly concerned for the future.

The above findings outlined by Reynolds and Wells demonstrate the importance and usefulness of life style analysis in providing a better understanding of bank customers' behaviour, especially those who are frequent savers, and their psychological profile.

Nevertheless, two limitations can be mentioned here: (1) In spite of the extensive description provided by the study for frequent savers' life style, little information has been given about non-savers' and infrequent savers' life style; and (2) The study sample contains only adult males. In fact, frequent savers could be males and / or females. Therefore, the findings did not represent all consumers' life style characteristics.

Arora, Cavusgil, and Nevin (1985) examined a number of factors which are important in the choice of a financial institution, and identified those which are similar for, and different between, banks and savings and loan institutions. Data were analysed using two techniques. Factor analysis, using principle

component analysis with varimax rotation, were applied to analyse variables related to the importance ratings for financial institutions characteristics by respondents. Factor congruency, which gives the coefficient of congruence between two factors, was used to measure the degree of similarity between two factors.

The findings concluded that there were seven common factors extracted to both financial institutions, Banks and Savings and Loan Institutions. The variables with very high loading on the first common factor were: concern for customers, capable management, friendly and knowledgeable personnel, well established, good reputation, safe, reliable, and insured. The second common factor included variables such as: efficiency of tellers, and easy-in, easy-out drive-up tellers. Variables with large values of loading for the third factor were: availability of trust services and safe deposit boxes. The dominating variables of the fourth common factor were the number of branches and their proximity to place of work and place of shopping. The fifth common factor can be labelled as the convenience of non-personal banking. The sixth common factor was availability of a consumer loan and home loan. Finally, the seventh common factor can be labelled as interest competitive.

Collectively, these findings suggested that although consumers consider such factors important in their choice of a financial institution, still, these factors did not help consumers in chosing between a bank and a savings and loan institution. In the eyes of consumers, banks and savings other words, in institutions share mainly similar characteristics. However, the results of the study did indicate a number of factors which were unique to both banks and savings and loan institutions which can be used by customers as basis to choose between them. Such factors

included hours of opening, location near home, and variety of services.

Goudge and Green (1986) examined the role and contribution that market research had played in the development and launch of Midland Bank's savings accounts for children (Midland Bank is one of the big five retail banks in the UK). A series of group discussions were carried out with children aged 10 - 16 years, and parents of children aged 0 - 16 years in order to: (a) examine in general the attitudes of children and their parents to savings accounts for young people; (b) explore the attitude and usage of savings accounts by children in different age groups in order to identify which age groups offered greatest potential for the new accounts; and (c) determine what features the account should have if it were to attract and retain the interest and support of both children and parents.

The findings of twelve group discussions revealed that attitudes of both parents and children towards saving and savings accounts were very encouraging. Children regarded savings seriously, and even the youngest respondents (those aged 10 years) were aware that accumulating money and looking after it was a good thing. Also, opening a bank account was regarded as one of the conscious steps a child can take into adulthood. Meanwhile, parents considered it their duty to teach their children how to save and manage their money. But, their task was made very difficult as the child became teenager by the child's increasing independence.

The findings regarding incentives provided by the bank to children who are savings accounts holders indicated that incentives were warmly welcomed by both parent and children provided they were sensible, useful, and not too extravagant. Furthermore, children

were very happy and even keen to wear, use, and carry goods bearing the name of the company that had provided it, and parents were particularly interested in incentives that saved them some money.

The next stage of the Goudge and Green study directed towards carrying out qualitative work among children and parents in order to assess the overall appeal of new savings packages for children. The findings of this stage revealed that the parents of children aged 10 - 15 years were concerned about what their children like or dislike. Also, the free gifts offered by the bank, as incentives, were very well received by both parent and children, and in particular children in the 10 -15 years age group were most impressed. Their parents, while still highly rating the free gifts, attached more importance to high interest rate.

The findings above provided some insights about a particular market segment, i.e., children. Moreover, in the highly competitive savings market they helped to clarify what banks should offer to gain such a type of customer, who will represent the next real generation of bank customers. Therefore, undertaking a market research investigation about children and their parents' attitudes towards savings and savings accounts provides a good opportunity to the bank to define such type of consumers' wants, needs, and desires. However, although the investigation was a good attempt to understand such a market segment, it lacks the use of advanced marketing research techniques, especially a quantitative method in investigating such an important issue.

Langrehr (1986) investigated competition between banks on one hand, and savings and loan institutions on the other, for consumers' savings, and the criteria for consumers to select between them.

A longitudinal study of consumer images of financial institutions was conducted to measure the possible changes in financial institution evaluations. One assumption to be tested was that as the product offerings of the two types of financial institutions became more similar, so would consumer perceptions of these organizations' products. Another assumption was that since the changes in the industry occurred on product features and not on institutional or subjective characteristics, consumer would not change their perceptions of financial institutions on these subjective characteristics. Data were collected at two different times. The first survey was conducted in 1980, and the second, in 1983. In each survey, a random sample of 750 respondents was used.

The findings of the study can be divided into two dimensions. The first dimension can be called product evaluation. Regarding this dimension, the findings confirmed that there were significant changes in the evaluation of banks and savings and loan institutions in terms of their products (types of services). An overall increase between 1980 to 1983 in consumer acceptance of services provided by savings and loan institutions was reported. These services included checking accounts, credit cards, car loans mortgage loans, savings return, and other useful services. These findings did not mean that consumers preferred savings and loan institutions over banks, but rather referred to changes in consumer perceptions. All the same, the findings confirmed that banks still enjoy a higher acceptance and better perception.

The second dimension can be labelled as institution evaluation, and included seven variables. The study reported that there were significant shifts between 1980 and 1983 in the responses on four variables, namely convenient hours, providing useful

information, responsive to needs, and safety. However, there were not any significant shifts in evaluations of convenient location, fastest service, and experienced personnel. Langrehr concluded that consumers' preference for banks responses over savings and loan institutions remained unchanged.

The study reported here enjoys the following advantages: (a) the time scale has been used effectively for the purpose of evaluation; (b) such an approach (i.e., a study carried out in two different periods) can provide accurate and revised results about the problem under investigation; and (c) the evaluation process has been carried out according to two dimensions: product evaluation, and institution evaluation. Nevertheless, such an approach has certain disadvantages, such as being time consuming, costly and possibly affected by other social and economic changes.

Alessandra, Knierim and Yavas (1986) attempted to provide some insights into the banking behaviour of consumers in terms of their age focusing on older consumers.

The principle idea behind this study was the growth in the number of older consumers on one hand, and their untapped investment potential on the other. Older consumers represent an attractive market segment for financial institutions since beside their growth in numbers, given an increase in their income, they are likely to save or invest. Therefore, the main purpose of the study reported here was to uncover the attitudes and behaviour of older consumers toward selected aspects of financial services, with particular emphasis on Savings and Loan Institutions (S&Ls).

The findings revealed that in relation to the importance of selected factors in choosing a Savings and Loan Institution, the factor "convenience to home" ranked first, followed by helpful and

pleasant personnel (second); good reputation (third); free services (fourth); full range of services (fifth); fast and efficient service (sixth); convenient to shopping (seventh); convenient hours (eighth); parking facilities (ninth); automated teller machines (tenth); and convenient to work (eleventh).

However, regarding the influence of information sources in choosing a Savings and Loan Institution, the source "saw it in area" ranked first, followed by friends or neighbours (second); television (third); newspaper (fourth); real estate agent (fifth); billboard (sixth); advertising by mail (seventh); magazine (eighth); radio (ninth); and advertising on transit bus (tenth).

The banking behaviour patterns of the older consumers indicated that only one fourth of the respondents frequented their savings and loan branch once or more per week, whereas 44.3% did so once a month, and 25% twice a week or more. Only half of older consumers had the plastic card. Moreover, only 14% of them were familiar with automated teller machines, and less than 7% of them had ever used the automated teller machines.

These findings confirmed that older consumers were reluctant to use the modern banking technology, and the overall attitudes of older savings and loan customers seemed to be somewhat conservative in relation to S&Ls innovations. Also, it can be concluded that older consumers prefer banking services provided to them in their local area, i.e., near their place of living.

Kaynak and Yavas (1986) attempted to extend the application of usage rate of a product / service as a method for segmenting the market. Their study was designed to achieve the following purposes:

1. To determine the importance of selected patronage factors in choosing a commercial bank by frequent and infrequent segments.

2. To determine the perceived usefulness of several services offered by commercial banks to the same groups.

Two categories of variables were used. The first was the dependent variable (frequent users / infrequent users). Thus. respondents were divided into two groups: 58 frequent users and 146 infrequent users. The second category of variables consisted of independent variables divided into two sets. The first set was related to importance criteria. Thus, respondents were asked to rate the level of importance they gave to a series of factors in choosing a commercial bank. The second set of independent variables was designed to generate data concerning the perceived usefulness of several services provided by the commercial banks to their customers.

The comparisons between frequent and infrequent users were made by using t-test statistical method. The findings of the t-test revealed that there were statistically significant differences between frequent and infrequent bank customers on certain attributes. Location near home or work was more important for infrequent customers, whereas available parking space was more important to frequent customers than to infrequent customers.

Also, comparisons were made by using a two-tailed Mann-Whitney U-test regarding the usefulness of services. The findings of U-test revealed that there were statistically significant differences between frequent and infrequent bank customers as regards to rating of the usefulness of selected banking services. The usefulness of savings accounts and savings bonds was rated

higher by infrequent bank customers than by frequent customers. Frequent customers, on the other hand, rated the usefulness of credit cards higher than their infrequent counterparts.

One can conclude from the above findings that: (a) there are a certain differences between frequent and infrequent users in terms of the importance they attach to different bank attributes; and (b) infrequent users appear to be mostly those who can be characterised as savers. By contrast, frequent users appear to be mainly those who can be characterised as spenders.

Although, the study reported here provided good comparison between two types of bank customers (frequent and infrequent), a number of limitations can be mentioned here: (a) the findings did not mention the interaction between demographic variables and other independent variables and how they can affect either type of frequent and infrequent bank customers in their banking behaviour and responses; and (b) the size of the frequent bank customers sample was too small, being only 58, in contrast to that of infrequent bank customers were 146.

Goodfelow (1987) undertook exploratory research to define the savings market in customer or demand-orientated terms, such that customer needs and requirements had primacy. The main objectives were to gain insights into the basic motivations, attitudes and beliefs of savers, and to investigate consumers' perceptions and attitudes toward savings and investments in general.

The research findings regarding consumer perceptions and attitudes towards savings and investments revealed that at a superficial level of analysis, the terms "savings" and "investment" were perceived to define different areas. In terms of reported behaviour and in - depth analysis of the definitions, there was no

consistent and meaningful differentiation between the two terms. The common denominator between savings and investment according to respondents was that they were both terms for money which is not spent and is tied up for varying periods of time.

The findings regarding life - cycle stage and experience confirmed that: (a) stage in the life cycle had been identified as a significant factor in influencing attitudes and behaviour in a variety of product fields; and (b) stage in the life cycle had been found to be of major importance in consumer savings behaviour. Several respondents made reference to the fact that their attitudes towards financial matters had been formulated at an early stage (childhood), and that these had not significantly changed.

In a more advanced stage of research, respondents were divided into groups according to their attitudes towards savings and investments. The findings indicated that the most immediately apparent difference in attitudes was between those whose philosophy was basically to "spend", and those who seek to "save" almost regardless of their circumstances.

However, from the plethora of attitudes, the findings elicited a pattern of five groups, labelled as follows:

1. Financial managers: The major characteristics of this group were a high level of interest and activity in financial matters.

2. Ethical Savers: This group represented the most prudent type who adopted a cautious approach to financial matters. Borrowing was avoided whilst saving was a way of life regardless of financial circumstances.

3. Credit Users: This group might be viewed in certain respects as the polar opposites of the ethical savers in

that they were risk takers and perceived little merit in saving.

4. Finance Avoiders: The major characteristics of this group were a disinterest in money and a desire to minimise their involvement in financial matters.

5. Acceptors: The key characteristic of this final group was that they accepted that a level of involvement in financial matters was an inevitable and essential element in modern society.

The findings of the study reported here provided better understanding for consumer behaviour in two important financial subjects, namely savings and investments. Such an understanding can help many financial organisations in formulating their strategies to gain and keep customers. All the same, the following limitations can be mentioned: (1) The sample size was too small (only 25 respondents were interviewed), since such a sample can not be statistically representative of the population; (2) The findings lack reliability and stability because of the small sample size; (3) It is possible that the sample over - represented the more knowledgeable and active savers; and (4) Because the study was exploratory, no attempt had been made to explain the interrelationship between factors affecting Adding to that, there is the difficulty of consumer behaviour. the beliefs, explaining complex pattern of attitudes. and notivations which influence saving / investment behaviour.

Erol and El-Bdour (1989) investigated attitudes, behaviour and patronage factors of bank customers towards Islamic banks in Jordan. The authors used primary collected from a sample of Islamic bank customers. Percentages were used to analyse the demographic and socio-economic characteristics of respondents. The t-test and factor

analysis were used to analyse customers' attitudes towards Islamic banks in Jordan. The main findings of the study reported here can be summarised in the following: (a) in considering motives responsible for selecting Islamic banks, religious motives did not stand out as being the major significant motive; (b) bank customers were shown to be profit motivated which is in line with Islamic legal doctrines which encourage a fair return on investment; and (c) customers indicated an important consideration to the new branches' role in increasing the utilisation of services provided by Islamic banks. One limitation can be mentioned with regard to the above mentioned study, that is no mention was made to the role of competition between traditional commercial banks and Islamic banks and how different segments of consumers of banking services perceive them. In other words, the role of competition between the newly established Islamic banks and traditional commercial banks has not been mentioned.

Kaynak, Kucukemiroglu and Odabasi (1991) investigated commercial bank selection in Turkey. Their study was based on survey data collected from 250 respondents in Turkey. Multivariate analysis of variance model and factor analysis model were used to analyse the data. The findings indicated that there were significant differences between male and female bank customers in terms of their attitude towards different commercial banks and the services they provide in Turkey. In other words, sex of respondents was a significant factor in influencing consumers bank selection in Turkey. The findings indicated also that there were differences between frequent and infrequent bank customers in Turkey. Age of respondent was significant in discriminating between bank customers in Turkey.

It can be argued that the study mentioned above is related to the present investigation. That is to say that the above mentioned study was carried out in Turkey and the present investigation is carried out in Iraq. Turkish and Iraqi financial markets share many similarities in terms of institutions, consumers, types of services, the level of advancements in banking system and facilities. In other words, the two countries are comparable. As is known and concluded by the authors of the above mentioned study, Turkish bank customers share many similarities with those bank customers in Middle Eastern countries. This can be attributed to cultural, economic, social and consumer behaviour similarities. Therefore, most of the findings of the above mentioned study about Turkey are relatively applicable to the Iraqi bank customers as it will be discussed in the research findings of the present investigation.

Finally, exhibit (3.1) provides a summary for the above mentioned previous research publications which includes the focus of each study, methodology, findings and limitations.

Exhibit 3.1: Summary of Previous Research Publications Related to Consumers' Saving Behaviour.

RESEARCHER	FOCUS OF STUDY	METHODOLOGY	MAIN FINDINGS	LIMITATIONS
Claycamp (1965) The USA	Identifying the differences between savers with Commercial Banks/ Savings and Loan Institutions/ with both.	Discriminant Analysis	Socio-economic variables (except age) were of little value in discriminating between groups. Differences exited between the groups in terms of their psychological characteristics. Savings motives were not important as discriminators.	Later studies indicated that both socio- economic characteristics and motivations were of significant importance in discriminating between different groups. The sample contained only holders of savings accounts.
Martin (1967) The USA	Searching for predictive factors related to savings inflow in S&L institutions.	Regression Analysis	The independent variables no. of competitors, consumers aged 65 and over and population growth were positively related to annual savings inflow. Economic and demographic variables of consumers were positively related to annual savings inflow.	The study did not investigate the influence of consumers' perceptions and motivations on annual savings inflow. The study did not explain the relationship between annual savings inflow and offices' area economic and demographic attributes.
Allaire (1972) Canada	Investigating the reasons behind the selection of a financial institution to deposit at it.	Fercentages	Convenience in terms of location and time was the prime reason for the selection of a bank and received 48.8% of the rate of importance. Other variables such as higher rate of interest, security and familiarity with the bank received also high rates of importance.	The study lacks using statistical methods of data analysis to provide statistically tested results.

(continued)
3.1
Bxhibit

RESEARCHER	FOCUS OF STUDY	METHODOLOGY	SUNDING NIA	LIMITATIONS
Mason & Mayer (1974) The USA	Determining consumers' reasons for maintaining checking accounts with banks and savings accounts with banks, S&LIS and Credit Unions.	Percentages	Convenience of location was the prime reason for maintaining a checking account with a bank. Convenience of location and having a convenience of location and having a checking account at the bank were the prime reason for maintaining a savings account at the bank. Favourable interest on savings was the prime reason for maintaining a savings account at SkIIs and Credit Unions.	The study lacks using statistical methods of data analysis to provide statistically tested results.
Reynolds & Wells (1978) The USA	Investigating the life style of three groups of consumers: non-savers infrequent savers frequent savers.	Percentages	The life style of frequent savers is different in comparison with non-savers and infrequent savers. The importance of life style analysis in providing better understanding for bank consumers' behaviour was underlined.	Little information was given about the life style of non-savers and infrequent savers. The sample consisted of males, i.e., females were excluded from the analysis. The study lacks more reliable statistical methods of analysis.
Arora, Cavusgil & Nevin (1985) The USA	Examining a number of factors which are important in the choice of a financial institution (banks & S&LIS).	Factor Analysis & Factor Congruency	The findings indicated that seven common factors influencing the choice of banks and S&L institutions these are: dependability of institution, convenience, variety of services, accessibility, convenience of non- person banking, availability of loans and interest competitive.	The response rate was very low 31% of the original sample. Factor congruencies were calculated manually due to lack of computer programme for calculating them at that time.

Exhibit 3.1 (cont	inued)			
Goudge	Examining the role	Qualitative	Children and parents attitudes	The research was qualitati
с Стеел	and contribution of market research	research	towards saving and savings accounts were positive and very encouraging.	only on group discussions.
(1986)	had played in the		Incentives provided by the bank to	before and during the disc
The UK	development and		children were warmly welcomed by	No statistical analysis wa
	launch of savings		both parents and children.	out.
	accounts for			
	children.			
Langrehr	Investigating the	Percentages	With regard to product evaluation the	Using the same sample and
(1986)	competition	ય	study concluded that there were	of questions over two peri-
The USA	between banks and	Сћі-вдиате	changes in favour of banks over	disadvantages such as: tim
	S&LIS for savings	test	S&LIS between 1980 to 1983.	costly and can be effected
	and the criteria		With regard to institution evaluation	social and economical chan
	for consumers to		again there were changes in favour of	
	select between		banks over S&LIs between 1980 to 1983.	
	them between 1980		The overall conclusion was that	
	to 1983.		consumers' preference for banks over	
			S&LIS remained unchanged.	
Alessandra,	Providing some	Simple	Older consumers were reluctant to use	The original sample contai
Knierim &	insights into the	ranking of	the modern banking technology and	groups therefore only 11%
Yavas	banking behaviour	importance	their were conservative towards bank	response was achieved. No
(1986)	of consumers in	according to	innovations. Also, they preferred	statistical methods of dat
The USA	terms of their age	percentages	banking services provided to them in	were used.
	focusing on older		their local area.	
	consumers.			
Кулак	Extending the	тье т-теаt	There were significant differences	The overall sample was sma
ى ئە	application of		between frequent and infrequent users	respondents) in particular

Goudge	Examining the role	Qualitative	Children and parents attitudes	The research was gualitative and based
4	and contribution	research	towards saving and savings accounts	only on group discussions.
Green	of market research		were positive and very encouraging.	There was no prepared list of questions
(1986)	had played in the		Incentives provided by the bank to	before and during the discussions.
The UK	development and		children were warmly welcomed by	No statistical analysis was carried
	launch of savings		both parents and children.	out.
	accounts for			
	children.			
Langrehr	Investigating the	Percentages	With regard to product evaluation the	Using the same sample and the same list
(1986)	competition	ধ্য	study concluded that there were	of questions over two periods has some
The USA	between banks and	Chi-square	changes in favour of banks over	disadvantages such as: time consuming,
	S&LIS for savings	test	S&LIS between 1980 to 1983.	costly and can be effected by other
	and the criteria		With regard to institution evaluation	social and economical changes.
	for consumers to		again there were changes in favour of	
	select between		banks over S&LIs between 1980 to 1983.	
	them between 1980		The overall conclusion was that	
	to 1983.		consumers' preference for banks over	
			S&LIS remained unchanged.	
Alessandra,	Providing some	Simple	Older consumers were reluctant to use	The original sample contained all age
Knierim &	insights into the	ranking of	the modern banking technology and	groups therefore only 11% rate of
Yavas	banking behaviour	importance	their were conservative towards bank	response was achieved. No advanced
(1986)	of consumers in	according to	innovations. Also, they preferred	statistical methods of data analysis
The USA	terms of their age	percentages	banking services provided to them in	were used.
	focusing on older		their local area.	
	consumers.			
Kynak	Extending the	тре т-теяt	There were significant differences	The overall sample was small (250
ъ	application of		between frequent and infrequent users	respondents) in particular the sub-
Yavab	usage rate of a		in terms of the importance they	sample of frequent users (only 58
(1986)	service as a		attached to some independent variables	respondents. The findings did not
Салада	method for		related to the location of the bank	mention the interaction between the
	segmenting the		and the services it provides.	demographic variables and the
	market (frequent			dependent variable, i.e., the usage
	vs. infrequent			rate.
	users).			

(continued)
3.1
Exhibit

RESEARCHER	FOCUS OF STUDY	МЕТНОDОLOGY	MAIN FINDINGS	LIMITATIONS
Goodfelow (1987) The UK	Defining the savings market in consumer or demand-oriented terms.	Qualitative research	Respondents viewed "savings" and "investment" as they were both terms for money which is not spent. Life cycle stage was a significant factor in influencing attitudes and behaviour of consumers. The most apparent difference is between two groups of consumers "spenders" and "savers".	Qualitative research based upon small sample not necessarily representative of the target population and lacks the reliability and stability of large- scale quantified research.
Erol and El-Edoui (1989) Jordan	r Investigating attitudes of bank customers toward Islamic banks in Jordan.	Fercentages, s T-test and Factor Analysis	Religious motives did not stand out as being the only major significant motive for selecting an Islamic bank. Bank customers were profit motivated. New branches have important role in increasing the utilisation of services provided by Islamic banks.	No mention was made to the role of competition between traditional commercial banks and the relatively newly established Islamic banks.
Kaynak, Kucukemi- roglu & Odabasi (1991) Turkey	Investigating commercial bank selection in Turkey.	Multivariate Analysis of Variance & Factor Analysis	Sex and age of respondents were significant factors in influencing consumers' bank selection. There were differences between frequent and infrequent bank customers in terms of their attitudes and demographic and socio-economic characteristics.	The size of the sample was not large enough especially because non-response rate and response errors reduced the actual number of participants to answer the questions included in the questionnaire.

3.5 Conclusions:

It has been emphasised in this chapter that one of the most important roles of commercial banks is the role of accumulating and investing savings.

On one hand this role of savings mobilisation is very important for any national economy, while on the other hand, it is a banking service provided by the banks for customers, when they need it. Thus, customers (savers) seek to put their savings in a safe and secure place with an acceptable level of liquidity and profitability.

Bank marketing is the process of creating and delivering satisfactory services to consumers at a profit to the bank. Bank marketing in the 1950s and before was understood by banks to include merely advertising, public relations, and personal selling. By the late 1950s and throughout the 1960s, banks discovered that they had to adopt the marketing concept in order to survive in the increasingly competitive market environment.

The marketing concept for banks means that banks must recognise that bank customers are changing in terms of their wants, needs, desires, expectations, and problems. In other words, banks must adopt customer - oriented rather than product - oriented marketing strategies.

The use of marketing research techniques in banking is relatively new. That is, throughout the 1960s and later on, few banks realised the importance of marketing research in banking. However, by the early 1970s and throughout the 1980s, many banks and market researchers have carried out some marketing research studies to solve some marketing problems in the banking industry.

The application of marketing research in banking has become wider, especially during the 1970s and 1980s. In banking, marketing research is used as a problem solving instrument. However, five major areas have been mentioned widely in bank marketing research: (a) Market segmentation; (b) Customer behaviour and bank selection; (c) Customer loyalty; (d) Bank service; and (e) Bank branch studies. But, more recently and because of the deployment of the computer technology in banking new trends and applications have developed in the use of marketing research in banking. Thus, the use of advanced computerised technology by banks has provided the opportunity to find quantitative answers to different types of marketing problems facing the banking industry.

The following conclusions can be derived from the discussion of the previous similar and comparative marketing research studies reported in this chapter:

1. Most previous studies have confirmed that consumers' socio-economic, demographic, and psychological characteristics have a great impact on consumers' savings behaviour.

2. Consumers' socio-economic, demographic, attitudinal, and psychological characteristics can be used effectively in discriminating between different groups of consumers. For example, they can be used in discriminating between savers and non-savers effectively.

3. In addition to the above mentioned variables, some bank attributes, and quality of service variables, have an important role in influencing consumers' savings behaviour and their perceptions of the service of savings accounts.

4. Modern consumers often seek more than just a good return on their savings. Many studies have confirmed that modern consumers seek convenience in terms of time and place, a good personal relationship with the bank, and availability of modern banking facilities and services. In addition, they also look for a safe and secure place to deposit their savings, liquidity, and easy access to their savings.

5. Little attention has been paid by previous studies to the role of basic personal motivational factors, such as reasons for having savings accounts, that influence consumers' savings decisions and behaviour, especially in the developing countries..

6. A few attempts, if any, have been made to study in a <u>comprehensive</u> way the influence of all socio-economic, demographic, psychological, attitudinal, and motivational factors, taken together, in consumers' savings behaviour, especially in the developing countries.

However, the present study is an attempt to provide better understanding of consumers' savings behaviour by using a more comprehensive approach. Thus, in this investigation an attempt is made to combine all the factors mentioned in item (6) above in investigating consumers' saving behaviour.

Three facets have been defined in this study, as will be discussed extensively in chapter five, in order to investigate consumers' saving behaviour in Iraq:

Facet D1: Consumers' attitudes towards savings accounts.

Facet D2: Bank attributes.

Facet D3: Consumers' financial psychographic characteristics.

In addition there has been an attempt to investigate the role of consumers' profile characteristics, i.e., socio-economic and demographic characteristics, in influencing consumers' saving behaviour in Iraq.

Each of these facets contains a number of sub-facets and elements which have been included in the analysis in order to comprehensively investigate the problem of understanding consumers' savings behaviour (decisions) in Iraq..

In fact, the most important aim in doing this is to explain: (a) What are the factors that influence consumers' savings decisions to save at a commercial bank; (b) What are the relationships between the identified factors, especially the relationship between consumers' profile characteristics (socio-economic and demographic) and other attitudinal factors; (c) What are the differences between consumers who are "banked with savings accounts" and "non-banked" in terms of their attitudes towards the idea of saving, having savings accounts, and other bank attributes; (d) What are the differences between savers and non-savers in terms of their socio-economic and demographic characteristics; and (e) What are the differences between consumers' saving behaviour according to their: age, family size, income, educational level, and saving habits.

CHAPTER FOUR

CONSUMER BEHAVIOUR.

4.1 INTRODUCTION.

4.2 MODELS OF CONSUMER BEHAVIOUR: THE COMPREHENSIVE

MODELS.

4.2.1 THE HOWARD - SHETH MODEL.

4.2.2 THE HOWARD MODEL.

4.2.3 THE ENGEL, KOLLAT, AND BLACKWELL MODEL.

4.2.4 COMPARISON OF THE MODELS.

4.3 FACTORS AFFECTING CONSUMER BANKING BEHAVIOUR.

4.3.1 THE EXTERNAL FACTORS.

4.3.2 THE INTERNAL FACTORS.

4.3.3 THE CONSUMER BUYING PROCESS.

4.4 CONCLUSIONS.

4.1 Introduction:

The main objective of this chapter is to discuss the theoretical aspects of consumer behaviour. Thus, an extensive discussion for the comprehensive models of consumer behaviour is to be presented. Also, the main factors related to consumer behaviour in general, and consumer banking behaviour in particular are to be discussed. Finally, a conclusion is to be made regarding the relationship between consumers' banking behaviour and other models of consumer behaviour.

4.2 Models of Consumer Behaviour: The Comprehensive Models:

In this chapter the attention is turned to the examination of various alternative models of consumer behaviour. However, before examining each model, it is important to highlight some thoughts about the meaning of a model, and its uses in consumer behaviour studies.

Markin (1974) defines consumer behaviour as a host of complex activities and actions, some physical and some mental. In this definition both activities and actions can be represented by consumer decision processes such as: the purchase activity of problem recognition, awareness, attention, search, and evaluation, with final acts of choice such as: acceptance, rejection, or postponement.

Also, this definition acknowledges a concern with postpurchase phenomena such as: cognitive dissonance (difference) and all those activities related to dissonance reduction. Furthermore, this definition refers to all subjective phenomena such as: mood, introspection, attitude, self-theory, and cognitive structures, that

are likely to effect the consumer's choice of goals, decision processes, and solutions.

In regard to the development of the concept of consumer behaviour, Walters (1978) mentions that consumer behaviour has grown up in the era of the computer. Specific and comprehensive models were made available while the subject was still in its relative infancy. Thus, some of the trial and error was avoided. The comprehensive models have given consumer behaviour development a distinct advantage over earlier marketing concepts, and the models, no doubt, have been a significant factor in the rapid development of consumer concepts.

In addition to the above, Engel and Blackwell (1982) indicate that once consumer behaviour began to achieve a measure of sophistication as a discipline around the middle 1960s, some of the pioneering writings on this subject were built upon models of consumer behaviour.

Loudon and Della Bitta (1984) define a model as a simplified representation of reality. It simplifies by incorporating only those aspects of reality that are of interest to the model builder. Other aspects that are not of interest and only add to the complexity of the situation can be ignored.

Based on the above definition of a model, the term consumer model can be described as the representation of reality in a simplified way. Thus, anything used by the model builder to represent the reality, includes only those variables or aspects that are of interest to him. The representation of a consumer model can be: (a) verbal, i.e., a model builder can model something by verbally describing it; (b) algebraical, i.e., a model may be expressed algebraically by representing it with mathematical

symbols; and (c) pictorial, i.e., a model can be represented by using, for example, a map or flow chart.

Whatever the method of model representation used by model builders, a model is used in order to achieve some purposes. Williams (1981) suggests that models fulfil two main purposes. First, they assist in the development of theories of consumer behaviour; and second, they aid understanding of the complex relationship involved.

However, by simplifying consumer behaviour into its principle components and by identifying the relationships between variables, a model serves as a frame of reference that can be used to direct research. Thus, a model can be used to build a systematic and logical thinking about consumers, this can entail as described by Loudon and Della Bitta (1984) the following: (1) identifying the relevant variables; (2) indicating their characteristics; and (3) specifying their inter-relationships, that is, how they influence each other.

4.2.1 The Howard - Sheth Model:

Williams (1981) mentions that the Howard - Sheth model of buyer behaviour is based on a model originally conceived by John Howard, which has been revised and refined with the assistance of Jadish Sheth in 1969. The input stimuli in the model discussed here consist of information obtained from the social and commercial environment, and the output responses are buyer behaviour, including attitudes and motivational sets as explained in figure (4.1).

Williams (1981) also mentions that the Howard - Sheth model is essentially a model of consumers' rational choice decisions. It assumes a degree of rationality on the part of the consumer to the extent that decisions are rational within the limits of consumer's

perceptual, learning, and information-processing capabilities. The consumer in Howard - Sheth model is not assumed to consider all the various alternatives but rather forms an evoked set of brands meeting the criteria necessary to be deemed satisfactory alternatives. The Howard - Sheth model is suitable to explain brand choice decisions. It is less capable of explaining selection decisions between two different products, for example, opening a savings account or buying a life insurance policy.

The model concentrates on the way in which information is perceived and the adaptive behaviour that results through the learning process. Implicit to the model is the assumption that the consumer will always try to simplify his buying decision through the acquisition of more information and through gaining experience. The model does not, however, fall into the trap of assuming that behaviour takes place in a vacuum.

However, in terms of evaluating the Howard - Sheth model Williams (1981) has emphasized that the model has attracted a great deal of interest, not least because it embraces the principle determinants of behaviour; perception; communication; learning; and attitudes. It does, however, suffer from a number of shortcomings and these stem mainly from the fact that the variables and the relationships between them are not well defined. The distinctions between the endogenous (internal) variables and the exogenous (external) variables, for example, are not sharp, and empirical evidence has suggested that some of the exogenous variables have as much influence on the outcome as some of the endogenous variables. However, for more details about the Howard - Sheth model see, for example: Markin (1974), Walters (1978) and Engel and Blackwell (1982).

4.2.2 The Howard Model:

The Howard model is a revised version of the original Howard - Sheth model. The revision of the model had been done by John Howard in 1977. Figure 4.2 shows the revised Howard model. The main difference between the two models as has been described by Engel and Blackwell (1982) is that the dotted lines, as shown in figure 4.2, are designated as feedback, meaning that a variable appearing later in the process "feedback" and affects an earlier variable in a functional way.

However, Engel and Blackwell (1982) concluded that in the essence of what has been mentioned above, the original Howard -Sheth model and Howard model are very much alike.

4.2.3 The Engel, Kollat, and Blackwell Model:

The Engel, Kollat, and Blackwell model of consumer behaviour had been first offered in 1968. As is described in figure 4.3, the model contains five components: (1) input stimuli; (2) information processing; (3) a decision process; (4) decision-process variables; and (5) external influences.

Loudon and Della Bitta (1984) mention that the form of these components and how they relate to each other varies depending on the degree of involvement that the consumer is experiencing in a given purchase situation. In the high-involvement version, as shown in figure 4.3, the core component is the decision-process, which includes five basic stages: (1) problem recognition; (2) information search; (3) alternative evaluation; (4) choice; and (5) outcomes.

Engel and Blackwell (1982) explained that their fourth revision for the model had several distinct purposes:



Figure 4.1: The Howard - Sheth Model, (1974 version).
Source: J.U. Farley, A. Howard, and L.W. Ring, Consumer Behaviour:
Theory and Application, Allyn and Bacon, 1974, p. 11.

.

.



Figure 4.2: The Howard Model.

Source: J. A. Howard, Consumer Behaviour: Application of Theory,

New York: McGraw - Hill, 1977, p. 133.


Figure 4.3: The Complete EKB Model of High-Involvement Decision Process. Source: J. F. Engel and R. D. Blackwell, Consumer Behaviour,

Fourth Edition, The Dryden Press, Japan, 1982, p. 500.

1. To highlight more clearly the inter-relationship between stages in the decision process and the various endogenous and exogenous variables.

2. To clarify the relationship between attitudes and behaviour to reflect the contributions of the Fishbein extended model. Thus, beliefs and intentions were introduced as explicit variables for the first time as was normative compliance.

3. To define the variables with greater precision and to specify functional relationships to permit empirical testing.

However, for more details about the EKB model see, for example: Walters (1978), Williams (1981), and Hawkins, Best and Coney (1986).

4.2.4 Comparison of the Models:

The Howard - Sheth, and Howard model are very comprehensive. However, the Howard - Sheth model has been described by Walters (1978) as a very precise construction with the variables clearly connected in their major interactions. It points emphatically to the great variety of inputs that affect consumer behaviour. In its construction the authors of the model have done an excellent job of drawing from business, psychology, sociology, management, etc., and of bringing these areas into the concept of consumer behaviour.

The Howard - Sheth model has its problems as had been stated by Walters (1978): (1) the very complexity of the model causes problems with quantification; (2) the interaction of the inputs, stimulus ambiguity, and perceptual bias are inadequately explained;

(3) the redundancy in learning constructs and outputs which can lead to confusion; and (4) variables such as: needs, communications, and decision processes are not emphasized in the model.

While, the EKB model, as had been described by Engel and Blackwell (1982), is unique in highlighting the decision process and explicity including the proven relationships of the Fishbein behavioural intentions models under alternative evaluation.

However, Walters (1978) suggests that the EKB model has several advantages for experimentation: First, it combines relative completeness with a logical ordering of information. Second, it emphasizes decision making more than other theoretical models. Third, it is the easiest to follow in terms of its revised form. Fourth, it does a better job of emphasizing how relationships occur.

The EKB model has its deficiencies as had been outlined by Walters (1978): First, important internal variables such as: needs and motivations are not specified. Second, it is the most mechanical of all experimental models. Third, it is not clear how the environment is related to the central control unit and to the stimuli. Finally, as mentioned by Engel and Blackwell (1982), confidence in ability to judge the product has not been included in the EKB formulation.

In general, a close examination of the above mentioned models reveals that there are some similarities and dissimilarities between them. Also, all the models are primarily experimental and designed for specific reasons.

In addition, as it has been described by Engel and Blackwell (1982), the EKB and Howard models share a general agreement on the functioning of memory. The EKB model, however, goes into far more

detail on specifying the steps which occur between the stages of exposure and retention.

Finally, in addition to the above discussed models of consumer behaviour, there are a number of other less known models of consumer behaviour such as: The Nicosian Model (see Walters, 1978); The Andreason Model (see Markin, 1974), and The Holocentric Model (see Markin, 1974 and Walters, 1978).

4.3 Factors Affecting Consumer Banking Behaviour:

The above mentioned models of consumer behaviour have indicated that consumer behaviour is subject to influence by several types of decision process variables some of which are external and others are internal.

However, consumer behaviour in banking is subject to the same influences. In other words, consumers' banking behaviour can be influenced by a number of internal and external factors. Evans and Beckman (1974) defined three sets of factors that might influence consumer banking behaviour, these factors are the following: (1) external factors; (2) internal factors; and (3) the consumer process.

4.3.1 The External Factors:

Meidan (1984) indicates that there are a number of external factors that might influence consumer banking behaviour. These can be divided into: (1) cultural; and (2) social factors.

1. The cultural factors:

Consumers are likely to be influenced by their own culture, i.e., the social culture they have grown up in. Thus, consumers'

needs, wants, perceptions, and desire can be affected by their own cultural characteristics.

Therefore, many banks have considered this fact, and have based some of their operations on cultural factors. For example, national banks have attempted to open branches in countries where there is a large number of immigrants of their origins. In fact, in many countries such branches are very successful in attracting the targeted segments of consumers, i.e., consumers whose nationality is similar to the bank's nationality.

(2) The social factors:

In their empirical investigation Tan and Chua (1986) concluded that in an Oriental culture, where people are still strongly rooted in traditional cultural values and family ties, social influences in the form of friends' and family's advice in buying are a dominant factor in bank selection.

However, the influence of social factors is not limited to an Oriental culture only. Thus, in many Western societies and others consumers can be influenced by social factors, but, the importance of each of these factors may differ from one society to another. Joy, Kim and Laroche (1991) indicate that the topic of ethnicity has emerged as an important construct in the study of consumer behaviour. The authors carried out a study to investigate the relationship between English-French Canadian ethnicity and the use (ownership) of financial services. The findings of their study confirmed that the predictor variable English-French Canadian ethnicity has an influence on the ownership of financial services in Canada. Thus, the findings of their study confirmed that there were differences in the use of financial services between the English-Canadian ethnicity and the French-Canadian ethnicity. In fact, the

authors concluded that ethnicity as a factor has a greater importance than factors such as income, family life cycle and family size in explaining the use (ownership) rate of financial instruments in Canada.

Meidan (1984) mentioned the following types of social factors:

1. Reference groups: The most significant factor under this category of influences is the consumer's need to relate to others. Thus, consumers are likely to be influenced by the opinions of their reference groups when, for example, they want to select a bank. In fact, in many cases the effect of the reference group on consumer behaviour is stronger than the effect, for example, of a bank advertisement shown on Television.

2. Family: The family of the consumer has a strong effect in determining his / her behaviour. For example, many consumers based their decision to select a bank on advice from family members. In other words, consumer's banking behaviour can be influenced strongly by this social factor. In many cases if one member of the family deals with a particular bank, other member of the same family will do the same. In banking, the notion "bank of the family" is wellestablished in many societies.

4.3.2 The Internal Factors:

Two types of factors can be classified as internal factors, these are the following:

- 1. The psychological factors: The psychological factors can be divided into two groups:
- 1. Motivation and Perception:

The behavioural element of motive is a key part of consumer's basic psychological make-up. Engel and Blackwell (1982) mention that basically motives are behavioural dispositions to act towards goal satisfactory objects. They are learned from past experience and serve to direct purchase or patronage decisions of individuals towards behavioural patterns that are perceived as satisfactory.

When consumers open, for example, a savings account at a particular bank they seek to satisfy some motivations. However, consumers can be motivated to act by different motives. For example, in the case of consumers' saving behaviour, some savers are motivated by higher rate of interest on their savings. While, others can be motivated by some other advantages they may gain by opening a savings account, such as keeping surplus money in a safe place for future use.

It can be said that each individual has a unique set of perceptions towards a particular, for example, bank or service. Pezzullo (1989) defines perception as the process by which people receive information (or stimuli) through the five senses, recognise it, and assign a meaning to it. How people perceive information depends on the nature of the information itself, on its context (that is, what surrounds it), and on conditions within themselves.

For example, consumers perceive banks as profit-making organisations, concerned about their interests and dealing with rich people. To the extent that in some cases, some lower class consumers prefer not to deal with banks, because, they feel that these institutions are established to serve rich people only. Taking that on board, many banks have started to change their image, in other words, attempting to change consumers perceptions of banks. They are doing this by telling different segments of consumers that they

have something to offer them, and that they are important to the bank as customers. In other words, banks attempt to give the impression that they are open for every body, and all segments of society.

Consumers' buying behaviour in banking can be influenced or motivated by the relationship between the consumer and the bank. Kimball (1990) emphasises that the conflict between relationship and product, and the effect on consumer buying behaviour, is likely to be the most important marketing issue of the decade. Thus, in order to possess operational content by banks, their strategy must clearly define the target customer and how the bank will satisfy that customer's needs. The same author also mentions that market research has shown that a substantial proportion of consumers purchase multiple banking products from an institution they consider to be their primary bank. That is to say that when considering the purchase of an additional product, relationship-oriented customers first approach their primary bank to determine availability and price, thereby giving it the "right of first refusal". This buying behaviour on the part of relationship-oriented segment gives the primary bank an advantage in cross-selling additional products and winning a larger share of customers' total banking business. Kimball also mentions that market research has shown a high correlation between the bank where consumers keep their principle transactions account (e.g., checking account) and the one they designate as their primary bank. Therefore, banks adopted a relationship strategy focus on understanding their customers' banking behaviour and profiling them into different groups (segments) in order to satisfy the each different segment financial needs of and keep good relationship with them.

2. Attitude, Belief, and Intention:

Peter and Olson (1987) emphasise that today most researchers agree that the simple concept of attitude proposed by Thurstone and Fishbein is the most useful. That is, attitude represents person's favourable or unfavourable feelings toward the object in question. Beliefs (cognition) and intentions to behave (conation) are seen as related to attitude, but are separate concepts, not part of attitude itself.

However, attempts have been made by a number of scholars to develop a model for predicting consumers attitudes towards an object. In the following a summary discussion for the major Multi-Attribute Models of Consumer Attitude:

The Rosenberg Model:

Engel and Blackwell (1982) mentioned that the Rosenberg's model as originally published contained two variables: (1) values (equivalent to evaluative criteria) and their importance in arriving an attitude; and (2) perceived instrumentality (a complex term that simply estimates the degree to which the taking of a point of view or following an action will either enhance or block the attainment of a value).

The Fishbein Model:

Peter and Olson (1987) stated that the key proposition in Fishbein's theory is that the evaluation of the salient beliefs cause overall attitude. Simply stated, people tend to like objects they associate with "good" characteristics and dislike objects they belive to have "bad" ones. In Fishbein's multi-attribute model, overall attitude toward an object is a function of two factors: (1) the strengths of the salient beliefs associated with the object; and (2) the evaluations of those beliefs.

However, most of the marketing applications, or attempts to investigate consumer attitudes towards an object did not follow Fishbein or Rosenberg models, but, they use modified hybrid models in investigating consumer attitude.

Most of marketing attempts to investigate consumer behaviour have dealt with three concepts which are related to consumer attitude towards an object, these are:

(a) Cognition, which refers to the person's knowledge or belief about the object. For example, consumers attitudes towards savings accounts can be affected by the knowledge they have about this bank service and its advantages;
(b) Feeling, which refers to a person's favourable or unfavourable feelings towards the object. For example, consumers attitudes towards savings accounts can be affected by their feelings towards the idea of having a savings account, and their feelings about banks in general; and

(c) Conation, which refers to person's intention or actual behaviour towards the object. For example, person's decision to have or not to have a savings account is based on his / her intention.

2. Personal Factors:

Personal factors are the second set of internal factors, that have influences on consumer behaviour. In banking, however, the personal factors have proved to be of great importance in influencing consumer' banking behaviour.

Personal factors can be divided into three main types, these are: (1) Socio-economic, such as: income, educational level, and occupation; (2) Demographic, such as: sex, marital status, age,

family size, and life cycle stage; and (3) Other personality factors, such as: lifestyle, personality and self concepts.

4.3.3 The Consumer Buying Process:

The consumer buying process has been described by Meidan (1984) as the most critical part of understanding consumer behaviour in banking.

The consumer buying process is developed usually in stages. Pezzullo (1989) defined four main stages for consumer buying process, these are: (1) Unsatisfied need; (2) Purchase activity, which can be divided into: (a) Search for alternatives, and (b) Evaluation of alternatives; (3) Purchase decision; and (4) Postpurchase feelings.

While, Meidan (1984) divided the consumer buying process in banking into nine stages, these are: (1) Cognitive balance; (2) Cognitive imbalance; (3) Awareness; (4) Information Search; (5) Knowledge, (6) Evaluation behaviour; (7) Preference; (8) Decision; and (9) Post-purchase decision.

The two above mentioned models of consumer buying process refer to the fact that consumer buying process starts with unsatisfied need, and ends with purchase and post-purchase decisions. However, between the starting point and the end the consumer seeks information about the available alternatives in order to make an evaluation, before making his / her decision to buy a particular product or service.

However, in regard to consumers' saving behaviour, consumers' decisions to open a savings account can be summarized to consist of the following stages as shown in figure 4.4: (1) Consumers desire to save; (2) Searching for information about alternatives; (3)

Evaluating the available alternatives; (4) Making the decision by choosing a bank and an account; (5) Re-evaluating the decision in terms of satisfaction or dissatisfaction; and (6) Making the decision of keeping or changing the bank or the type of savings account.



Figure 4.4 Stages of Consumers' Savings Decisions

One of the main objectives of this study is to establish a method of investigating consumers' savings behaviour. Therefore, based on the discussions of consumer behaviour mentioned in this chapter, this study's method of investigating consumers' savings behaviour is shown in figures 4.4 and 4.5.

Figure 4.5 shows that consumer's saving behaviour is a product of a number of factors. These factors can be: (a) internal,

such as: attitude, perception, belief, socio-economic, and demographic; and (b) external, such as: the influence of family, friends, and culture. Then, due to the influences of these factors consumer tends to take his / her decision, i.e., either to be saver or non-saver.

The method used in this study is based on investigating those internal and external factors that influenced savers decisions to open savings accounts. Then, investigating the differences between savers, i.e., banked with savings accounts, and non-savers, i.e., non-banked, in terms of their attitudes, psychographic, socioeconomic, and demographic characteristics.

The main conclusions of such a method in investigating the differences between savers and non-savers can be used to provide important explanations for consumers' saving behaviour. Also, they can be used by banks in working out how to influence and change consumers' saving behaviour. In other words, what should be done by banks to keep consumers who are banked with savings accounts, and influence non-banked to be savers, if that is possible.



Figure 4.5: A method to investigate consumers' saving decisions.

Consumer Behaviour vs. Consumer banking Behaviour

Consumer behaviour is a complex activities and actions, some of them are physical and others are mental. Consumer banking behaviour is a part of consumer behaviour. Consumer decision to select a bank and to use a banking service can be influenced by sets of variables some are internal and others are external, as mentioned earlier in this chapter.

It can be suggested, therefore, that consumers behaviour with regard to banking is similar to their behaviour with regard to any other type of services or goods they use or buy. The only obvious difference here is the nature of banking services itself.

In other words, general models of consumer behaviour did not contain the aspects of bank and banking services attributes in their context. The attempt is made here to develop a model of consumer banking behaviour which contains the banking aspects as well as other consumer behaviour variables.

Banking services are intangible and distinct. Consumers of banking services tend to attache more importance to banking matters because they are related to their present and future financial interests. It is natural, therefore, to find consumers of banking services are very concerned about satisfying their want and needs when they deal with a bank.

Most models of consumer behaviour and methods of bank marketing research were developed in the Western Developed World. In other words, they are based on the specific environment where they were developed. In this study, the attempt is made to use these models and methods as a base to develop a model and a method to investigate consumer banking behaviour in one of the less developed countries, i.e., Iraq. In order to achieve that figures 4.4, 4.5 and 5.1 were developed further in order to design this study's model of investigating consumer banking behaviour in Iraq.

Figure 4.6 explains this research's model of investigating consumer banking behaviour in Iraq. It shows that a consumer decision to select a bank and a service is based on two sets of factors, external and internal. The external factors consist of aspects of bank attributes, such as bank's image, relationship with customers, advertisement, location, services, personnel, management and regulations.



Figure 4.5: A model of consumer behaviour in banking

Also, the external factors consist of: (a) the role of competition in the market between banks and other financial institutions; (b) social and cultural factors, such as the influence of family, friends and reference groups on consumers; and (c) government regulations. Government regulations in Iraq have a significant importance in influencing the banking activities of banks and consumers because the Iraqi banking system is state-owned and subject to direct intervention and control by the government, as explained in chapter two of this study.

The internal factors represent consumer's motives, attitudes, intentions, belief and three sets of consumer's characteristics, these are financial psychographic, socio-economic and demographic characteristics. Finally, this study's model of consumer banking behaviour, as presented in figure 4.6, explains that the consumer is the most important and central component of the model because he / she is the decision maker.

4.4 Conclusions:

In this chapter the theoretical aspects of consumer behaviour have been discussed. It has been concluded that consumer behaviour represents complex activities and actions, some physical and some mental.

Since the mid 1960s, especially in the era of the computer, consumer behaviour has grown up, and some specific and comprehensive models of consumer behaviour have been developed.

A model is a simplified representation of reality. Thus, a consumer behaviour model is the representation of reality by the model builder in a simplified way. This representation of reality,

in other word the model, can be verbal, mathematical, or pictorial, i.e., in maps or chart flows.

However a number of consumer behaviour researchers have developed a number of models for consumer behaviour. The most relevant models to this investigation are the comprehensive models of consumer behaviour. Therefore, three major comprehensive models of consumer behaviour have been discussed in this chapter. From the discussion of the comprehensive models of consumer behaviour the following conclusions can be made:

1. Each model contains three broad variables, these are the individual, the firm, and the environment.

2. All the models emphasize that the consumer is the central component of the model, because he is the decision-maker.

3. Through examining the major models of consumer behaviour, it can be said that consumer behaviour is a highly complex subject, which requires a great deal of investigation.

4. All the models are primarily experimental and designed for specific reasons, and they have not been tested through empirical investigations.

5. No model is a complete work, thus, models are subject to criticisms, modifications, and developments.

Consumer buying behaviour in banking, as one part of consumer behaviour, can be related to the comprehensive models of consumer behaviour. Thus, it has been concluded that consumer behaviour in banking can influenced by three sets of factors, these are: (a) external; (b) internal; and (c) consumer buying process. However, all these factors combined have an important role in influencing

consumers' banking behaviour in terms of their selection of a bank, usage of a service, and other dealings with financial institutions.

Finally, the main objective of this chapter was to provide the theoretical background to understand this study's method of investigating consumers' saving behaviour. In other words, the objective was to develop a theoretical model, as presented in figures 4.4, 4.5, 4.6 and 5.1 to guide the empirical part of this research.

CHAPTER FIVE

THE FACETED DESIGN FOR THE EMPIRICAL STUDY

5.1 INTRODUCTION.

5.2 THE FACETED APPROACH FOR THE RESEARCH DESIGN.

5.3 THE MAPPING SENTENCE OF THE DEFINITION BY FACETS OF THE RESEARCH PROBLEM.

5.4 THE CONTEXT OF CONSUMERS' SAVING BEHAVIOUR IN IRAQ.

5.4.1 CONSUMERS' ATTITUDES TOWARDS SAVINGS ACCOUNTS.

5.4.2 BANK ATTRIBUTES.

5.4.3 CONSUMERS' FINANCIAL PSYCHOGRAPHIC CHARACTERISTICS.

5.5 CONCLUSIONS.

5.1 Introduction:

The main objective of this chapter is to describe the research design, which has been adopted for this study. Kinnear and Taylor (1987) defined research design as the basis plan which guides the data collection, and analysis phases of the research project. It is the framework which specifies the type of information to be collected, the sources of data, and the data collection procedures.

On the other hand, Simon (1969) suggested that there is no standard or idealised research design to guide the researcher, since many different designs may accomplish the same objective. In other words, a researcher may decide to adopt a particular research design which he/she may think is the most appropriate to his/her study.

Therefore, a Faceted Design has been adopted in this investigation as the appropriate method of research design to guide this investigation. Therefore, this chapter reviews in brief the theory of Facet Design, and how it is used in approaching this research problem. In addition, it covers how the Faceted Design is applied in establishing the mapping sentence for the definition by facets and sub-facets of the research problem, and how it is used in exploring the extent of consumers' saving behaviour, and consumers' responses to the service of savings accounts and other bank attributes, which are the main focus of this research and its problem.

5.2 The Faceled Approach For The Research Design:

Applying the faceted approach in research design means in general dealing with the research problem in a systematic manner. A faceted approach provides the opportunity to investigate the research problem in a more scientific way by establishing the structural framework for the research.

Consumer researches in marketing mainly deal with attitudes, opinions, and other behavioural aspects. Such sets of ordinal scaled variables require measures, and they can usually be obtained from a sample or samples of population (respondents), through specific survey instruments, such as: interviews and / or questionnaires.

So, the main purpose of Facet Design has been defined by Guttman (1959) and Meidan (1975, A) as it tries to systemise a structure which identifies the different elements and variables of which the phenomenon to be studied is composed and by which it is shaped. Also, the same author outlined the problem which to be dealt with by applying Facet Design in the following:

(a) laying out the complete domain (i.e., universe of content) of the research, its limits and sub-parts;

(b) "sampling" systematically the universe of content of analysis; and

(c) analysing systematically the items sampled.

The pioneers of the concept of facet design are Guttman (1954) and his colleague Foa (1965), who were concerned with the selection of items for test construction and with the weakness in factor analytic procedures.

The faceted design as developed by Guttman, and as applied to the study of various phenomena by Foa is based on Fisher's approach to the design of experiments. This has been followed under the same principle (Fisher's approach) by many researchers, who undertook several behavioural studies in Psychology, Sociology, and other social sciences topics, such as the studies of: L. Guttman (1959), D. Elizur and L. Guttman (1967), P. Runkel and J. McGarth (1972), R. Payne (1976), I. Borg (1982), D. Canter (1982) and (1983), and R. Stone (1982).

In marketing research, in particular, one of the pioneer attempts to employ a faceted approach to investigate a particular marketing problem has been made by Meidan (1974). Then, followed by a limited number of attempts carried out by a few researchers to investigate various types of marketing research topics, such as: Moutinho (1982) applied the faceted design in his investigation of vacation tourist behaviour in Portugal. Harby (1985) used the faceted approach in investigating consumers' life insurance purchasing decisions in Egypt. Sayadian (1989) used the faceted design in a comparative analysis of marketing life insurance policies in the UK. Edris (1989) applied the faceted design in investigating the use of food nutrition labelling by consumers in the UK.

However, previous attempts conclude that the adoption of the faceted approach provides several advantages for the construction of the research and interpretation of its findings.

Canter (1983) said that a facet is a distinct way of classifying the universe of observations. Thus, one of the main advantages of the faceted approach lies on the formalisation of the process of defining the components of the problem, and testing whether a particular facet design produces similarity patterns which are confirmed by the empirical findings.

Also, Canter (1982) mentioned that facet theory utilises three important constituents of a scientific activity:

(a) formal definition of the variables being studied;

(b) hypothesis(es) of some specified relationships between the definition and aspects of empirical observations; and(c) a rationale for the correspondence between (a) and (b).

Moreover, Meidan (1975, A) stated six advantages that can be gained by applying facet design in marketing research for the researcher and his research:

1. The whole possible universe of content of the research area is systematically presented.

2. The sample of items included into the chosen research project are logically selected.

3. The other items, and areas, left out and subject for further investigation are defined and specified.

4. The limitations of a research, because of area and (subject) self delimitations, are supposed to be known in advance.

5. The questionnaire is much easier to structure by putting each item (or element) of the facets into a question.

6. Finally, the analysis of the data collected by the questionnaire could be more systematic since the approximate relations between the items (variables) are known from the faceted analysis of the universe of content of the research.

However, the application of the faceted design in this investigation is based on the above mentioned theoretical background of facet design in terms of its definition, application, and advantages. The universe of content of the faceted design in this investigation represents consumers' attitudes towards commercial banking services in Iraq - as expressed by consumers' attitudes towards savings accounts and other bank attributes. In other words, the chief objective of this study is to investigate consumers' saving behaviour in Iraq. Therefore, the universe of content of this study represents consumers' saving behaviour in Iraq.

Thus, in order to explore the relevant facets to reach this aim, the following steps have been followed: (1) the relevant literature was surveyed; (2) responses are obtained from respondents through self-administered questionnaires, which contain all the variables under investigation; and (3) the variables whose interrelationship is hypothesized are introduced as elements of at least one of the facets.

So, based on the above assumptions and procedures three facets were generated to describe the content of consumers' saving behaviour in Iraq, with special reference to the service of savings accounts provided by commercial banks in Iraq, which are:

Facet D1: Consumers' attitudes towards savings accounts.

Facet D2: Bank attributes.

Facet D3: Consumers' financial psychographic characteristics.

5.3 The Mapping Sentence of the Definition by Facets

of the Research Problem:

Shye (1978) defines a mapping sentence as "a verbal statement of the domain and the range of a mapping including connectives between facets as in ordinary language". It is also, a method of sampling the universe of content of the research problem to match the number of questions, i.e., variables / elements / items, to be answered by respondents.

Meidan (1975, A) explained that the faceted formula for a project can be written in a general form:

$$P \div Q = R$$

Where:

P = Population being studied;

Q = The direct product of different facets: A, B, C..., where

each of these facets are composed of elements:

al, a2, a3 ..., and b1, b2, b3; and

R = The set of possible responses or results.

So, for example, in this investigation the direct product of different facets, i.e., (Q) of the above equation is represented by facets (parameters): A, B, C, D, and E. Therefore, the total elements which represent (Q) will be equal to the number of combinations of all elements in the five facets or (parameters):

i.e., A * B * C * D * E = Q

3 * 5 * 3 * 65 * 1 = 2925 (see Figure 5.1)

In other words, it means the researcher should ask (2925) questions in order to clarify the content of all of the universe presented in Figure (5.1). However, a mapping sentence of the universe of content of the research problem is useful in this instance to reach a systematic limitation for the universe of content to be researched. Thus, by using the faceted design it was made possible to systematically limit the number of questions to be asked in this study's questionnaire to a more relevant and manageable set of questions.

Meidan (1975, A) in his attempt to explain that mentioned that the delimitation of the universe of content of the research is: (a) possible; and (b) necessary because the whole content is presented in the mapping sentence. Moreover, the same author stated that such a delimitation (sampling) of the content of the research should reflect the interest of the researcher, the limitations on the resources, and time at his disposal, the formal title of his investigation and the character of the main hypotheses of the research.

Figure 5.1: The Mapping Sentence for the Definition by Facets of Investigating Consumers' Saving Behaviour in Iraq.

	F: Range High
as <u>C: Tine</u> with perceived <u>CI present</u> regard by the <u>C2 future</u> to the referent <u>C3 past</u> aspects (respondent) at of	E: Objective el Investigating consumers' saving behaviour (decisions) with special reference to consumers' attitudes towards the idea of saving and having savings accounts at connercial (retail) banks in Iraq.
B: Referent <u>(Actor)</u> bl Individual consumer: b2 Bank's management b3 Bank's personnel b4 Government b5 Competitors	ccounts. nt. hip. racteristics.
A: Rttitudes (Behaviour) al Cognition (belief / knowledge) a2 Emotion (feeling) a3 Conation (intention)	D: Content of Consumers' Saving Behaviour D1 Consumers' attitudes towards savings a D1 Benefits of having a savings accoun D16 Reasons for having a savings accoun D16 Attributes of savings accounts. D16 Access to savings. D28 Bank attributes. D28 Bank advertising. D28 Bank Incation variables. D28 Bank Competition variables. D28 Bank Location variables. D28 Bank Location variables. D28 Bank Location variables. D28 Bank Lopalty variables. D28 Bank Lopalty variables. D28 Bank Lopalty variables. D29 Bank Lopalty variables. D38 The influence of reference groups. D38 Personal financial psychographic chan

R Study of

The mapping sentence for the faceted design for this investigation can be summarised as follows:

1. The condensation of the mapping sentence:

The mapping sentence for the faceted design for this investigation can be summarised as follows:

The <u>Attitude</u> (A) of the <u>Referent</u> (B) as perceived by the referent at <u>Time</u> (C) towards the aspects of <u>Content</u> (D) for the main <u>Objective</u> (E) where the range (F) has a <u>Range</u> (R).

Thus, the domain of content under investigation is represented by the first five facets (i.e. A, B, C, D, and E). Facet (F) is the range of responses in terms of how each respondent expressed his / her attitudes in response to the survey questions, which are measured by 5-point Likert Scale (see Figure 5.1). 2. The resources for designing the mapping sentence:

In order to design the mapping sentence for this study's facet design, four resources were used, these are:

 Utilisation of the literature review, i.e., previous studies in bank marketing;

2. Pre-testing interviews with a sample of consumers conducted in Iraq, before starting the actual data collection procedures;

Observation for the variables that may influence consumers' saving behaviour (decisions) in Iraq; and
 The research hypotheses, thus, the interrelationships between variables that have been hypothesized mean that each variable should be an element of at least one facet.

In the following a description for each facet included in the mapping sentence of this investigation, as described in figure 5.1, is provided:

Facet (A) Attitudes:

As has been mentioned in chapter four of this study Peter and Olson (1987), in their review of attitude definitions, conclude that most definitions of attitude have one thing in common that: they refer to the person's effective reaction to, or overall evaluation of, an object or concept. Thus, today, most researchers agree that three concepts are related to attitude which are:

(a) Cognition, which refers to the person's knowledge about the object;

(b) Feeling, which refers to a person's favourable or unfavourable feelings toward the object; and

(c) Conation, which refers to person's intention or actual behaviour toward the object.

These three related components to attitude are constructed Facet (A) of this research, as follows (see also Figure 5.1):

- al Cognition (Belief).
- a2 Feeling (Emotion).
- a3 Conation (Intention).

Facet (B) Referents:

Facet (B) refers to the possible referents or actors, whose attitudes toward the variables under investigation is measured, and the possible referents in this investigation are:

bl Consumers (individuals).

b2 Bank's management.

- b3 Bank's personnel (employees).
- b4 Government.
- b5 Competitors.

Facet (C) the Time:

Facet (C) represents the time at which the referent's attitudes are referred, i.e.:

- cl Present.
- c2 Past.
- c3 Future.

Facet (D) Context of Consumers' Saving Behaviour in Iraq:

Facet (D) refers to consumers' attitudes towards the concept of saving in general, and savings accounts in particular. Together with, other bank attributes might influence consumers' saving behaviour, with reference to other personal financial psychographic characteristics, which are expected to effect consumers' saving behaviour. So, this facet contains the following:

- D1 Consumers' attitude towards savings accounts.
- D2 Bank attributes.
- D3 Consumers' financial psychographic characteristics.

Facet (E) Research Objectives:

Facet (E) in this study refers to this study's chief objective of investigating consumers' saving behaviour (decisions) in Iraq.

5.4 The Context of Consumers' Saving Behaviour:

This section illustrates in detail the main facets, subfacets, and elements, which structured the context of consumers' saving behaviour in Iraq, i.e., Facet (D).

The formulation of the content of Facet (D), as mentioned earlier in this chapter, was based on the review of the literature related to bank marketing, and on a number of preliminary interviews with individual consumers.

Thus, the context of consumers' saving behaviour in this study, as shown in table 5.1, contained the following facets, subfacets, and elements:

Facet D1 : Consumers' attitudes towards savings accounts:

This facet included a number of sub-facets, each of them consisted of a number of elements (items), that are related to consumers' attitudes towards the idea of having a savings account. However, the sub-facets, and elements, which consisted this facet are the following:

Sub-facet Dla: Benefits of having a savings account:

This sub-facet contained nine elements, these are:

- 1. Savings help to meet increasing living expenses.
- 2. Savings help to beat inflation.
- Savings help to make sure family is protected in case of emergencies.
- 4. Savings help to build up a retirement fund.
- 5. Savings help to increase the overall standard of living.
- 6. Savings help in buying property for the family.
- 7. Savings are an investment.
- 8. Savings help to start a new business.

9. Savings help to cover marriage expenses.

The above mentioned nine elements, which were included in this sub-facet, represent a number of benefits (advantages), which can be gained by consumers from having a savings account in Iraq. Sub-facet D1b: Reasons for having a savings account:

Three elements were included in this sub-facet, these are:

1. I feel my money is safe.

- 2. A savings account is a good method to keep surplus of money.
- 3. Own decision (personally motivated).

The above mentioned reasons for having a savings account represent some possible reasons, which may motivate consumers to open a savings account in Iraq.

Sub-facet Dlc: Attributes of savings accounts:

This sub-facet consists of five elements, these are:

1. Convenient hours of operation for savers.

- 2. Special lines for savings accounts holders.
- 3. Accuracy of bank transactions.
- 4. Confidentiality of bank transactions.

5. Easily of processing bank transactions.

The above mentioned five elements were included in this subfacet because they represent some issues that are related to savings accounts attributes in terms of convenience, accuracy, and confidentiality of savings accounts transactions in Iraq.

Sub-facet Dld: Aspects of waiting time:

This sub-facet was represented by two elements, these are:

- 1. Length of time required to process transactions.
- 2. Length of waiting time before being served.

The above mentioned two elements were used in this study, because they are related to the time factor, which is very important

for consumers of banking services. Thus, a bank's ability to serve consumers in the shortest possible waiting time plays an important role in making the customer satisfied. In other words, consumers of banking services now want fast services and feel very unhappy with any delay when they visit their banks.

Sub-facet Dle: Access to savings:

This sub-facet consisted of two elements, these are:

1. I can easily deposit savings at the bank.

2. I can easily withdraw money from the bank.

These two elements were included in this study, because consumers' access to their savings accounts, in terms of depositing and withdrawing money, represents a crucial factor, that may influence their decisions to deal with a particular bank, and open a savings account at it.

Facet D2 : Bank attributes:

Consumers' decisions to select a particular bank, in order to open, for example, a savings account at it, can be influenced by some bank attributes (characteristics). However, in this study the facet bank attributes consisted of seven sub-facets, and each of those sub-facets included a number of elements, and each of those elements represents a particular dimension of bank attributes. Thus, the seven sub-facets, which were included in this facet, with the elements they consist of, are as follows:

Sub-facet D2a: Aspects of bank-customer relationship:

This sub-facet contained eleven elements that are related to the aspects of consumer satisfaction with their relationships with the bank's staff in terms of their abilities, and the facilities they provide to customers. These elements are the following:

1. Bank personnel are concerned about each of their customers.

2. Bank personnel are knowledgeable.

3. Bank managers listen to their customers' needs.

4. There is a good interaction between staff and customers.

5. Bank personnel are friendly.

6. Bank personnel give advice to customers.

7. Bank personnel are willing to explain the services offered.

8. Bank management is always available for consultation.

9. Friendliness of tellers.

10. Complaints are dealt with.

11. Bank personnel make me feel important as a customer.

Sub-facet D2b: Bank advertising:

It is known that advertising plays an important role in influencing consumers' buying behaviour. However, advertising in banking is a very important source of information for consumers about the banks, and the services they provide. It is also one of the most effective tools used by banks to influence consumers, and win them from the market. Therefore, five elements were included in this sub-facet, and they were selected because, they are the only used methods of advertising in Iraq, these are:

1. Bank advertises on T.V.

2. Bank advertises on Radio.

- 3. Bank advertises in Newspapers.
- 4. Bank advertises in Magazines.

5. Personal contact by bank personnel.

Sub-facet D2c: Customer's confidence in bank:

The elements, which were included in this sub-facet, represent some issues that are related to the aspects of customer's confidence in a bank. Five elements were included in this sub-facet, and these are:

1. Bank has a large building.

2. Bank is well-known.

3. Bank has been well-established for many years.

4. Bank has a good reputation.

5. Bank has many branches.

Sub-facet D2d: Bank location variables:

This sub-facet contained five elements, and each of them represented an aspect of convenience of bank location. Thus, the location of the bank, and its branches is an important characteristic, and has a great influence on consumers' selection of a bank. Therefore, five elements, that are related to convenience of bank location were used in this study, and these are:

1. Bank is located near shopping centre.

2. Bank is located near home.

3. Bank is located near place of employment.

4. Bank has many branch locations.

5. Bank is located near public transportations.

Sub-facet D2e: Bank competition variables:

This sub-facet contained five elements, that are related to the aspects of competition between banks. Among a large number of competition variables, the selected five variables were included in this study, because they are the most known competition elements between banks in Iraq. Thus, as mentioned earlier in this investigation (see chapter three), competition between banks in Iraq is very poor. Therefore, only five competition elements were thought to be of value for this investigation in terms of influencing consumers' saving decisions, i.e., to open / or not to open a savings account at a particular bank in Iraq. However, the selected five variables are the following:

1. Bank has full-range of services.

2. Bank provides regular statements of savings balances.

3. Bank has competitive service charges.

4. Bank provides different promotions for savers.

5. Bank pays higher interest on savings.

Sub-facet D2f: bank building design variables:

This sub-facet covers the area of the characteristics of bank building design, in order to uncover its importance to consumers. Four elements were included in this sub-facet. Each one of them was used to find the extent to which consumers' decisions can be influenced by the bank building design and its characteristics, when they select a bank to put their savings in it. However, the four elements, which were included in this sub-facet are the following:

1. Bank has parking facilities for customers.

2. Easy entrance to and exit from bank building.

3. Bank building has modern design.

4. Availability of waiting areas inside the bank's building. Sub-facet D2g: Bank loyalty variables:

This sub-facet consisted of two elements, that are related to customers' loyalty to a bank. each of these elements represents a reason, which may lead a customer to be loyal to a particular bank. However, these two elements are the following:

1. I feel I own part of the bank.

2. I already use other services at the bank.

Facet D3 : Consumers' financial psychographic characteristics:

This facet is related to consumers' personal financial psychographic characteristics. In fact, the decision was made to choose only the most relevant of those characteristics to the subject of consumers' saving behaviour in Iraq. However, this facet
consisted of two sub-facets, and each of them consists of a number of elements, as follows:

Sub-facet D3a: The influence of reference groups:

Consumers' saving behaviour (decisions) can be influenced by their reference groups in terms of the advice, they may receive from them. Therefore, two types of possible reference groups were included in this sub-facet, and these are:

1. Advice from friends.

2. Advice from family members.

Sub-facet D3b: Personal financial psychographic characteristics:

This sub-facet contained five consumers' financial psychographic characteristics. Each one of them was used in order to provide better understanding for respondents financial thinking in terms of the value of money, savings, and credit. However, the five elements, which were included in this sub-facet, are the following:

1. I am a good saver.

Respondents were asked to answer this question, mainly because in Iraq not all savers have savings accounts in banks or other financial institutions. Thus, some people are used to saving their surplus money not at banks, but, at home in a safe place. While, others buy gold in terms of jewels, and other methods of transferring money into valuable things for future use. Therefore, the purpose of this question was to find out if some of the nonsavers in terms of having bank accounts are indeed savers.

2. I like to pay cash for everything I buy.

3. To buy anything, other than a house on credit is unwise.

4. A person can save a lot of money by looking for the lowest possible prices when shopping.

5. It is good to have a savings account because it is useful.

Table 5.1: The Context of Consumers' Saving Behaviour (Decisions) in Iraq (Facets, Sub-facets, and Elements).

No of Facet	Title of Facets, Sub-facets, and Elements				
Facet D1	Consumers' attitudes towards savings accounts:				
Dla	Sub-facet: Benefits of having a savings account:				
	vl Savings help to meet increasing living				
	expenses.				
	v2 Savings help to beat inflation.				
	v3 Savings help to make sure family is				
	protected in case of emergencies.				
	v4 Savings help to build up a retirement fund.				
	v5 Savings help to increase the overall				
4	standard of living.				
	v6 Savings help in buying property for family				
	accommodation.				
	v8 Savings are an investment.				
	v12 Savings help to start a new business.				
	v13 Savings help to cover marriage expenses.				
D1b	Sub-facet: Reasons for having a savings account:				
	v7 I feel may money is safe.				
	vl6 A savings account is a good method to				
	keep surplus of money.				
	v19 Own decision (personally motivated).				
Dlc	Sub-facet: Savings accounts attributes:				
	v33 Convenient hours of operation for savers.				
	v34 Special lines for holders of savings				
	accounts.				
	v35 Accuracy of bank transactions.				
	v36 Confidentiality of bank transactions.				
	v37 Easily of processing bank transactions.				

Table 5.1 (continued)

No of Facet	Title of Facets, Sub-facets, and Elements				
Dld	Sub-facet: Aspects of waiting time:				
	v31 Length of time required to process				
	transactions.				
	v32 Length of waiting time before being served.				
Dle	Sub-facet: Access to savings:				
	v9 I can easily deposit savings at the bank.				
	v10 I can easily withdraw money from the bank.				
Facet D2	Bank attributes:				
D2a	Sub-facet: Aspects of bank-customer relationships:				
	v20 Bank personnel are concerned about each of				
	their customers.				
	v21 Bank personnel are knowledgeable.				
	v22 Bank managers listen to their customers'				
	needs. v23 There is a good interaction between staff and customers. v24 Bank personnel are friendly.				
1					
	v25 Bank personnel give advice to customers.				
	v26 Bank personnel are willing to explain the				
	services offered.				
	v27 Bank management is always available for				
	consultation.				
	v28 Friendliness of tellers.				
	v29 Complaints are dealt with.				
	v30 Bank personnel make me feel important as a				
	customer.				
D2b	Sub-facet: Bank advertising.				
	v52 Bank advertises on T.V.				
	v53 Bank advertises on Radio.				
	v54 Bank advertises in Newspapers.				

Table 5.1 (continued)

No of Facet	Title of Facets, Sub-facets, and Elements			
	v55 Bank advertises in Magazines.			
	v56 Personal contact by bank personnel.			
D2c	Sub-facet: Customer's confidence in bank.			
	v47 Bank has large building.			
	v48 Bank is well-known.			
	v49 Bank has been well-established for many			
	years.			
	v50 Bank has a good reputation.			
	v51 Bank has many branches.			
D2đ	Sub-facet: Bank location variables:			
	v38 Bank is located near shopping centre.			
	v39 Bank is located near home.			
	v40 Bank is located near place of employment.			
	v41 Bank has many branch locations.			
	v43 Bank is located near public transportations			
	(Bus Stops, Taxi Stands).			
D2e	Sub-facet: Bank competition variables:			
	v57 Bank has full-range of services.			
	v58 Bank provides regular statements of savings			
	balances for customers.			
	v59 Bank has competitive service charges			
	(such as interest on loans and other types			
	of charges).			
	v60 Bank provides different promotions for			
	savers (such as free prize draws).			
	v15 Bank pays higher interest on savings.			
D2f	Sub-facet: Bank building design variables:			
	v42 Bank has parking facilities for customers.			
	v44 Easy entrance to and exit from Dank			
	pullulug.			
	v45 Dank Dulluing nas Bouein uesign.			

.

Table 5.1 (continued)

No of Facet	Title of Facets, Sub-facets, and Elements			
	v46 Availability of waiting areas inside the			
	bank's building for customers.			
D2g	Sub-facet: Bank loyalty variables:			
	vll I feel I own part of the bank.			
	vl4 I already use other services at this bank.			
Facet D3	Consumers' financial psychographic characteristics:			
D3a	Sub-facet: The influence of reference groups:			
	v17 Advice from friends.			
	v18 Advice from family members.			
D3b	Sub-facet: Personal financial psychographic			
characteristics:				
	v61 I am a good saver.			
2	v62 I like to pay cash for everything I buy.			
	v63 To buy anything, other than a house on			
	credit is unwise.			
	v64 A person can save a lot of money by looking			
	for the lowest possible prices when shopping.			
	v65 It is good to have a savings account			
	because it is useful.			
Facet E	Research chief objective:			
el	Investigating consumers' saving behaviour			
	(decisions) In Iraq.			

5.5 Conclusions:

In this chapter, a research design for this investigation has been developed and formulated. The faceted approach was adopted as a research design to deal with the research problem of investigating consumers' saving behaviour in Iraq, with special reference to opening savings accounts at commercial banks in Iraq. The survey of the related literature and other informal preliminary interviews with individual consumers in Iraq, were of great help in the process of formulating the facets, sub-facets, and elements, which were mentioned in this chapter.

However, in summary, the faceted approach which has been described in this chapter, concludes that this study is:

A Study of the Cognitive aspects (a1), of Individual Consumers (b1), as perceived by them at present Time (c1), with regard to the aspects of Context of Consumers' Saving Behaviour in Iraq (D) (all the elements), towards the Study Chief Objective of investigating consumers' attitudes towards saving (saving decisions),, with special reference to opening savings accounts at commercial banks in Iraq (e1).

CHAPTER SIX

DATA COLLECTION AND SAMPLING METHODS

6.1 INTRODUCTION.

6.2 DATA MEASUREMENTS.

6.2.1 MEASUREMENTS OF RESPONDENTS ATTITUDES.

.

6.2.2 MEASUREMENTS OF RESPONDENTS

CHARACTERISTICS.

6.3 DATA COLLECTION.

6.3.1 METHODS OF DATA COLLECTION.

6.3.2 DEVELOPMENT OF THE QUESTIONNAIRE.

6.4 SAMPLING.

6.4.1 METHODS OF SAMPLING.

6.4.2 SAMPLE SIZE.

6.5 SUMMARY.

6.1 Introduction:

The main objective of this chapter is to underline the theoretical and practical methods used in the process of data collection from respondents. Thus, a detailed review for the types of data, methods of data collection and sampling methods used is presented in this chapter.

In addition, an introduction to the subject of data measurements is also presented. Such an introduction is presented in order to provide an understanding of the types of data used, and how it is treated in the process of data collection and analysis.

6.2 Data Measurements:

This study, as a marketing research investigation, was oriented towards consumers attitudes and characteristics. It was based mainly on primary data collected from a sample of respondents through questionnaires. Thus, a questionnaire was used in order to explore consumers' attitudes towards (a) the idea of saving; (b) the service of savings accounts provided by commercial banks; (c) some bank attributes may effect consumers attitudes; and (d) to investigate consumers' profile characteristics and its importance in influencing consumers' saving decisions.

Data measurement has been defined by Kerlinger (1973) as the assignment of numerals or numbers to objects or events according to rules. In this definition the word numeral means any symbol that is associated with an object or event, but does not have a quantitative meaning.

Thus, in this study numeric symbols were used to assign the profile characteristics of consumers. For example, in order to measure the variable "sex of respondents" in this study, "O" was

assigned to each male respondent, and "1" was assigned to each female respondent.

Therefore, it can be said that, the main purpose of data measurement is to give numerals or numbers to the variables used in such a way that the results reflect the true situation in the real world.

However, before moving to another section in this chapter it is important to mention the levels of data measurements. Stevens (1946) mentioned that in social science research we recognize four levels of measurements and they are the following:

Nominal measurement:

In this level of measurement, numerals or other symbols are assigned to objects without quantitative meaning. For example, in the case of sex of respondent "O" for male, and "1" for female. Thus, nominal measurement can be used to measure variables such as: sex, race, religion, class standing, and marital status. Ordinal measurement:

Tull and Hawkins (1990) define ordinal measurement as it represents numbers, letters or other symbols used to rank items. These items can be classified not only as to whether they share some characteristics with another item, but, also whether they have more or less of this characteristic than some other objects.

In other words, the ordinal data indicates the relative position of two or more items on some characteristics, but, not the magnitudes of the differences between the items. Ordinal measurements can be used to measure quality rating, ranking preference, classify items, and arrange categories in some meaningful order.

Interval measurements:

Berger and Patchner (1988) indicated that interval measurement has all the properties of both nominal and ordinal measurement: categorisation and rank ordering. Interval measurement has an additional characteristic: numerically equal distance on interval measures indicate equal differences in the property being measured. This characteristic of equal interval allows us to make valid statements about differences between and among categories or values.

However, different types of data can be treated as measured by interval scale used in marketing research such as: level of knowledge about brands, and index numbers. In addition, another type of marketing research data, i.e., attitude measures is treated as interval scaled data. For example, a Likert scale, which requires respondents to state the rate of importance they attached to a statement, or their degree of agreement (satisfaction) or disagreement (dissatisfaction) with a statement by selecting a response from a list.

Ratio measurement:

The ratio scale is a sub-class of the interval scale. In other words, it has all the characteristics of nominal, ordinal, and interval measures: categorisation, rank order, and equal intervals with a meaningful zero base.

There is a number of variables which can be treated as ratio variables such as: age, income, weight, height, the number of years of marriage, and the number of pages in a book.

However in this study two broad types of primary data are collected from respondents: consumers' attitudes and consumers' profile characteristics, i.e., their socio-economic and demographic

characteristics. In the following a description to the data, which were used in this study, and the methods of data measurement, which were used to measure the collected data from respondents.

6.2.1 Measurements of Respondents Attitudes:

It has been concluded in chapter five of this study, that: today most researchers agree that three concepts are related to attitude, these are: (a) Cognition, which refers to a person's knowledge about the object; (b) Feeling, which refers to a person's favourable or unfavourable feelings towards the object; and (c) Conation, which refers to person's intention or actual behaviour towards the object. Therefore, these three related concepts to consumers attitudes were implemented in this study through facets D1, D2, and D3 of this research facet design (see chapter five).

However, regarding the subject of attitude measurements, there are a number of scaling techniques in attitude measurements, which can be used in this regard. Baran (1976) explains that while, there is a large number of scaling techniques, bank marketing researchers tend to use the following most frequently: (a) Ranking; (b) Numerical rating; (c) Verbal rating; (d) Likert scales; and (e) Semantic differential scales. However, among the above mentioned scaling techniques, the decision was made to use the Likert scale in this study, in order to measure the attitudinal data.

The Likert Scales:

In this type of attitude scaling respondent is asked more than one question in order to determine how a respondent feels about something. For example, in this study respondents were asked to indicate the rate of importance they attach to a series of statements related to consumers' saving behaviour.

Thus, in this research and regarding the measurement of respondents attitudes, a 5-point Likert scale was used to measure the attitudinal data gathered from respondents. Respondents were asked to indicate the rate of importance or degree of agreement with a number of statements from "1" very important / or strongly agree to "5" not important at all / or strongly disagree.

In addition, for the purpose of the requirements of data analysis statistical techniques, the 5-point Likert scale was used in measuring the respondents attitudes, because data measured in this scale can be treated as interval data.

Tull and Hawkins (1990) suggested that another common type of marketing research data generally treated as interval scale is attitudes measures. A Likert scale, for example, requires the respondents to state their degree of agreement or disagreement with a statement by selecting a response from a list. Most researchers treat the data from such scales as if they were equal interval in nature, since the results of most standard statistical techniques are not affected by small deviation from the interval requirements.

In addition to the above, the Likert scale was used in this study because: (a) it is easy to construct; and (b) it is easy to understand by respondents. Thus, during the pre-testing of this study's questionnaire, it had been concluded that respondents in Iraq were very satisfied with the 5-point Likert scale, and they mentioned that it is very easily understood. In fact, the obvious reason for such a response towards the Likert scale is because, respondents in Iraq, as in many Third World countries, dislike complicated questions and during the pre-testing of the questionnaire in Iraq the Likert scale was found to be the easiest to understand by respondents. Therefore, they were willing to answer

the questionnaire, because the questions were not complicated in their opinions.

In addition to that, the multiple-choice answer method, i.e., the Likert scale, was used in this study because: (a) it is applicable with most statistical analytical techniques; (b) it is easier to handle; and (c) the choice of respondents helps in reaching respondent's intensity of attitudes regarding the element of universe of content analysed.

However, during the process of data collection, one limitation of using the Likert scale technique was noticed. That was, the questionnaire took a long time to complete by respondents. Generally speaking, taking into account the characteristics of the respondents in Iraq, i.e., generally unfamiliar with mail survey, the Likert scale proved to be successful as a method of attitudinal data measurement in this investigation.

6.2.2 Measurements of Respondents Characteristics:

In this study seven consumers' profile characteristics, i.e., socio-economic and demographic characteristics, were used. These predictors (variables) are the following: Sex of respondent; Age of respondent; Marital status of respondent; Family size group of respondent; Family income group; Household head's educational level; and Household head's occupation.

However, in terms of data measurement and as explained earlier in this chapter, these predictors consisted of nominal and ratio scale variables. Thus, the variables sex of respondent, and marital status of respondent are classified as nominal data. While, the remaining five predictors used in this study are classified as ratio data, these are: age of respondent, family size of respondent,

family income of respondent, educational level (years) of respondent, and occupation of respondent.

The range of questions, i.e., the possible answers, that are related to respondents' socio-economic and demographic characteristics, varied from question to question, and was between 1 and 10.

6.3 Data Collection:

The most important task in conducting a marketing research is the actual gathering of data related to the research problem. However, data can be classified into two main types:

1. Secondary data:

Secondary data is a formal data already available from formal sources such as: previous researches, banks' publications, periodicals which can be collected whenever it is required to provide a better understanding for the research problem.

2. Primary data:

Clowe and Diina (1978) define primary data as information that has to be developed and does not exist in any form, or if it does exist, it is not available for the researcher. This means that such a data has to be created by the researcher from the field of study by using specific methods that best serve his / her study objectives. However, there are three types of primary data: observational, experimental, and survey.

a. Observational data can be gathered by observing objects. For example, a market researcher can stand at a shopping centre and observe the number of people entering a particular super store, and analyse them according to their age groups, sex, and time of entering.

b. Experimental data is another type of observational data, with the difference of observing people and measuring changes in their behaviour. For example, as with the same example above the researcher tries in addition to observe the number of people entering a particular super store, but, registers the number of people who came out of the store with goods, and the number of those without.

c. Survey data is the most useful and complex type of data in marketing research. However, survey data can be obtained only by establishing a direct communication link with respondents.

6.3.1 Methods of data collection:

In bank marketing research primary data collection refers mainly to conduct some type of survey research. In other words, it means asking respondents questions through one or more of primary data collection methods (instruments).

There are mainly three methods of communication with respondents in order to collect the required primary data from them.

The first method of primary data collection is personal interviews. Personal interviews simply conducted by trained interviewers with respondents in a suitable place such as their home or office.

The second method is telephone interviews, such interviews are simply carried out by calling respondents through their personal telephone numbers and asking them the stated questions.

The third method is mail questionnaires. Thus, prepared questionnaires containing the stated questions can be sent to respondents at their fixed addresses. Then, respondents can answer them at their convenience and return them in the provided addressed envelopes.

The mail questionnaire:

In this study a mail survey was conducted in order to collect the required primary data from respondents. However, because of the low standard of the post office services in the country of the field study, i.e., Iraq, all questionnaires were handed over to respondents in their home or office addresses. In addition, assistance was given to respondents in order to explain some formalities and easing some difficulties facing respondents in answering the questionnaires.

The decision was made to use a mail questionnaire, because other methods of data collection, such as: personal and telephone interviews, are less known, impractical, and expensive in terms of time and money in Iraq. For example, using telephone interviews as a method of data collection in Iraq is impractical because: (a) There are no up-to-date telephone directories available, and (b) many people do not have a telephone set at home. Therefore, conducting telephone interviews in a such situation is impossible, and has a great effect on sample representativeness.

Therefore, it was concluded that a simple and easy to understand questionnaire. Which could be handed over to respondents personally and accompanied with personal assistance to ease difficulties that might face respondents, was the only workable method to collect the required primary data from respondents in Iraq.

In addition to the above, the mail questionnaire method of data collection has some advantages and some disadvantages. The main advantages of mail questionnaire method, as described by Clowe and Diina (1978) are the following:

- Wider and more representative distribution of sample is possible.
- 2. Cost per questionnaire is relatively low.
- 3. People are more open to answer questions.
- No interviewer bias is expected, because the answers are in respondents own words.
- 5. Respondent can answer at his / her leisure time and has more time to think about each question.
- 6. Certain segments of population are more easily approachable, such as: those segments of people, who do not have telephone sets.

While, the main disadvantages of the mail questionnaire method are the following:

1. It requires longer time.

- 2. The ability to handle complex questionnaire through mail survey is poor.
- 3. Bias due to non-response is often indeterminate.
- 4. Control over questionnaire may be lost.
- 5. Interpretation of omissions is difficult.
- 6. Only those interested in the subject may reply.
- 7. Questionnaire in most cases must be short.
- 8. Certain segments of population are not approachable, for example, illiterates.

However, in this study an attempt was made to benefit from the advantages of the mail questionnaire method, and overcome its disadvantages. This was done through taking into account respondents characteristics in Iraq and the applicability of each data collection method in Iraq. Providing personal assistance for respondents was of great value in overcoming the disadvantages of

using a mail questionnaire method for the purpose of data collection.

6.3.2 Development of the Questionnaire:

1. The basis of the questionnaire:

It has been mentioned earlier in chapter five of the present study that one of the advantages of using the faceted design, is that the questionnaire is much easier to structure simply by putting each item (or element) of the facets into a question. Therefore, the basis for this study's questionnaire is the facets, sub-facets, and the elements of this research's faceted design as described in figure 5.1 and table 5.1. In other words, the faceted design helped in identifying the questions to be asked and included in the research's questionnaire. In addition, the main resources for formulating the questionnaire in terms of its contents, i.e., questions, are the following:

(a) utilisation of the literature review, i.e., previews studies in bank marketing; (b) observation of the variables that may influence consumers' saving behaviour in Iraq; (c) pre-testing the questionnaire through the pilot interviews carried out with a sample of 50 Iraqi respondents before embarking into the real data collection process; and (d) the mapping sentence of the faceted design of the present study, as shown in figure 5.1.

The first stage in developing the questionnaire is to identify the items (variables), which will be transformed into questions to be included in the questionnaire. In this study and as mentioned above, the use of the faceted design as research design helped in making the designing of the questionnaire easier. Thus, the design of the questionnaire was based on the faceted design of

the universe of content of this research. In other words, each element in each facet was represented in a question.

This research work involved the use of one identical questionnaire to all respondents. The investigation was oriented towards consumers' attitudes about commercial banking services, with particular reference to savings accounts, and consumers' saving behaviour (decisions) in Iraq. In addition, it attempted to underline the relationships between consumers' saving decisions, i.e., hold / not hold a savings account, and consumers' profile characteristics, i.e., socio-economic and demographic characteristics.

Therefore, a questionnaire was used for collecting the required primary data from respondents, in order to investigate consumers' saving behaviour (decisions) in Iraq. In other words, the questionnaire was used to gather data from all groups of respondents. Thus, in the case of investigating consumers' saving decisions, it is natural to find two groups, i.e., "banked with savings accounts" and "non-banked".

Then, respondents were asked to answer a number of (72) questions included in the identical questionnaire. Of which (65) questions were related to consumers' attitudes, motivations, psychographic characteristics. While, the remaining (7) questions were related to consumers' socio-economic, and demographic characteristics.

2. The contents of the questionnaire:

This study's questionnaire was divided into three parts, (see Appendix A), as follows:

Part one:

The first part of the questionnaire contained a number of (60) questions. These questions were designed to investigate how respondents rated the importance of different statements, that are related to savings, savings accounts, bank attributes and facilities. For this part a 5-point Likert scale was used to measure respondents responses, from "1" very important to "5" not important at all.

Part two:

The second part of the questionnaire contained a number of (5) questions, that were related to consumers' financial psychographic characteristics. Thus, respondents were asked to indicate their degree of agreement with statements designed to explore their financial psychographic characteristics. For this part, again, a 5-point Likert scale was used to measure respondents responses, from "1" strongly agree to "5" strongly disagree.

Part three:

The third part of the questionnaire contained a number of (7) questions, that related to consumers' profile characteristics (i.e., socio-economic and demographic characteristics). In this part questions were shaped into categories and the respondents were asked to tick the category they fall in. For this part, the range of questions varied from question to question, and it was between 1 and 10.

As in most survey researches undertaken in developing and less-developed countries, problems in using mail questionnaires were expected. Therefore, in order to overcome the problems and minimise sources of errors in the process and administration of the questionnaires the following procedures were followed:

1. Pre-testing the questionnaire:

The questionnaire was pre-tested, in order to minimise respondents error, with a smaller sample of respondents. This sample was drawn from the same population of this study, i.e., the population of the City of Baghdad in Iraq.

Thus, a sample of 50 respondents was selected by using a probability area sampling method. However, this method of sampling is to be discussed in detail in the forthcoming section, i.e., section 6.4.

The selected sample of 50 respondents was used for the purpose of pilot testing the questionnaire in terms of its design and contents. The difficulties the 50 respondents faced and the comments they gave, helped in improving the questionnaire and restructuring certain questions.

2. Characteristics of the questionnaire:

The questionnaire was kept as simple as possible taking into account the type of respondents likely to answer it. Thus, in Iraq, as in many Third World countries, people are not used to participating in surveys of this type. Therefore, in addition to keeping the questionnaire as simple as possible, simple instructions were included and personal assistance was provided for respondents. 3. Assurance to respondents:

Respondents were given the assurance, that any information given would be treated as confidential and analysed in aggregate. Also, respondents were asked not to give any names or addresses.

Justification for using the questionnaire with the group of "non-banked":

The present study is an investigation of consumers' saving behaviour carried out through empirically investigating consumers' attitudes towards the service of savings accounts provided by commercial banks in Iraq.

Consumers can be divided into two main groups, users and nonusers of the service at present time, i.e., the moment in which the questionnaire is distributed to respondents for the purpose of primary data collection. Therefore, the decision was made to send this study's questionnaire to two groups. The first group represents holders of savings accounts with commercial banks in Iraq, i.e., "banked with savings accounts". The second group represents nonholders of savings accounts, i.e., "non-banked".

It was expected that respondents who are banked with savings accounts will find it much easier to respond intelligently to the questions asked mainly because they are actual users of the service. However, the problem was to secure similar responses from respondents who are non-banked. Achieving such an aim is not an easy job. In fact, during the process of data collection for the present research the most difficult part was collecting the data from the group of "non-banked". Thus, non-banked respondents, generally speaking, were found to be unfamiliar with the service and some of the questions they were asked. Therefore, a personal contact by the researcher with each non-banked respondent was required in order to explain to respondents the questions which they found difficult to understand. In some cases, samples of banking forms such as bank statements were shown to respondents to make it easier for them to understand.

6.4 Sampling:

Churchill (1987) defines sampling as the selection of a subset of elements from a larger group of objects. These elements (objects) are to be drawn from the sample frame as a portion of the population, in order to collect the required data.

The term "study population" or "sample frame" has been defined by Kinnear and Taylor (1987) as the aggregate of elements from which the sample is drawn. Thus, in this study there were two sample frames. The first sample frame was all the elements, who have savings accounts for the first sub-sample of "banked with savings accounts". The second sample frame was all the elements without savings accounts for the second sub-sample of "non-banked".

However, in this study the total population represented all consumers who are "banked with savings accounts" and "non-banked" within the field of study, i.e., all the population of the City of Baghdad in Iraq, consumers who are "banked with savings accounts" and "non-banked".

6.4.1 Methods of Sampling:

Clowe and Diina (1978) explain that perhaps the biggest problem in bank marketing research for both researcher and research credibility for bank management is the area of sampling. Therefore, a number of sampling methods are used in marketing research.

However, there are two main sampling methods which can be used in selecting a representative sample from a particular population, and they are as follows:

1. Random or Probability Samples:

Random or probability sampling refers to the idea of giving each element (respondent) of the population an equal chance to be

selected in the sample. In other words, each element has a known, non-zero, opportunity of being included in the sample.

There are a number of random or probability sampling methods such as: simple random sampling, cluster sampling, systematic sampling, area sampling, and stratified sampling.

2. Non-Probability Sampling:

Kinnear and Taylor (1987) define non-probability sampling as the selection of a population element to be part of the sample is based in some part on the judgment of the researcher or field interviewer. In other words, chance selection procedures are not used as it is used in probability sampling. However, there are three main methods of non-probability sampling, and these are: convenience sampling, judgement sampling, and quota sampling.

However, in this study a combination of two probability sampling methods was used in order to derive the required number of respondents from the study population, these are:

1. Area probability sampling:

Area sampling is one of the probability sampling. This method of sampling is useful, especially, when the aim is to select respondents from population living in a particular area or areas. However, in this investigation, this method of sampling was used because of two main reasons:

The first reason for using area sampling was that the research was undertaken in Iraq. In Iraq, as is the case in many developing countries, current and accurate lists of population elements are rarely available. Telephone or other directories of all those living in a city at a particular moment simply do not exist for many cities in Iraq, and when they do exist, they are obsolete when published.

While, the second reason was the desire to guarantee a personal contact with respondents for several reasons:

(1) A high expectation of delay was expected, and even loss of questionnaires, if the questionnaires were sent by mail, because of ineffective post office services in Iraq;

(2) Receiving a letter from some one you do not know is unusual and sometimes unacceptable in Iraq. In fact, approaching people at their home / work address to make personal contact with them is more acceptable and usual than to send the questionnaire by mail;

(3) To illustrate some questions, if required to respondents; and

(4) To help respondents during the process of answering the questionnaire, especially as many have rarely dealt with such an experience.

However, the extent (area) of this study was the City of Baghdad. Baghdad was chosen because: (a) it is the capital of Iraq; (b) it represents a highly competitive banking environment, where consumers have a wide range of banks, bank branches, and other financial institutions to choose from; (c) the population of the capital, as in most countries, are more likely to be users of banking services than the population of other cities and towns, especially in Iraq; and (d) given the difficulties, that are explained earlier in reaching respondents, choosing the capital was the only way to complete the job, within the limited time, costs, and efforts available.

This research study involved an investigation of consumers' attitudes towards commercial banking services in Iraq, with particular reference to savings accounts. Therefore, the

questionnaires, which were designed to collect the primary data, were distributed to two groups of respondents "banked with savings accounts" and "non-banked". The distribution of the questionnaires was carried out by the following procedures:

Firstly, the city of Baghdad was divided into (20) main districts (areas). A number was given for every district, from 1 to 20. Then, the decision was made to take from each main district a number of (40) respondents, (20) of them should be "banked with savings accounts", and the others should be (20) "non-banked".

Secondly, for the purpose of selecting the (20) "non-banked" from each district, the following procedures of area sampling were used (see figure 6.1):

A listing was made for the geographic areas within the sample location, i.e., District in this case, which was divided then into four geographic areas (north, south, east, and west) .

Each geographic area contains a number of chunks, a probability sample of chunks was then selected from each geographic area (one chunk only).

Each chunk contains a number of housing units (dwellings), therefore, a listing was made for the housing units in the selected chunk. Then, a probability sample of housing units from each chunk was selected (one housing unit only).

The selected housing units were divided into smaller units called segments. Each segment contains a number of houses, then a probability sample of segments was selected (one segment only).

From each segment which contains a number of houses, a probability sample of houses was then selected (five houses from each segment).

This means a number of (20) "non-banked" respondents' houses were selected from each district. In other words, a total of 400 "non-banked" respondents' houses were selected from 20 main districts in the city of Baghdad.

But, the total sample required was 800 respondents comprising 400 "non-banked" and 400 "banked with savings accounts" and as explained above the respondents of "non-banked" were secured through adopting the above mentioned sampling procedure.

Therefore, what remains is to explain how the 400 respondents who are "banked with savings accounts" were selected and secured. Thus, in the following a description for the sampling procedures, which were used to secure the remaining 400 "banked with savings accounts" respondents.

2. Systematic probability sampling:

In this method of probability sampling a selection frame is to be defined, such as a list of savers, i.e., "banked with savings accounts" at a bank. Then, every Kth element in the frame is to be selected, after a random start somewhere within the first K elements.

However, this method of probability sampling was used in this investigation in order to select the 400 "banked with savings accounts", through adopting the following sampling procedures:

In each, of the previously mentioned selected 20 main district in the city of Baghdad, there is a branch of a commercial bank. In fact, this is the actual situation in Baghdad, where the banks policy is to open branches in main districts, and even the branches are named after the name of the districts.

Therefore, for the purpose of selecting the 20 "banked with savings accounts" from each district, access was given, by the

branches of the banks, to the records of savers, i.e., lists of savers' names and addresses only, within each district (see figure 6.1).



Figure 6.1: The sampling procedures used in this investigation.

By utilising a probability systematic sampling method a number of 20 "banked with savings accounts" from each branch, i.e., from each district, was secured. In other words, a total of 400 "banked with savings accounts" were selected from 20 bank branches located in 20 districts in the city of Baghdad. Then, each selected saver was contacted at his / her address to complete the required questionnaire.

In fact, the decision of choosing respondents who are "banked with savings accounts" and "non-banked" from the same districts was made in order to increase sample representativeness. In other words, the assumption was that, if the two typologies were selected from the same districts, the chance of the sample being more representative will increase.

As it is known, people living in the same districts are likely to share common social, economic, and demographic characteristics. In fact, this is the real situation in Baghdad, where there are some districts characterised as the districts of the rich, while, other characterised as the districts of the poor. Therefore, dividing Baghdad into 20 district was aimed to include all types of districts, i.e., in terms of upper, middle, and poor districts.

Finally, through adopting the above mentioned two methods of probability sampling, i.e., area and systematic sampling, a total number of a sample of 800 respondents was secured (see figure 6.1). In other words, a total number of 800 questionnaires were distributed to respondents. Table (6.1) shows the number of questionnaires distributed and the rate of response.

Table 6.1: Rate of Response.

Description "I	"banked with	
Sav	ings accounts"	
Number of questionnaires distributed.	400	400
Number of questionnaires returned.	321	379
	(80%)	(95%)
Usable questionnaires (duly completed)	. 265	356
Non-usable questionnaires	56	23
(not duly completed)		

Table 6.1 shows a high percentage of response rate for both groups. This was due to personal communications with respondents, and the high level of co-operation offered by them.

According to Churchill (1987) probability samples are distinguished by the fact that each population element has a known, non-zero chance of being included in the sample. With nonprobability samples, on the other hand, there is no way of ensuring that the sample is representative of the population.

Based on that, the first decision was made to use probability samples in this investigation. While, the second decision was made to select the most suitable methods among the alternative methods of probability sampling in this investigation. These two decisions guided the research sampling method and helped in establishing a combined method of area and systematic probability sampling which was used in this investigation, as explained earlier in this chapter.

However, the method of sampling used in the present investigation are subject to the following limitations: (a) as is known, probability sampling methods are more difficult to use than non-probability sampling methods; (b) Combining two methods of probability sampling, i.e., area and systematic, was a complicated, time consuming and costly task to achieve the required sample size. By reviewing and testing the alternatives during the pilot study it was concluded that the selected methods were the most suitable to serve the purpose of selecting a representative sample drawn from the population in Iraq for the reasons mentioned earlier in this chapter.

6.4.2 The Size of the Sample:

As showed in table 6.1, a sample of (800) respondents was used for the purpose of collecting the primary data required from respondents. In deciding the size of the sample, two important points can be mentioned.

Firstly, Peterson (1982) referred to some rules of thumb when determining the size of the sample, like a minimum sample of (400) when estimating a single population parameter in order to get research findings, which are generally reliable. According to him, larger samples than (400) typically are not justified considering the incremental costs involved. Also, he mentioned a second rule of thumb, which is that there should be a minimum of four study participants (respondents) per questionnaire item, which generally results in reliable results from a decision-making prospective.

For instance, in this study the questionnaire contained (72) items (variables) (see Appendix A). Therefore, the minimum sample size of this study should be not less than N = 288, thus:

N = No. of questionnaire items x 4

 $N = 72 \times 4$

N = 288

While, a total sample of 800 respondents was used in this investigation of which, 400 "banked with savings accounts" and 400 "non-banked".

In addition to this, Kinnear and Taylor (1987) explain that the choice of a sample size depends on factors such as: statistical precision requirements, study objectives, time available, cost, and data analysis plan.

However, in this study two additional factors affected the decision of choosing a sample size. Firstly, the study population was all the residents of a big city (about four million people). Secondly, there were no accurate data, statistics, or directories about the residents, especially for the sub-population of "nonbanked".

In fact, the decision to choose a sample size of (800) respondents [400 "banked with savings accounts" and 400 "nonbanked"] was made for the following reasons: (a) increasing the chance of sample representativeness of the population to gain more reliable results; (b) allowing for the rate of non-responses and not duly completed questionnaires which might occur; and (c) as mentioned earlier this study was about two big sub-populations "banked with savings accounts" and "non-banked" drawn from one main population (residents of the City of Baghdad).

6.5 Summary:

This chapter was devoted to explain the types of data, methods of data collection and the sampling procedures, which were employed in this investigation.

Data measurements consist of four types of measurement: Nominal, Ordinal, Interval, and Ratio. Each type of these measurements required special treatment in terms of converting them into numerical symbols, in order to facilitate using them in data collection and data analysis.

A questionnaire was designed to collect the required primary data from respondents. The questionnaire design of this study was based on the research faceted design.

The questionnaire contained (72) items (questions), that are related to the aspects of consumers' saving behaviour, attitudes towards some bank attributes, and consumers' socio-economic and demographic characteristics.

A total of (800) questionnaires were distributed to respondents. A combination of probability systematic and area sampling was used in order to select the sample of respondents.

The response rate was very encouraging, thus, the response rate was (80%) in the case of respondents who are "banked with savings accounts", while, it was (95%) in the case of "non-banked".

CHAPTER SEVEN

METHODS OF DATA ANALYSIS

.

7.1 INTRODUCTION.

7.2 FACTOR ANALYSIS.

7.3 TWO-GROUP DISCRIMINANT ANALYSIS.

7.4 MULTIPLE CLASSIFICATION ANALYSIS.

7.5 THE KOLMOGOROV-SMIRNOV GOODNESS-OF-FIT TEST.

7.6 THE F-TEST.

7.7 THE T-TEST.

7.8 SUMMARY.

7.1 Introduction:

This chapter presents the statistical methodology of analysis which was used in this study to analyse the research primary data. The subject of selecting the most appropriate methods of data analysis is a complicated one and depends on many considerations such as:

1. The number of variables to be analysed at a time. Thus, (a) if the number of variables to be analysed at a time is one, then, the appropriate methods are the univariate methods of data analysis; (b) if the number of variables to be analysed at a time is two, then, the appropriate methods are the bivariate methods of data analysis; and (c) if the number of variables to be analysed at a time is three or more, then, the multivariate methods of analysis are the appropriate.

2. The level of measurement of data. Thus, each type of data can be analysed by using the appropriate method of data analysis. In other words, qualitative (non-metric) data, i.e., nominal and ordinal, and quantitative (metric) data, i.e., interval and ratio, each of them can be analysed by using a set of statistical methods of data analysis.

3. The interrelationship between variables, i.e., interdependence versus dependence variables. Thus, if the data is a multivariate in type, i.e., more than two variables, then, the interrelationship among all variables taken together is the prime consideration in selecting the appropriate method of data analysis. For example, in the case of one or more variables are dependent on a number of independent variables the dependence multivariate methods are the appropriate. While, in the case of no variable or variables

are predicted by others the interdependence multivariate methods of analysis are the appropriate.



Figure 7.1: Methods of data analysis. Sources: Sheth (1971), Meidan (1975, B), and Kinear and Taylor (1987).

Figure 7.1 explains the alternative major methods of data analysis which can be used for the purpose of data Analysis in marketing research.

The selection of the methods of data analysis used in this study was based on the assumptions mentioned above and explained in figure 7.1. Thus, because the data of this investigation is
multivariate, qualitative (non-metric) of ordinal and nominal type and the variables are of dependence and interdependence nature only the multivariate methods are appropriate for the purpose of data analysis. Therefore, three major statistical multivariate methods were used for the purpose of data analysis in this investigation, these are: Factor Analysis, Discriminant Analysis and Multiple Classification Analysis. In addition, a number of statistical tests was used to test the relevant research hypotheses.

Each method of data analysis is discussed below through the following formula: (a) What is the method? (b) What are the reasons or purposes for selecting the statistical method? and (c) What are the alternative statistical methods to each method used?

7.2 Factor Analysis:

Massy (1971) defined Factor Analysis (FA) as basically a method for reducing a set of data into a more compact form. The user of FA focuses on the set of variables for which information has been collected and poses the question: Can the information be summarised in a smaller number of factors?

The answer to this question was provided by practical applications of FA in marketing research which proved that FA can be used to accomplish the aim of reducing data with a minimal loss of information.

Morrison (1976) suggested that technically Factor Analysis is a set of methods in which the observable or manifest responses of individuals on a set of variables are represented as functions of a small number of latent variables called "factors".

Green and Tull (1978) mentioned that rather than trusting strictly to intuition, Factor Analysis systematically explores which

variables exhibit high intraset correlations and low interset correlations, how many such sets there are (each set defining a dimension), and whether the dimensions can be considered as uncorrelated themselves.

Thus, the purpose of FA is to define "factors" or "dimensions" which can describe the data. For example, one of the objective of this study is defining a number of factors or dimensions which can describe the most important variables related to savers, i.e., banked with savings accounts, attitudes towards savings and having savings accounts at commercial banks. In other words, FA can help in this study by reducing the data from a number of 60 variables into a more compact form and provide better identification in terms of dimensions or factors with a minimal loss of information.

Figure 7.1 shows that Factor Analysis is one of the multivariate methods of data analysis which can be used when the variables are interdependent in nature. Therefore, in this study Factor Analysis was used in order to analyse the attitudinal data related to savers, i.e., banked with savings accounts, attitudes towards savings in general and having savings accounts at commercial banks in particular.

The main purpose for using Factor Analysis in this investigation is related to achieving the research objective of determining a minimal number of saving behaviour factors to replace the large number of independent variables (60 variables) with a minimal loss of information. In other words, FA was used to achieve the objective of determining those attitudinal <u>FACTORS</u> which are perceived by <u>SAVERS</u>, i.e., <u>BANKED WITH SAVINGS ACCOUNTS</u>, as of significant importance in influencing their decisions to open

savings accounts at commercial banks in Iraq. Table 7.1 provides further justifications for using FA in this investigation.

Research Objective	Justifications
To determine those universal	Factor Analysis is a multivariate
attitudinal factors which are	technique can be used in order to
perceived by the Iraqi savers,	reduce the data into a more
as important in influencing	compact form and provide better
their decisions to open	identification for consumer saving
savings accounts at the	behaviour variables. Also, Factor
commercial banks.	analysis allows to examine the
	strength of the overall association
	among variables in terms of a
	smaller set of linear composites of
	the original variables. Therefore,
	FA was used in this investigation.

Table 7.1 Justifications for using Factor Analysis.

However, as it has been explained in figure 7.1, there are other alternative multivariate methods of data analysis which can be used instead of Factor Analysis in marketing research, such as: Cluster Analysis, Multidimensional Scaling Analysis; and Latent Structure Analysis.

In fact, the selection of FA to be used in this investigation among other alternatives was due to the reasons mentioned in table 7.1. In other words, the stated objective for using FA in this study was to determine attitudinal factors, i.e., the search for common factors to replace the attitudinal variables. Therefore, FA was found to be the most appropriate statistical technique which will do just that, i.e., replace the large number of variables into a minimal number of common factors with a minimal lose of information.

Also, Factor Analysis was chosen among other alternative techniques because as it has been described by Sheth (1971) as a data-reduction method which summarises the commonality of all the manifest variables into a few factors.

However, the decision of a particular statistical method of analysis among other alternative methods depends entirely on a number of criteria, such as:

1. The research objective, in other words, the purpose for using the statistical method of data analysis.

2. The advantages and disadvantages of each statistical method of data analysis and how it may affect the research findings.

3. The type of data which is to be analysed by using a particular statistical method of data analysis.

4. The availability of computer programmes to carry-on the analysis with a particular method of analysis.

5. If all the above criteria are considered and the market researcher concluded that more than one method can be used, then he / she have to make his / her decision and select the most appropriate statistical method of analysis to his / her research.

Therefore, the decision to select FA in this investigation among other previously mentioned alternatives was based on the criteria and assumptions mentioned above.

Finally, the statistical results of this investigation's FA are presented in chapter eight of this investigation.

7.3 Two-Group Discriminant Analysis (TGDA):

Discriminant Analysis (DA) has been defined by Klecka (1980) as a statistical technique which allows the researcher to study the differences between two or more groups of objects with respect to several variables simultaneously.

Sheth (1970) mentioned that DA is useful in situation where a total sample is divided into known groups based on some classificatory variable, and the researcher is interested in understanding group differences or in predicting correct belonging to a group of a new sample based on the information on a set of predictor variables.

However, as shown in figure 7.1, DA is one of the multivariate methods of data analysis which can be used when the classificatory variable is dependent in nature, and the predictor independent variables are qualitative (non-metric).

DA is divided into two types: Multiple Discriminant Analysis (MDA) which can be used when the study has a single dependent variable of three or more categories; and Two-Group Discriminant Analysis (TGDA) which can be used when the study has a single dependent variable of two categories, i.e., dichotomous. Therefore, the interest in this investigation was in Two-Group Discriminant Analysis because the dependent variable for the DA carried out in this investigation was dichotomous, i.e., two groups only: consumers who are "banked with savings accounts" and those "non-banked".

The selection of DA among other multivariate dependent methods, as shown in figure 7.1, to be used in this investigation

was based on the previously mentioned criteria. In addition, Frank and Massy (1963) mentioned that Regression Analysis takes a continuous dependent variable, such as durable good expenditures, and estimates the degree to which each of a set of independent variables like income and age are related to it. In contrast, Two-Way Discriminant Analysis is a tool for estimating the relationship between a set of independent variables and a dichotomous attribute such as whether families are Folger's (Coffee) loyal or non-Folger's customers.

In other words, Two-Group Discriminant Analysis classifies respondents into two exclusive groups and Multiple Discriminant Analysis classifies respondents into three or more exclusive groups and that depends on the number of categories in the dependent variable used in the analysis. While, Regression Analysis, for example, relates a continuous dependent variable to a set of independent variables.

Grimm (1974) mentioned that DA classifies individuals or objects into one of two or more exclusive groups, in contrast to Regression Analysis and Multiple Classification Analysis (MCA) that position individuals or objects on a continuum. The dependent variable is categorical in DA. While, it is continuous for Regression Analysis and MCA.

Therefore, TGDA was used in the present investigation to analyse the relationship between the 65 independent attitudinal variables and a dichotomous dependent variable (attribute). Also, it was used in order to classify respondents into two mutually exclusive and exhaustive groups of respondents. Thus, TGDA was used in the present investigation in order to determine the relationship between the 65 independent attitudinal variables and the dependent

dichotomous variable consumers' saving decisions, i.e., consumers who are banked with savings accounts vs. non-banked. Table 7.2 provides the justifications for using TGDA in the present investigation.

Research Objective	Justifications
To establish the extent to	Two-Group Discriminant Analysis is
which differences exist	one of the dependent methods of
between banked and non-banked	multivariate analysis, which can be
consumers in terms of their	used when the study has one single
attitudes towards the service	dependent variable which is
of savings accounts provided	dichotomous. The primary objective
by commercial banks in Iraq.	of DA is to correctly classify
	entities into mutually exclusive
	and exhaustive classes or groups.
	Therefore, DA was used in this
	investigation.

Table 7.2 Justifications for using TGDA.

Finally, the statistical results of this investigation's TGDA are presented in chapter nine of this study.

7.4 Multiple Classification Analysis (MCA):

The Multiple Classification Analysis (MCA) is a multivariate method of data analysis, which can be used when the study has a single dependent variable. The MCA is a suitable statistical method to examine the relationship between a dependent variable and a set of independent predictors (variables). For example, the MCA can be used to examine the relationship between consumers' saving decisions (hold / not hold a savings account) and a number of predictors such as: sex, marital status, age, family size, income, qualification and occupation.

Anderson and Nevin (1975) indicated that the MCA program is a convenient variant of dummy variable regression analysis, which, like all dummy variable programs, does not require interval scaled independent variables.

Andrews, Morgan and Sonquist (1973) mentioned that the major statistics which can be provided by MCA program are the following: (a) the Eta statistic; (b) the Beta statistic; (c) the multiple coefficient of determination (R); and (d) the F-ratio. However, each of these statistics and its advantages are described in more detail in chapter ten of this investigation together with the statistical results of MCA.

However, the MCA was selected to be used in the present investigation among other alternative techniques shown in figure 7.1 because it was found to be the most appropriate technique for the purpose of analysing the categorical data of this research.

Thus, the main objective for using the MCA in the present investigation was to examine the relationship between the dependent variable (consumers' saving decisions to hold / not hold a savings account at commercial banks) and seven independent variables (predictors) taken together or separately. In other words, the main objective was to examine the relationship between consumers' saving decisions and seven of their socio-economic and demographic characteristics, such as: sex, marital status, age, family size, income, educational level and occupation. Table 7.3 explains the justifications for using MCA in the present investigation.

Table 7.3: Justifications for using MCA.

Research Objective	Justifications
To examine the relationship	The MCA is a multivariate method of
between consumers' saving	data analysis which can be used
decisions (hold / not hold	when the purpose of the study is to
savings accounts) and seven	examine the relationship between
of their socio-economic and	several predictors (taken together
demographic characteristics:	and separately) and a single
1. Sex; 2. Marital status;	dependent variable. Therefore, the
3. Age; 4. Family size;	MCA was used in the present study.
5. Income; 6. Qualification;	
and 7. Occupation.	

The statistical results of this investigation's MCA are presented in chapter ten of this investigation.

7.5: The Kolmogorov-Smirnov Goodness-of-Fit Test:

Siegel (1956) mentioned that the Kolmogorov-Smirnov onesample test is a test of goodness of fit. That is, it is concerned with the degree of agreement between the distribution of a set of sample values (observed scores) and some specified theoretical distribution. It determines whether the scores in the sample can be reasonably be thought to have come from a population having the theoretical distribution. Also, the same author said that the Kolmogorov-Smirnov is the strongest statistical test among all the one sample non-parametric tests.

However, in the present study the Kolmogorov-Smirnov test was selected among other alternative non-parametric test because it is the most appropriate statistical test to test a research hypothesis

when the study has one sample and the level of measurement of data (variables) is in an ordinal scale. In other words, the Kolmogorov-Smirnov test was used in this study to test the research hypothesis of chapter eight of this investigation where one sample (a sample of 265 "banked with savings accounts") only used and the data (variables) was of an ordinal scale. The statistical results of this investigation's Kolmogorov-Smirnov test are presented in table 8.14 of chapter eight.

7.6 The F-Test:

The F-test is a parametric statistical test. This test was used in the present investigation to test the research hypotheses which are related to the use of Multiple Classification Analysis (MCA) of this investigation. In other words, the basic statistics for calculating the F-test are provided by the MCA programme which are used as input for the purpose of testing the related research hypotheses as shown in chapter ten.

In fact, the intentions from using the F-test were to determine: (a) whether all the independent variables taken together explain the variation in the dependent variable; and (b) whether each independent variable taken separately explains the variation in the dependent variable.

Therefore, two types of F-tests were carried-out in this investigation. The first test, i.e., whether all predictors taken together explain the variation in the dependent variable, was carried out by applying the following equation:

Where:

E = Explained sum of squares.

C = Total number of categories in all variables.

P = Number of predictors (variables).

Z = Residual sum of squares.

N = Number of cases (respondents).

The second test, i.e., whether each predictor taken separately explains the variation in the dependent variable, was carried out by using the following equation:

$$F = \frac{U / (C - 1)}{i i}$$

$$F = \frac{U / (C - 1)}{(T - U) / (N - C)}$$

$$i i$$

Where:

i

i = The predictor (variable).

U = Sum of squares based on unadjusted deviation from the i grand mean of the MCA output for the predictor.

C = Total number of categories in the predictor.

T = Total sum of squares.

N = Number of cases (respondents).

However, the statistical results of using the F-test for the purpose of testing the relevant hypotheses to the MCA programme are presented in chapter ten of this investigation.

7.7 The T-Test:

The t-test is one of the parametric tests. It is useful to examine (test) the differences between two groups. It is known that the parametric tests, such as the t-test, are more powerful than the non-parametric tests. In the present investigation the t-test was used to examine the differences between Iraqi consumers of banking services if divided according to their saving habits, age, income, education, and family size. The SPSS-X computer programme provides the results of t-tests, which are discussed in section 9.2.2 of chapter nine of this study.

7.8 Summary:

This chapter was devoted to providing insights into the statistical methods of data analysis which were used in this investigation.

The discussion for each method of analysis was based on its definition, use, alternatives, applications and the reasons for selecting it to be used in the present investigation among other alternative methods of data analysis.

Therefore, the relevant theoretical aspects about Factor Analysis (FA), Two-Group Discriminant Analysis (TGDA) and Multiple Classification Analysis (MCA) were presented in this chapter. In addition to that, the three statistical tests which were used for the purpose of testing this investigation's hypotheses are also discussed in the same manner.

CHAPTER EIGHT

FACTOR ANALYSIS FINDINGS

8.1 INTRODUCTION.

8.2 INTERPRETATION OF FACTOR ANALYSIS FINDINGS.

8.3 TESTING RESEARCH HYPOTHESIS.

8.4 CONCLUSIONS.

8.1 Introduction:

Factor analysis was used in this study to determine a number of saving behaviour factors, to replace the larger number of independent variables.

Frank and Green (1967) suggested that factor analysis can be used as a technique to analyse the inter-correlations within a set of variables. Its primary purpose is to find a way of summarising the information contained in a number of original variables with a minimum loss of information.

However, factor analysis was used in this investigation to summarise the attitudinal data gathered from "respondents who are banked with savings accounts" only. The reason for doing this was based on achieving the first objective of this investigation. That is, determining those attitudinal factors which are perceived by "respondents who are banked with savings accounts", as nf significant importance in influencing their decision to open a savings account at a commercial bank in Iraq. In other words, the first stage of this study's plan, as shown in figure 1.1 of chapter one, is to determine actual savers attitudes towards the service of savings accounts provided by commercial banks in Iraq. The responses of 265 out of a sample of 400 "banked with savings accounts" are to be analysed in this chapter. This analysis was considered as a first step. It was assumed that knowing the attitudes of actual savers who are "banked with savings accounts" is very important as an advance step, before going into analysing the differences between consumers attitudes according to their saving decisions, saving habits, and socio-economic and demographic characteristics. This will be explained in chapters nine and ten of this investigation.

Respondents who are "banked with savings accounts", were asked to answer 60 different attitudinal questions, which were related to their decision to save by opening a savings account at a commercial bank in Iraq. Their responses were analysed by using FA. 8.2 Interpretation of Factor Analysis Findings:

A Factor Analysis using Principle Components Analysis (PCA) computer programme, with Varimax Rotation, was used to analyse the attitudinal data gathered from the sample of "banked with savings accounts" (265 respondents). The analysis was based, first of all, on the following assumptions:

1. The correlation matrix between the attitudinal variables was used as input for the analysis. Norusis (1985) indicated that "since one of the goals of Factor Analysis is to obtain "factors" that help to explain these correlations, the variables must be related to each other for the factor model to be appropriate". The higher the correlation between variables, the more likely they share common factors. Therefore, two indicators are adopted in this investigation to make sure that the correlation matrix can be used in the factor model of this study.

The first indicator, as had been described by Norusis (1985), is called Bartlett's Test of Sphericity (based on a chi-square transformation of the determinant of the correlation matrix). That is, if the value of the test statistics for sphericity is large and the associated significant level is small, it is unlikely that the population correlation matrix can not be used in factor model.

The result of the Bartlett's Test of Sphericity for this study indicated that the value of the test statistics for sphericity was large 8663.17, and the associated significant level was very

small .000. So, the correlation matrix between the variables can be used in the factor model of this investigation.

The second indicator is called Kaiser-Meyer-Olkin (KMO) measure of sampling adequacy. This measure is used as an index for comparing the magnitudes of the observed correlation coefficients to the magnitudes of the partial coefficients. High values for the KMO measure indicate that a Factor Analysis of the variables may be a good idea. Kaiser (1974) characterizes measures in the .90's as marvellous, in the .80's as meritorious, in the .70's as middling, in the .60's as mediocre, in the .50's as miserable, and below .50 as unacceptable. The value of the overall KMO statistic for this investigation was calculated to be equal to .85. This means, that it carry-on with Factor was comfortable to Analysis in this investigation, because the value of the KMO measure was very high, i.e., can be classified as meritorious.

2. Regarding determining the number of factors to be extracted by factor analysis, Tull and Hawkins (1990) suggested that un-aided judgement sometimes is used. This generally involves the researcher stopping at the first factor that is difficult to interpret. The most common approach is to use a rule of thumb referred to as the eigenvalue - greater - than - one rule. Therefore, for the purpose of determining the number of factors to be extracted by Factor Analysis, only factors that accounted for variance greater than 1.0, i.e., the eigenvalue is greater than 1.0, was included in the final statistics of the factor analysis of this study.

3. In respect to the concept of factor loading, different scholars referred to different opinions in defining the significant level of factor loading to be adopted by analysts. Fitts (1975) mentioned that factors are interpreted in terms of the variables which load

significantly on that factors. Wells and Sheth (1971) ignored the issue of whether a factor loading is significant and referred to the concept of "high" and "low" loading only. While Mukherjee (1965) referred to the concept of choosing an arbitrary cut-off point such as .50. Harman (1967) presented a table of standard errors of factor coefficients which can be used to determine whether a factor loading is statistically significant at any predetermined level of significance.

Based on the above, factor loading of .45 was used as cutoff value for factor loading in this investigation. In other words, only variables with factor loading of .45 and over with at least one of the factors extracted by Factor Analysis were accepted, and all loading below .45 were eliminated. However, it should be explained that the decision of choosing .45 as cut-off value for factor loading, was made in order to minimise the loss of information. Thus, if a higher cut-off value than .45 was selected, it is possible that some important information would have been lost. 4. Tull and Hawkins (1990) indicated that in factor analysis another rule of thumb requires that enough factors be retained to explain a "satisfactory" percentage of the total variance. Unfortunately, "satisfactory" is difficult to specify. Therefore. in this investigation and in order to reach a satisfactory percentage of the total variance, thirteen factors were retained. They were also retained because their eigenvalues were greater than 1.0 (see item 2

above).

Based on the above assumptions, and in order to provide a multivariate perspective on the 60 variables related to respondents who are "banked with savings accounts" attitudes toward commercial banking services, with special reference to savings accounts, and

other bank attributes, a Principle Components Factor Analysis with Varimax Rotation was carried out. A thirteen factor solution was obtained, which accounted for 64.5 per cent of the explained variance, and included 57 attitudinal variables, i.e., with a minimum loss of information (57 out of 60 variables were retained). In other words, a number of thirteen factors together explained 64.5 per cent of the variance in the original consumers rating of the importance of some of 60 attitudinal variables.

The decision to include a variable in a factor was based on factor loading greater than or equal to .45. Therefore, a number of three variables were excluded from the rotated factor matrix, because their factor loading with any of the 13 factors extracted were less than .45 (see Appendix C).

The results of Factor Analysis represented in the thirteen factors extracted, with the variables and their factor loading, and the variance explained by each factor, are shown in Tables 8.1 to 8.13 below:

Factor One: Customers' satisfaction:

As shown in table 8.1, this factor accounts for 20.2 per cent of the explained variance. It includes twelve variables, that are closely tied to customers' satisfaction in their relationship with the bank in terms of certain personal aspects, other facilities provided by the bank staff to customers and how respondents who are "banked with savings accounts" evaluate their importance in influencing their decision to open savings accounts at a particular bank. This factor indicates that aspects of customers' satisfaction are very important in influencing consumers saving decisions. Therefore, banks should attempt to improve the channels of bank-

customer relationship in order to keep existing customers and gain more customers. This factor indicates that Iraqi customers are very concerned about their relationship with the bank, in particular how bank management and personnel treat them. In their study Arora, Cavusgil and Nevin (1985) concluded a similar factor and they described it as "representing the dependability of the institution". Thus, customers feeling secure and satisfied when they deal with a bank, reflects a positive and reliable bank-customer relationship. Also, Yavas (1988) suggested that bank's standard of customer care is the prime consideration for the Saudi bank customers when they choose a bank.

Loading	Variable Code	Variable Name	
.68	v20	Bank personnel are concerned about customers.	
.71	v21	Bank personnel are knowledgeable.	
.73	v22	Bank managers listen to customers.	
.74	v23	Good interaction between staff and customers.	
.74	v24	Bank personnel are friendly.	
.77	v25	Bank personnel give advice to customers.	
.84	v26	Bank personnel explain services to customers.	
.81	v27	Bank management provides consultation to customers.	
.77	v 28	Friendliness of tellers.	
.76	v29	Customers complaints are dealt with.	
.69	v30	I feel important as customer.	
.50	.50 v34 Special lines for savers.		
Eigenvalu Per cent Cumulativ	Eigenvalue = 12.11 Per cent of variance = 20.2 Cumulative variance explained = 20.2		

Table 8.1: Factor One: Customers' Satisfaction.

Factor Two: Advertising:

This factor accounts for 9.4 per cent of the explained variance. It includes five variables, that are related to bank

advertising methods and how respondents who are "banked with savings accounts" evaluate their importance in influencing their decision to open savings accounts. The five variables loaded with this factor are described in table 8.2. Advertising is one of the methods of informing consumers of banking services offered by the bank. Also, it is an important method of influencing consumers and competing in the market. This factor indicates that advertising is the second most important factor which has an influence on consumers of banking services in Iraq. In their studies Erol and El-Bdour (1989) and Yavas (1988) concluded that advertising was a factor in bank marketing and was perceived by consumers of banking services as of significant importance in influencing consumers banking decisions to select a bank and a service.

Table 8.2: Factor Two: Advertising.

Loading	Variable Code	Variable Name	
.88	v52	Bank advertises on T.V.	
.91 v53 Bank advertises on Radio			
.92	v54	Bank advertises in Newspapers	
.88 v55 Bank advertises in Magazines			
.49 v56 Personal contact by bank.			
Eigenvalu	ae = 5.64		
Per cent	of variance $= 9$.	.4	
Cumulativ	ve variance expla	ained = 29.6	

Factor Three: Benefits of Savings:

This factor accounts for 6.8 per cent of the explained variance, as shown in table 8.3. It includes seven variables, that are closely tied to the importance attached by respondents who are "banked with savings accounts" to the benefits, they seek from having savings accounts at commercial banks. In this factor Iraqi savers indicated the most important benefits they want to achieve from having a savings accounts at a commercial bank. The range of benefits sought by one customer to another is vary. However, this factor indicates that seven personal benefits from holding a bank savings account were given a priority by Iraqi bank customers, as shown in table 8.3. The findings of, for example, Yavas (1988) and Erol and El-Bdour (1989) suggested that some general benefits were sought by bank customers when they deal with a bank. However, their studies were not specifically carried out to investigate benefits of savings, but it mentioned some general benefits of using banking services. Therefore, it can be mentioned here that the attempt to explore these seven common benefits of savings, which are suggested in this study, is a more specific attempt to identify savers' motivations to hold a bank savings account.

Loading Variable Code Variable Name				
.75	.75 vl Meeting increasing living expenses.			
.65	ν2	Beating inflation.		
.48	v3	Family is protected.		
.60	v4	Building up a retirement fund.		
.73	.73 v5 Increasing standard of living.			
.57	.57 v6 Buying property.			
.45 v13 Covering marriage expenses.				
Eigenvalue = 4.08 Per cent of variance = 6.8 Cumulative variance explained = 36.4				

Table 8.3: Factor Three: Benefits of Savings.

Factor Four: Bank's Image:

As shown in table 8.4, this factor accounts for 4.5 per cent of the explained variance. It includes five variables, that are closely tied to the aspects of bank reputation and customer's confidence in bank. Also, the five variables loaded in this factor represent the most important confidence variables which may influence consumers' decision to open savings accounts at commercial banks. A bank's image in the market has a significant influence on consumers. Banks anywhere attempt to reflect the best image they can in the market, especially in countries where the competition between banks is very high. In this factor, Iraqi consumers of banking services indicated the most important aspects of bank image which they consider as important in influencing their perception of the bank. Previous studies of bank marketing extracted a similar factor, see for example Robertson and Bellenger (1977) and Erol and El-Bdour (1989).

Table 8.4: Factor Four: Bank's Image.

Loading	Variable Code	Variable Name
• 58	v47	Bank has large building.
.76	v 48	Bank is well-known.
.78	v49	Bank is well-established.
.68	v50	Bank has a good reputation.
.58 v51 Bank has many branches.		
Eigenvalue = 2.73 Per cent of variance = 4.5 Cumulative variance explained = 40.9		

Factor Five: Location Convenience:

This factor accounts for 3.6 per cent of the explained variance and includes three variables, as described in table 8.5, relative to the aspects of bank location and its convenience. Thus, this factor consists of the most important location variables, that are perceived by respondents who are "banked with savings accounts" as of significant importance in influencing their decision to open savings accounts at commercial banks. Bank customers in Iraq indicated that convenience of bank location near shopping centres, home, place of employment and overall good location of bank branches are considered as important issues which have an influence on their decisions. All the same, convenience of bank location was suggested by previous comparative studies of bank marketing in other countries as of significant importance in influencing consumers selection of a bank. In their studies LeBlanc and Nguyen (1988), Yavas (1988), Robertson and Bellenger (1977) and Erol And El-Bdour (1989) concluded that convenience of bank location near home, work and shopping centre is one of the factors which has an influence on consumers banking decisions.

	Table	8.5:	Factor	Five:	Location	Convenience.
--	-------	------	--------	-------	----------	--------------

Loading	Variable Code	Variable Name
.64 .86 .82 .49	v38 v39 v40 v41	Bank is located near shopping centre. Bank is located near home. Bank is located near place of employment. Location of branches.
Eigenvalue = 2.14 Per cent of variance = 3.6 Cumulative variance explained = 44.5		

Factor Six: Quality of Transactions:

This factor accounts for 3.4 per cent of the explained variance. Also, it consists of three variables related to the concept of the quality of savings accounts transactions. The three variables included in this factor reflect the role of some qualitative attributes of quality of savings accounts transactions in influencing respondents who are "banked with savings accounts" and their decision to open savings accounts at commercial banks. Table 8.6 provides a description for this factor, and the variables it includes. This factor indicates that consumers of banking services pay attention to the aspects of quality of bank transaction in terms of its accuracy and confidentiality. It is obvious that accuracy of bank transaction is important for a bank customer because he/she feels that a mistake make by the bank may result in a financial lost to him/her. In addition, it may destroy a customer's confidence in the bank and the quality of its services. Bank transactions are treated with a high level of confidentiality and that fact has pleased users of banking services throughout the history of banking. Therefore, the findings of the present study together with the findings of previous studies, such as Erol and El-Bdour (1989) and Arora, Cavusgil and Nevin (1985) confirmed the importance of the quality of banking services in terms of accuracy, confidentiality and ease of processing transactions to bank customers.

Table 8.6: Factor Six: Quality of Transactions.

Loading Variable Code Variable Name				
.73v35Accuracy of bank transaction79v36Confidentiality of bank transaction74v37Easily of processing bank transaction.				
Eigenvalue Per cent o Cumulative	= 2.02 f variance = 3.4 variance explai	ned = 47.9		

Factor Seven: Building Design:

This factor accounts for 2.9 per cent of the explained variance. It includes five variables, as shown in table 8.7, that are related to how respondents who are "banked with savings accounts" rate the importance of some bank building design attributes in influencing their decision to open savings account at a particular commercial bank. The findings of the present investigation with regard to the importance of this factor "bank's building design" confirmed the findings of the study carried out by Arora, Cavusgil and Nevin (1985). In their study they suggested that there are some similar aspects (variables) that are related to the design of the bank building and its convenience of significant importance in influencing consumers' perception of a bank.

Table 6.7: Factor Seven: Bullding Desi
--

Loading	Variable Code	Variable Name
.45	v42	Availability of parking facilities.
.53	v43	Building close to public
		transportation.
.65	v 44	Easy entrance to and exist from
1		bank building.
.52	v 45	Modern building design.
.64 v46 Availability of waiting areas.		
Eigenvalue = 1.71 Per cent of variance = 2.9 Cumulative variance explained = 50.7		

Factor Eight: Customer's Benefit:

As shown in table 8.8, this factor accounts for 2.8 per cent of the explained variance. It includes four variables related to the aspects of bank competition in terms of incentives, facilities, and promotions provided by the bank to customers. Also, how respondents who are "banked with savings accounts" perceived its importance in influencing their selection of a bank. This factor consists of some benefits (advantages) provided by the banks to customers. Factor three of this study's factor analysis includes some personal saving benefits sought by savers. The present factor (factor eight) consists of other types of benefits provided by the bank to its customers in terms of facilities, such as variety of services, regular bank statements, competitive charges and promotions which

are associated with general dealing with the bank. In other words, they are facilities provided to all users of banking services in general as well as to users of the service of savings accounts. In their study Arora, Cavusgil, and Nevin (1985) used factor analysis and concluded a similar set of customer's benefits provided by banks to their customers.

Table 8.8: Factor Eight: Customer's Benefit.

Loading	Variable Code	Variable Name	
.49 .65 .70 .69	v57 v58 v59 v60	Full-range of services. Bank provides regular statements. Bank has competitive charges. Bank provides different promotions.	
Eigenvalue = 1.66 Per cent of variance = 2.8 Cumulative variance explained = 53.5			

Factor Nine: Reasons for Saving:

Table 8.9 shows that this factor accounts for 2.5 per cent of the explained variance. It includes three variables, which represent the most important reasons, as indicated by respondents who are "banked with savings accounts", for having savings accounts at commercial banks in Iraq. Thus, this factor identified the most important reasons which motivated Iraqi consumers of banking services to hold a savings account at a commercial bank. It underlined that one of the reasons to save money by respondents is "to start a new business". In addition, respondents who are "banked with savings accounts" indicated that holding a savings account is a good method to keep surplus money together with being personally motivated to save are prime reasons which have influenced their saving behaviour.

Loading	Variable Code	Variable Name		
.45 .61 .74	v12 v16 v19	Starting a new business. Good method to keep money. Own decision (personally motivated).		
Eigenvalue = 1.50 Per cent of variance = 2.5 Cumulative variance explained = 56.0				

Table 8.9: Factor Nine: Reasons for Saving.

Factor Ten: Influence of Relationship:

Table 8.10 shows that this factor accounts for 2.4 per cent of the explained variance. It consists of two variables related to the role of reference groups in influencing respondents who are "banked with savings accounts" decisions to open savings accounts at commercial banks. It was concluded in chapter four of the present study that consumer banking behaviour can be influenced by some external factors. The influence of relationship, such as family and friends, is an important external factor on consumer banking behaviour. Therefore, the finding here confirmed the assumed importance of such relationships in influencing the behaviour of Iraqi consumers of banking services. In Iraq, as one of the Middle Eastern countries, social and family relationships are still very strong. Therefore, consumers are very likely willing to listen to their family and friends advice before making their decisions to buy a product or use a service. In his study of bank customers in Saudi Arabia, Yavas (1988) concluded that a friend's recommendation is an important factor to be considered by consumers.

Loading	Variable Code	Variable Name	
.79 .78	v17 v18	Advice from friends. Advice from family members.	
Eigenvalue = 1.44 Per cent of variance = 2.4 Cumulative variance explained = 58.4			

Table 8.10: Factor Ten: Influence of Relationship.

Factor Eleven: Bank Loyalty:

This factor accounts for 2.2 per cent of the explained variance. It includes two variables, as shown in table 8.11, related to the aspects of bank loyalty, i.e., customer loyalty to a particular bank, in terms of its importance in influencing respondents who are "banked with savings accounts"' decision to open and maintain their savings accounts at a particular commercial bank. The subject of customer's loyalty to the bank depends on many aspects related to the relationship between the bank and the and Malhorta (1987) customer. Jain. Pinson investigated the difference between two groups, the bank loyal and non-loyal customers. Their findings indicated that the bank loyal group places much greater emphasis on the human aspects of banking while the nonloyal group is swayed by the economic rationale. Thus, the non-loyal customers are concerned about the financial benefits and services they receive from the bank, and if they feel that there is no economic rationale to continue dealing with the same bank they will not hesitate to change it. However, the loyalty factor extracted by this study's factor analysis consists of two types of bank loyalty variables. The first variable "I feel I own part of the bank" can be considered as one of the human aspects of bank-customer

relationship, which is of interest to bank loyal customers. The second variable "using other services at this bank" can be considered as one of the economic rationale to make a customer continue his/her dealing with the same bank.

Table 8.11: Factor Eleven: Bank Loyalty.

Loading	Variable Code	Variable Name	
.65 .73	v11 v14	I feel I own part of the bank. Using other services at this bank.	
Eigenvalue = 1.31 Per cent of variance = 2.2 Cumulative variance explained = 60.6			

Factor Twelve: Access Convenience:

Table 8.12 shows that this factor accounts for 2.1 per cent of the explained variance. It includes three variables related to the subject of respondents who are "banked with savings accounts" access to their savings accounts, and how they evaluate the importance of their access to their savings in making their decision to open savings accounts at a particular commercial bank. Arora, Cavusgil and Nevin (1985) concluded a similar factor, and they labelled it as "accessibility". Thus, bank customers seek a convenience of access to the money they deposited at the bank, and this watter has an important influence on consumers decisions to select a bank. In the developed countries, the availability of the new banking technology, which provides easy access for bank customers to their money, is an important factor on consumers' bank selection decisions. However, in Iraq such facilities are not available to bank customers and other traditional methods of providing easy access to customers, such as many cashiers at a time,

are used by the banks to provide acceptable level of access convenience to customers.

Loading	Variable Code	Variable Name	
.47	v7	Money is in a safe place.	
.74	v9	I can easily deposit savings at	
		the bank.	
.68	v10	I can easily withdraw money	
		from the bank.	
Eigenvalu Per cent Cumulativ	e = 1.23 of variance = 2.1 e variance explai	ined = 62.6	

Table 8.12: Factor Twelve: Access Convenience.

Factor Thirteen: Waiting Time:

This factor accounts for 1.9 per cent of the explained variance, as shown in table 8.13. It includes two variables related to the waiting time to be spent by respondents who are "banked with savings accounts" to process a single savings account transaction and how they perceive the importance of the length of time consumed to process a transaction at the bank in influencing their decision to open savings account at a particular commercial bank. One of the advantages of the availability of the new banking technology, such as the Automatic Teller Machines, is providing easy access to customers. This factor includes a second advantage which is provided by the new banking technology, that is, reducing the waiting time to be spent by customers each time they visit the bank to process a bank transaction. However, not all bank transactions are processed by using banking technology. Thus, even in the developed countries, bank customers are required to visit their bank and spend some waiting time before being served. In this factor, the emphasis is on

two connected types of waiting times. The first is the length of waiting time before being served and the second is the length of waiting time required to process the bank transaction itself. The total of the two waiting times represents the time which a customer needs to wait in order to finish a bank transaction. Bank customers attache a significant importance to the aspects of waiting time. Therefore, banks should attempt to make waiting times shorter, because the shorter the waiting time the more likely that a customer feels satisfied.

Table 8.13: Factor Thirteen: Waiting Time.

Loading	Variable Code	Variable Name	
.79 .78	v31 v32	Length of time required to process transaction. Length of waiting time before being corved	
Eigenvalue = 1.13 Per cent of variance = 1.9 Cumulative variance explained = 64.5			

8.3 Testing Research Hypothesis:

The related research hypothesis, as stated in its null form, to be tested in this chapter is the following: HO : There are no significant similarities between respondents who are "banked with savings accounts" in terms of their attitudes towards the service of savings accounts provided by commercial banks in Iraq.

The Kolmogorov-Smirnov One-Sample Goodness-of-Fit Test was used to test the above research hypotheses. The main reasons for choosing this statistical test to test the research hypotheses of this chapter are the following:

1. The Kolmogorov-Smirnov Test is a statistical test which can be applied in a one sample case. Thus, the sample used in the factor analysis of this chapter was a one sample case, i.e., the random sample of "banked with savings accounts" only.

2. The Kolmogorov-Smirnov Test is a statistical test which can be applied with data measured in an ordinal scale. Thus, the data gathered from the sample of "banked with savings accounts" and analysed in this chapter, was measured in an ordinal scale, i.e., respondents were asked to answer questions measured by 5-Point Likert Type Scale.

3. In addition to that, Siegel (1956) suggests that it would seem that in all cases where it is applicable, the Kolmogorov-Smirnov test is the most powerful goodness-of-fit test of all the one-sample case statistical tests.

According to Kolmogorov-Smirnov test the value of the maximum deviation (D) for each variable is equal to the absolute value of the maximum deviation between the observed cumulative proportion [Sn (x)] and the theoretical cumulative proportion [Fo (x)].

i.e., D = maximum [Fo(x) - Sn(x)]

Where: n = number of respondents

x = score of rating for each category within the variable.

Using the above formula the value of (D) for each variable used in this analysis was calculated, as explained in Appendix B, and given in Table 8.14.

Variable No	D-Value	Variable No	D-Value	
vl	.350 a	v34	.321 a	
v2	. 275 a	v 35	.486 a	
v 3	. 532 a	v 36	.453 a	
v 4	.339 a	v37	.419 a	
v 5	. 268 a	v 38	.321 a	
v 6	. 283 a	v39	. 290 a	
v7	.476 a	v 40	.294 a	
v 8	.272 a	v41	.336 a	
v 9	.287 a	v42	.328 a	
v10	.325 a	v43	.374 a	
v11	.041 d	v 44	.430 a	
v12	.321 a	v45	. 268 a	
v13	.328 a	v46	.404 a	
v14	.060 d	v 47	.231 a	
v 15	.095 b	v 48	.283 a	
v 16	.386 a	v49	.249 a	
v17	.076 c	v50	.452 a	
v18	.231 a	v51	.351 a	
v19	.509 a	v52	.230 a	
v 20	.230 a	v53	.162 a	
v21	.337 a	v54	.207 a	
v 22	.260 a	v 55	.204 a	
v23	.208 a	v 56	.181 a	
v24	.253 a	v57	.389 a	
v 25	.280 a	v58	.264 a	
v26	.215 a	v59	.223 a	
v 27	.223 a	v 60	.392 a	
v28	.276 a			
v 29	.234 a			
v 30	.207 a			
v 31	.189 a			
v 32	.214 a			
v33	.378 a			
a Signi	a Significant level less than .01			
b Signi	b Significant level between .0105			
c Significant level between .0510				
d Not significant at all				

Table 8.14: D-values of the Kolmogorov-Smirnov Goodness-of-Fit Test for the Attitudinal Variables.

According to the table of Critical Values of (D) in the Kolmogorov-Smirnov One-Sample Goodness-of-Fit Test, the critical value of (D) for large sample under significant level of .01 is equal to:

 $1.63 / \sqrt{(N)}$

Where N = The number of cases.

In this case for N = 265 the critical value of (D) is equal to .100. In other words, for N = 265, D greater than .100 has an associated probability under HO of p less than .01. Calculated values of (D) as shown in Table 8.14 for most variables exceed the critical value of (D) according to the Kolmogorov-Smirnov Test, therefore, the null hypothesis was rejected and the alternative hypothesis was accepted.

Table 8.14 shows that out of the 60 attitudinal variables used in the analysis, 56 were at significant level far beyond .01, one variable was at significant level between .01-.05, and 3 variables were at significant level over .05. In other words, out of the 60 variables used, 57 variables were of significant importance, and 3 variables were not significant. These results conclude that, there are some generally agreed attitudinal variables perceived by respondents who are "banked with savings accounts" as of significant importance in influencing their decisions to open savings accounts at commercial banks. In other words, the attitudes of respondents who are "banked with savings accounts" towards the variables related to the services of savings accounts provided by commercial banks are similar.

8.4 Conclusions:

The main objective of this chapter was to identify a set of factors which are capable of explaining, how respondents who are "banked with savings accounts" perceive the importance of different attitudinal variables in influencing their decisions to open savings accounts at commercial banks.

The findings of Factor Analysis - Principle Components Analysis identified thirteen factors, which explain those

respondents' attitudes towards the service of savings accounts provided by commercial banks in Iraq. In other words, a number of 57 variables, out of 60 attitudinal variables, were loaded with the factors extracted. Therefore, the reason for using factor analysis in this investigation was achieved. Factor analysis extracted a minimal number of saving behaviour factors to replace the larger number of independent variables, with minimum loss of information.

In addition, the Kolmogorov-Smirnov One-Sample Goodness-of-Fit Test confirmed that there were some generally agreed attitudinal variables of significant importance in influencing respondents who are "banked with savings accounts" decisions to open savings accounts at commercial banks. In other words, respondents who are "banked with savings accounts", in general, share the same attitudes towards the service of savings accounts provided by commercial banks and there are similarities in their saving behaviour.

Finally, two tests for FA reliability were used, these are Bartlett's Test of Sphericity and Kaiser-Meyer-Olkin, which indicated that this study's factor analysis is reliable and the correlation matrix between the variables can be used in this study's factor model.

CHAPTER NINE

THE FINDINGS OF DISCRIMINANT ANALYSIS AND T-TEST

.

- 9.1 INTRODUCTION.
- 9.2 INTERPRETATION OF THE FINDINGS.
- 9.2.1 THE FINDINGS OF DISCRIMINANT ANALYSIS DISCRIMINATING BETWEEN RESPONDENTS ACCORDING TO THEIR SAVING DECISIONS "BANKED WITH SAVINGS ACCOUNTS VS. NON-BANKED".
- 9.2.2 THE FINDINGS OF T-TEST.
- 9.2.2.1 RESPONDENTS' SAVING HABITS (GOOD VS. POOR SAVERS).
- 9.2.2.2 RESPONDENTS' AGE GROUP.
- 9.2.2.3 RESPONDENTS' FAMILY SIZE.
- 9.2.2.4 RESPONDENTS' INCOME GROUP.
- 9.2.2.5 RESPONDENTS' EDUCATIONAL LEVEL.
- 9.3 COMPARISON WITH OTHER STUDIES OF CONSUMERS' SAVING BEHAVIOUR.
- 9.4 CONCLUSIONS.
9.1 Introduction:

The main objective of this chapter is to establish the extent to which dissimilarities exist between Iraqi consumers in terms of the rates of importance they attached to the variables related to the service of savings accounts provided by commercial banks, if divided according to their: (a) saving decisions "banked with savings accounts" vs. "non-banked"; (b) saving habits "good vs. poor savers"; (c) age group; (d) family size; (e) income group; and (f) educational level.

In other words, the purpose is to establish the extent to which differences exist between respondents' attitudes according to the above mentioned characteristics. Therefore, Two-Group Discriminant Analysis and T-test were utilised in order to achieve the stated above objective.

9.2 Interpretation of the Findings:

<u>9.2.1 The Findings of Two-Group Discriminant Analysis:</u> <u>Discriminating Between Respondents According to their</u> <u>Saving Decisions "banked with savings accounts" vs.</u> <u>"non-banked"</u>:

The main purpose of this section is to establish the extent to which dissimilarities exist between consumers who are "banked with savings accounts" and "non-banked" in terms of their attitudes towards the variables related to the service of savings accounts provided by commercial banks in Iraq.

The primary data collected from a number of 621 respondents, out of the original sample of 800 respondents, was analysed in this

section. A Step-Wise Discriminant Analysis for Two-Groups of Wilks' lambda (U statistic) method was used in this section. The Wilks' lambda (U statistic) method was adopted because, it helps to include only the significant variables in the process of groups classification, and eliminate the insignificant variables in order to provide a model contains the optimum set of variables.

The Two-Group Discriminant Analysis, as one of the dependent methods of multivariate analysis, can be used when the study has one single dependent variable, which is dichotomous. Thus, in this study the variable respondent saving decisions to be saver, i.e., banked with savings accounts, or non-saver, i.e., non-banked, is the dichotomous variable.

Sheth (1971) mentioned that the primary objective of Discriminant Analysis is to correctly classify entities into mutually exclusive and exhaustive classes or groups. Grimm (1974) demonstrates the case for choosing Discriminant Analysis among other dependent methods by explaining that: Discriminant Analysis classifies individuals or objects into one of two or more exclusive Regression groups, in contrast to Analysis and Multiple Classification Analysis (MCA) that position individuals or objects continuum. The dependent is on a variable categorical in Discriminant Analysis. While, it is continuous for Regression Analysis and MCA. The objective of Discriminant Analysis is to determine the group or category to which an individual belongs.

Based on the above assumptions, this section Discriminant Analysis objective is to classify respondents into two mutually exclusive and exhaustive groups, i.e., in this case consumers who are "banked with savings accounts" and "non-banked". Thus, the single dependent categorical dichotomous variable required by

Discriminant Analysis for the purpose of classification in this investigation is variable (pl), i.e., respondents' saving decisions (hold / not hold a savings account). Respondents were asked to indicate if they presently hold a savings account at a commercial bank. Those who replied "Yes" were classified as "banked with savings accounts", and those who replied "No" were classified as "non-banked". Based on the above, the entail Discriminant Analysis classification results shown in Table 9.1 indicated that out of the 621 cases used, 265 respondents were "banked with savings accounts"

Table	9.1:	: Number	• of	Cases	by	Groups.
-------	------	----------	------	-------	----	---------

Variable Pl	Label	Number of Cases
0	Yes	265 "banked with savings accounts"
1	No	356 "non-banked"
Total	=	621

A number of 65 independent predictor variables were used in the analysis, in order to establish which are of them are the most important variables in distinguishing between the two groups of respondents "banked with savings accounts" and "non-banked". Since the analysis contained only two groups, i.e., "banked with savings accounts" and "non-banked", the Discriminant Analysis produced only one function. The results of the canonical discriminant function of the analysis, as shown in table 9.2, indicated that both Wilks' Lambda .84 and its associated Chi-square statistic 85.23, which is significant at .0000 significance level, confirmed that there is a

significant discrimination between the two groups of respondents "banked with savings accounts" and "non-banked". Also, these results confirmed that the group mean values for both groups are not equal.

Table 9.2 The Canonical Discriminant Function.

Function 1	Eigenvalue .1452	% of variance 100	Cumulative %	Canonical correlation .621
After	Wilks'	Chi-squared	D.F 18	Significan-
Function O	lambda .84	85.23		ce .0000

However, the results of the Discriminant Analysis shown in table 9.3 indicated that a number of 18 variables are to be considered as having the most discriminating power between the two groups.

Table 9.3: Standardized Canonical Discriminant Function Coefficients and Group Means for the Discriminating Variables "banked with savings accounts" vs. "non-banked".

.

No	Variable Name	Rank	Function 1	Grou Banked	p Mean Non-banked
v15	Higher interest on savings.	1	5032	2.74	2.39
v53	AD on Radio.	2	4823	2.57	2.35
v61	I am a good saver.	3	.3987	2.14	2.37
v63	To buy anything other than a house on credit is unwise.	4	.3651	2.05	2.37
vl	Savings help to meet increasing living expenses.	5	2968	1.93	1.80
v44	Easy entrance to and exit from bank building.	6	.2901	1.72	1.88
v27	Bank management is always available for consultation.	7	2836	2.37	2.19
v18	Advice from family members.	8	2517	2.33	2.15
v12	Savings help to start a new business.	9	.2422	1.98	2.08
v64	A person can save a lot of money by looking for the lowest possible prices when shopping.	10	1988	2.19	2.18

Table 9.3 (Continued)

No	Variable Name	Rank	Function 1	Grou Banked	p Mean Non-banked
v40	Bank is located near	11	.1892	2.02	2.17
v20	Bank personnel are concerned about their	12	.1773	2.41	2.39
v3	customers. Savings help to feel	13	.1741	1.38	1.48
	family is protected in emergencies.				
v10	I can easily withdraw	14	.1703	2.03	2.13
v16	Savings account is a	15	1521	1.81	1.79
	good method to keep surplus money.				
v6	Savings help in buying property for the family.	16	1432	2.14	1.99
v50	Bank has a good	17	1395	1.59	1.60
v32	Length of waiting time before being served.	18	.1362	2.40	2.51

Table 9.3 includes the ranking order of the discriminating variables, which are extracted by Discriminant Analysis, together with their standardized coefficients, and group means. The ranking order indicates that variables with large coefficients (in absolute value) are contributing more to the overall discriminant function. In other words, they have more discriminating power to discriminate between the two groups. For example, variable (v15) has the highest contribution in the discriminant function in absolute value (function 1 = -.5032), followed by variable (v53), and so on. Thus,

the higher the standardized coefficient value, the more contribution in the discriminant function.

Norusis (1985) suggested that the actual signs of the coefficients of variables are arbitrary. Thus, for example, the negative coefficient of the variables shown in table 9.3 could just as well be positive, if the signs of other coefficients were reversed.

Furthermore, the value of the group mean for each variable indicates the group, which attached more importance to that variable. Thus, as the group mean value of a variable for respondents "banked with savings accounts" is less than that of "non-banked", that means respondents "banked with savings accounts" attached more importance to that variable than "non-banked". On the other hand, as the group mean value of a variable for "non-banked" is less than that of "banked with savings accounts", that means "non-banked" attached more importance to that variable than "banked with savings accounts". Therefore, it should be mentioned here that, the variables used in this study were measured by a 5-point Likert type scale, from 1 = very important to 5 = not important at all. Therefore, the lower the group mean value for a variable, the more importance attached to it by respondents.

In addition, the group mean values shown in table 9.3 explain that the group of "banked with savings accounts" attached more importance to 9 variables while the group of "non-banked" attached more importance to 9 other variables. Based on the above, the results of the DA, as shown in table 9.3, are to be discussed for each discriminating variable as follows:

Consumers' Allitudes Towards Savings Accounts:

Table 9.3 shows that seven variables, that are related to consumers' attitudes towards savings accounts, were extracted by Discriminant Analysis as variables of significant ability in discriminating between the two groups "banked with savings accounts" and "non-banked", and they are the following:

The variable (v1) "savings help to meet increasing living expenses" was ranked fifth in terms of its importance in discriminating between the two groups. However, its group mean values suggest that the group of "non-banked" attached more importance to the idea that savings help to meet increasing living expenses than the group of "banked with savings accounts". However, the rate of importance, i.e., the group means, indicated that both groups attached a high degree of importance to this variable. In other words, both groups think that savings help in meeting the increasing living expenses. It is true that the group of "nonbanked" attached more importance to this saving motivation than the group of "banked". The reason for such a difference lays in the fact that the group of "banked with savings accounts" are actual savers and they know that, for example, the real value of the interest they receive on their savings has little effect in overcoming the increase in the living expenses because the rate of inflation is much higher than the base rate of interest in Iraq. However, they indicated that they consider depositing their money at the bank and keeping it for future use and even receiving some interest on it is better than spending it or receiving nothing should they keep the money in their homes. The group of "non-banked", on the other hand, thought, for example, that the interest a saver may receive will help considerably in meeting the increase in future living expenses

merely because they are not actual savers and their opinion is based on their common sense rather than on real experiences.

The variable (v3) "feel family is protected in emergency" was ranked thirteenth in terms of its importance in discriminating between consumers who are "banked with savings accounts" and "nonbanked". However, its group mean values suggest that the group of "banked with savings accounts" attached more importance to this variable than "non-banked". In other words, the group of "banked with savings accounts" think that savings help them in feeling the family is protected in case of emergencies than think so the group of "non-banked". As shown in table 9.3 both groups attached a high rate of importance to the reason for saving, "feeling the family is protected in case of emergencies". In Iraq, people save the surplus money they can spare today to meet one of those emergencies which may happen to a member of the family, particularly since health insurance and other types of insurance, such as life and accident insurance, are not in common use. Saving through savings accounts or simply keeping money (cash) at home ready for use in case of emergency is generally considered by the Iraqi consumer as the best insurance they may have. This explained why this reason for saving, as shown in table 9.3, received the highest rate of importance by respondents of both groups in comparison with all other variables. One can conclude that this is the strongest reason for saving in Iraq.

:

The variable (v6) "buying property for the family" was ranked sixteenth in terms of its importance in discriminating between the two groups of respondents "banked with savings accounts" and "nonbanked". Also, its group mean values suggest that the group of "nonbanked" attached more importance to the idea that savings help in

buying property for family accommodation than the group of "banked with savings accounts". By analysing the group mean values for this variable one can conclude that this reason for saving received a moderate rate of importance by both groups. In fact, the group of "banked with savings accounts", as actual savers, attached less rate of importance to this reason because they may think that savings provide little help in buying a property for family accommodation. However, this can be real strong reason for saving if only the amount of savings is large enough to cover the cost of buying a property. In reality such a case is rare and most people need a mortgage to cover the price of buying a property for family accommodation.

The variable (v12) "starting a new business" was ranked ninth in terms of its importance in discriminating between respondents who are "banked with savings accounts" and "non-banked". However, its group mean values suggest that the group of "banked with savings accounts" attached more importance to the idea that savings help to start a new business than "non-banked". The possible reason for attaching a higher rate of importance for this reason for savings, i.e., starting a new business, by the group of "banked with savings accounts" due to the fact that they are actual savers and some of them perhaps used their savings when they started their businesses. Most Iraqi owners of small business, in particular, depend on their savings to finance their businesses. Thus, because Iraqi banks are reluctant to provide credit to small businesses. It can be said that banking facilities, such as credit, are mostly out of reach for owners of small businesses. This is because the banks terms for credit are very tight and mostly designed to be available for medium and large businesses.

The variable (v16) "a good method to keep surplus money", as one of the motivations to have a savings account, was ranked fifteenth in terms of its importance in discriminating between the two groups of respondents "banked with savings accounts" and "nonbanked". However, its group mean values suggest that both groups attached a high degree of importance for this motivation. In other words, although the group mean value for "non-banked" is marginally less than that of "banked with savings accounts", i.e., the group of "non-banked" attached more importance to it than "banked with savings accounts", the difference is still very small. This finding underlined the fact that both groups acknowledged that having a savings account is a good method to keep surplus money. Also, it is a good method to keep surplus money because the money saved at the bank is more likely to be kept longer as savings for future use in a secure place rather than money saved, for example at home, which are subject to possible spend or even theft. However, the difference in the rate of importance attached by the two groups is very small. Therefore, it emphasised the fact that both groups evaluated holding a savings account at a commercial bank as a good method to keep surplus money.

The variable (v32) "length of waiting time" is one of the service attributes. However, it has the lowest rank (eighteenth) in terms of its importance in discriminating between consumers who are "banked with savings accounts" and "non-banked". Also, its group mean values suggest that the group of "banked with savings accounts" attached more importance to this service attribute than "non-banked". This finding refers to the fact that the group, "banked with savings accounts" because, they experienced the importance of waiting time and its effect on their satisfaction,

they attached more importance to this variable than the group of "non-banked". Thus, the group of "banked with savings accounts" attached higher rate of importance to the length of waiting time before being served than the group of "non-banked" mainly because they are actual users of the service of savings accounts provided by commercial banks. In fact, customers of banking services in Iraq suffer from the problem of long waiting time they have to spend every time they visit their banks mainly because of the lack of new banking technology and poor banking services in Iraq.

The variable (v10) "easily withdraw savings" is one of the service attributes. It was ranked fourteenth in terms of its importance in discriminating between consumers who are "banked with savings accounts" and "non-banked". However, its group mean values suggest that the group of "banked with savings accounts" attached more importance to this service attribute than "non-banked". In fact, such a conclusion is expected, because consumers who are "banked with savings accounts" experienced the importance of an easy access to their savings whenever they need to withdraw some money. In addition, this variable is a reflection of the importance attached by actual customers (in this case consumers who are "banked with savings accounts") to the technological and other facilities provided by banks to its customers in satisfying their needs to have an easy access to their money. Thus, at the present time, banks have attempted to use the best of what technology can provide in order to satisfy their customers. Therefore, new technology such as, for example, the Automated Teller Machines serves as an advanced method to provide an easy access for customers to their money which they have deposited at the bank, and it is a service now highly appreciated by them.

Consumers' AULitudes Towards Bank Attributes:

Table 9.3 indicates that seven variables related to the aspects of bank attributes and consumers' attitudes towards them, were extracted by Discriminant Analysis as variables of significant ability in discriminating between consumers who are "banked with savings accounts" and "non-banked". These variables are the following:

The variable (v20) "bank personnel concerned about customers" is related to the subject of bank-customer relationship as one of the bank attributes. It was ranked twelfth in terms of its importance in discriminating between the two groups. However, its group mean values suggest that both groups "banked with savings accounts" and "non-banked" attached nearly the same rate of importance to this bank attribute. In other words, although it seemed that the group of "non-banked" attached marginally more importance to this attribute than the group of "banked with savings accounts", the difference is still very small between the two groups. Both groups of respondents indicated that they viewed "customer care" as expressed by the bank's staff, as an important aspect of bank-customer relationship. In fact, in the banking sector, where the service is the main commodity, customers want to be served with style and respect. This is the essence of old and modern banking. However, in Iraq such a banking atmosphere is still far from being attained and the bank-customer relationship is very poor for a number of reasons as described in detail in chapter two of the present investigation.

The variable (v27) "availability of bank management for consultation" is also related to the subject of bank-customer relationships (personal aspects). It was ranked seventh in terms of

its importance in discriminating between the two groups. However, its group mean values suggest that the group of "non-banked" attached more importance to the availability of bank management for consultation than the group of "banked with savings accounts". In other words, the availability of management consultation services seemed to be more important for "non-banked" than for "banked with savings accounts". Such a finding explains the fact that noncustomers "non-banked" have less knowledge about the services and some hesitation in dealing with banks. Therefore, they prefer the existence of such a facility. Existing customers, such as "banked with savings accounts" are less concerned about the availability of such a consultation since they believe that they are in possession of enough information about the services provided by the bank.

The variable (v53) "AD on Radio" is related to the subject of bank advertising techniques. However, it was ranked second in terms of its importance in discriminating between consumers who are "banked with savings accounts" and "non-banked". Also, its group mean values suggest that the group of "non-banked" attached more importance to using Radio stations by banks in advertising for their services than the group of "banked with savings accounts". In fact, this finding is related very closely to this research case study, i.e., the Iraqi consumers. Thus, Radio stations are still the dominant tool in public communications because they broadcast for longer hours (day and night), and can be heard in all parts of the country. Also, it is cheaper for people to buy a Radio than, for example, a T.V. set.

The variable (v50) "reputation of the bank" is one of the aspects of consumers' confidence in a bank. However, it has the second lowest rank (seventeenth) in terms of its importance in

discriminating between consumers who are "banked with savings accounts" and "non-banked". Its group mean values suggest that both groups "banked with savings accounts" and "non-banked" attached a great importance to bank reputation (the difference is very marginal). This finding confirms that in Iraq consumers consider a bank's reputation as one of the most important bank attributes. This finding is similar to the findings of previous similar studies carried out in other environments (countries) which suggested that a bank's reputation is considered as an important attribute by consumers as mentioned earlier in chapter three of this investigation.

The variable (v40) "bank located near place of employment" is one of the convenience of bank location variables. It was ranked eleventh in terms of its importance in discriminating between consumers who are "banked with savings accounts" and "non-banked". However, its group mean values suggest that the group of "banked with savings accounts" attached more importance to the idea that a bank is located near place of employment than "non-banked". Such a finding refers to the fact that consumers who are "banked with savings accounts", as actual users of the service in Iraq, concluded by their own experience that banks located near their place of employment are more convenient in terms of location. In other words, banks located near the place of the source of their income, are more convenient for the purposes of depositing or withdrawing money from the bank. All the same, the findings of previous similar studies carried out in other countries, as mentioned in chapter three. indicated that consumers consider banks located near their place of employment is more convenient.

The variable (v15) "higher interest on savings" is one of the competition factors, which may influence consumers decision to open savings accounts at a particular commercial bank. However, it seemed that this variable has the highest importance in discriminating between consumers who are "banked with savings accounts" and "nonbanked". Its group mean values suggest that the group of "nonbanked" attached more importance to higher interest on savings than the group of "banked with savings accounts". The obvious reason for such a finding is that, consumers who are "banked with savings accounts", in general and especially in this case study, i.e., the Iraqi consumers, seek a safer place to deposit their surplus money in, rather than to gain interest on their savings. Such a market situation is a generation of two major market characteristics: Firstly, consumers who are "banked with savings accounts". generally, are people with a surplus money who do not have the opportunity to invest it, or are people who seek only a safer place to keep their surplus money. Even so, there are some consumers who are "banked with savings accounts" who wish to gain interest as well. Secondly, the rate of interest in this case study, i.e., the rate of interest in Iraq, is very low if compared with any rate of return that can be gained from investing in business. Thus, this finding explains why both "banked with savings accounts" and "nonbanked" attached some importance to the rate of interest on savings. In addition to the above, in most Islamic countries a fixed interest rate is perceived with caution for religious reasons. In other words, because in Islam receiving interest on savings or lending is not allowed, many people perceive the fixed rate of interest as not important. In fact, some actual holders of savings accounts in Iraq, as is the case in most Islamic countries, used to ask the bank not

to give them any interest on their savings, because they sought only a safe place to put their surplus money. In his study of consumer banking behaviour in Saudi Arabia, Yavas (1988) suggested that the rate of interest has the least important effect in influencing consumers' saving decisions. This does not suggest that consumers of banking services in Islamic countries are not profit motivated. Erol and El-Bdour (1989) indicated that customers of Islamic banks in Jordan are profit motivated as long as the rate of return they receive is compatible with Islamic legal doctrines. In other words, according to Islam receiving a fixed in advance rate of interest on savings is not permitted, but receiving a fair unfixed in advance rate of return on investment is acceptable. This is the fundamental idea behind the newly established Islamic banks in some Islamic countries. However, in Iraq there is not-as-yet an Islamic bank, and only traditional commercial banks are providing commercial banking services for consumers. Therefore, customers of commercial banks in Iraq who do not want to receive the fixed rate of interest paid by banks usually declare that they do not want to receive any return on their savings. Other Iraqi bank customers who have no objection towards the fixed rate of interest receive their interest as usual.

The variable (v44) "easy entrance to and exit from bank building" is related to the subject of the attributes of bank building design. However, it was ranked sixth in terms of its importance in discriminating between the two groups. Its group mean values suggest that the group of "banked with savings accounts" in Iraq, as they are actual customers, attached more importance to this attribute of bank building design than "non-banked". That is mainly because they are the segment of consumers who know the importance of bank building design, and the facilities it provides for customers.

Thus, consumers who are "banked with savings accounts" are actual customers, and their opinions on the effect of bank building design on their selection of a bank to put their savings in is a product of real experience. Thus, the findings of previous similar studies, as mentioned in chapter three, suggested that bank customers give more importance to the design of the bank building, when they select a bank.

Consumers' Financial Psychographic Characteristics:

As shown in table 9.3 four variables, that are related to consumers' financial psychographic characteristics, were extracted by Discriminant Analysis as variables of significant ability in discriminating between consumers who are "banked with savings accounts" and "non-banked". These four variables are the following:

The variable (v18) "advice from family members" is an indication used to show the importance of the family, as one of the reference groups, in influencing consumers' saving decisions (hold / not hold a savings account). Table 9.3 indicates that this variable has the eighth rank in terms of its importance in discriminating between the two groups. But, its group mean values suggest that the group of "non-banked" attached more importance to the advice of family members in influencing their saving behaviour than the group of "banked with savings accounts" in Iraq. The group of "banked with savings accounts" attached a moderate importance to the advice of family members in influencing their saving behaviour. This finding concludes that one of the important financial psychographic characteristics of the actual users of the service of savings accounts provided by commercial banks is that they are more independent in making their financial decision than appeared to be the group of "non-banked" in Iraq. In fact, most previous studies,

229

L

as mentioned in chapter three, indicated that users of banking services are more independent in making their financial decisions than non-users.

The variable (v61) "I am a good saver" was used to let each respondent evaluate his / her saving behaviour. It was ranked third in terms of its importance in discriminating between the two groups. However, as expected, consumers who are "banked with savings accounts" expressed more agreement with this statement than "nonbanked". It should be added that both groups "banked with savings accounts" and "non-banked" expressed a mild agreement with this statement. Such a conclusion indicates that both groups are dissatisfied with their saving behaviour. In other words, in the case of the group of "banked with savings accounts" they evaluate themselves as not good enough as savers and they wish to be better savers in the future. In the case, i.e., the group of "non-banked", they know that they are not good savers in terms of having savings accounts at banks, but their group mean value suggests that some of them are actual savers. That is because some of them indicated that they are good savers, when they were questioned if they consider themselves as good savers. Thus, the methods of saving they used were different. This is to say that in Iraq not all savers save through having a savings account at banks. Thus, there are other popular methods to save money rather than putting them in savings accounts or investing them in properties or businesses. In other words, many people prefer to put their money in a secure place in their homes or transform it into gold, and other jewelry, rather than putting it into savings accounts or investing it.

The variable (v63) "buying other than a house on credit is unwise" was used to provide more insights about respondents

financial behaviour. It was ranked fourth in terms of its importance in discriminating between the two groups of "banked with savings accounts" and "non-banked". However, its group mean values suggest that both groups attached a moderate degree of agreement with this statement. In other words, they think that this statement is an extreme opinion. That is, they think that, it is better not to buy anything except a house on credit but some of them think, that in some circumstances, one can buy, for example, a car on credit. Such a conclusion indicates that one of the psychographic characteristics of both groups is they are generally cautious people in terms of borrowing money. In other words, they think that borrowing should be the last recourse especially when it comes to borrow money for reasons other than buying a house for family accommodation. By analysing the group mean values of both groups it can be concluded that the group of "banked with savings accounts" appeared to be more cautious regarding the use of credit, than their counterpart the group of "non-banked".

Finally, the use of the variable (v64) "buying in lowest prices saves money" was another attempt to discover more about consumers' financial thinking and behaviour. However, it was ranked tenth in terms of its importance in discriminating between the two groups. Its group mean values suggest that both groups, "banked with savings accounts" and "non-banked", attached a moderate degree of agreement with this statement. In addition, the difference between the two groups mean values were very marginal. By looking at the group mean values one can conclude that both groups, "banked with savings accounts" and "non-banked", gave the price of a product moderate importance when they decided to buy it. In other words, they viewed the price of a product as one factor affecting their

buying decisions, in addition to other characteristics, such as the quality of the product.

Finally, the findings of the DA reported in this section can summarised, as shown in table 9.4, in order to indicate the most important dissimilarities between the two groups of respondents "banked with savings accounts" and "non-banked".

Table 9.4: Summary of dissimilarities between "banked" and

"non-banked" respondents.

Banked with savings accounts	Non-banked
 View themselves as good savers	 Acknowledged that they
and cautious in terms of	are not good savers. More concerned about the
borrowing money. Motivated to be savers by a set	the return on their
of saving motivations. Banking facilities provided by	saving if they decide to
the bank has a positive influence	save money. Attached high importance
on their banking decision. Concerned about aspects of bank	to financial advice and
location, bank building design and	consultation. Attached high importance
bank reputation. Attached high importance to the	to information about the
length of waiting time they have	bank services in terms of
to spend every time they visit	advertisement. In general, this group
the bank. In general, this group enjoys	lacks knowledge in
a good knowledge in banking and	banking and it has a
it has positive influence on	negative influence on
their banking behaviour.	their banking behaviour.

Sample Classification Results:

Table 9.5 shows the sample classification results of this study's Discriminant Analysis. It shows that the discriminant function correctly classified 408 cases out of 621. In other words, 65.70 per cent of grouped cases were correctly classified by the analysis. The value of chi-square statistic for the classification results of the discriminant function is equal to 85.23, which has a corresponding significant level of (.0000). Also, table 9.5 shows

that 66.8 per cent of respondents who are "banked with savings accounts" were correctly classified as "banked with savings accounts", and 64.9 per cent of "non-banked" were correctly classified as "non-banked". Thus, the results shown in table 9.5 concluded that: (a) there is a substantial discriminating power between the two groups; and (b) the equally likely chance of classification without knowing the population distribution of "banked with savings accounts" and "non-banked" should be equal to 50 per cent. Therefore, a classification result of 65.70 per cent confirmed that the model improved over equally likely chance

Figure 9.1 each and all-groups stacked histogram shows that there are only two groups, "banked with savings accounts" and "nonbanked", therefore, only one discriminating function was extracted. Thus, the horizontal axis is the discriminant function derived by the analysis. The groups into which the cases were classified, given their discriminate function scores, were indicated by the labelling of the horizontal axis. Thus, the canonical discriminant function evaluated at group mean (group centeroid) for the group of "banked with savings accounts" is equal to (-.45714), and for the group of "non-banked" is equal to (.33652). This findings indicate that the group means for both groups are not equal. In other words, there are significant dissimilarities between the two groups in terms of their attitudes towards the service of savings accounts provided by commercial banks in Iraq.

Actual Group	No. of Cases	"banked"	"non-banked"
Group Yes "banked"	265	. 177	88
		66.8%	33.2%
Group No "non-banked"	356	125	231
		35.1%	64.9%

Table 9.5: Groups Classification Results.

Testing the Research Hypothesis:

Significance level = .0000

The research hypothesis to be tested, as stated in its null form, is the following:

• • • • •

Hypothesis HO: There are no significant dissimilarities between respondents who are "banked with savings accounts" and "non-banked" in terms of their attitudes towards the service of savings accounts provided by commercial banks in Iraq.

The output of Step-Wise Two-Group Discriminant Analysis indicates that the canonical discriminant functions evaluated at group means (group centeroids) for the two groups are not equal ("banked with savings accounts" = -.45714 and "non-banked" = .33652). These results indicated that the two groups of "banked with savings accounts" and "non-banked" differ significantly.



In addition to above, the output of the analysis provided the following information, which refers to the fact that there are significant differences between the two groups:

Firstly, the significant level of the observed Wilks' lambda can be based on chi-square transformation of the statistic. Therefore, the value of Wilks' lambda in this analysis was equal to .84 and its associated Chi-square value was equal to 85.23, which has an associated significance level of .0000. Therefore, the null hypothesis (H0), that the group mean values are equal for both groups is rejected.

Secondly, the classification results presented in table 9.5, together with the all-group stacked histogram shown in figure 9.1, indicate that there are significant differences between the two groups.

Thirdly, the results of the analysis proved that 18 variables were found to be of significant importance in discriminating between the two groups of "banked with savings accounts" and "non-banked". This conclusion is based on the contribution made by these variables in discriminating between the two groups, according to their group mean values and their standardized canonical discriminant function coefficients, as shown in table 9.3.

Fourthly, based on the values of Wilks' lambda, which were converted into variance ratio (F-values) for the purpose of testing the significance of the discriminant function. The output of the Discriminate Analysis confirmed that the values of the Wilks' lambda

for the 18 discriminating variables were all significant at .000 significance level.

Therefore, based on the above findings the null hypothesis (HO) is rejected in favour of the alternative hypothesis. In other words, this section analysis results confirmed that there are significant dissimilarities between respondents who are "banked with savings accounts" and "non-banked" in terms of their attitudes towards the variables related to the service of savings accounts provided by commercial banks in Iraq.

9.2.2 The Findings of T-Test:

In the previous section (section 9.2.1) respondents were compared according to their saving decisions, i.e., "banked with savings accounts" vs. "non-banked". The attempt will be made to explore the dissimilarities between respondents in terms of the rates they attached to a number of motivations and reasons for saving together with some service and bank attributes. Therefore, the t-test will be used to investigate the dissimilarities between respondents if divided according to their saving habits, age, family size, income and educational level.

9.2.2.1 Respondents' Saving Habits

The results of the t-test for comparing respondents according to their saving habits, i.e., good vs. poor savers, as presented in table 9.6, indicated that sixteen service attributes and motivations (reasons) for saving received different rates of importance by the both groups of savers. In other words, the group mean values for each of the sixteen characteristics are not equal. The t-value for the sixteen characteristics, as shown in table 9.6, indicated that

they are all significant at less than .01 significance level. Also, the t-values together with the group mean values (mean ratings) indicated that good savers attached more importance to all the sixteen variables than poor savers. Such a finding is expected since by dividing respondents according to their savings habits the group of good savers consisted of all 265 respondents who are "banked with savings accounts" plus a segment of respondents who are "nonbanked" but view themselves as good savers. Perhaps because they are using other methods for savings, as will be explained later in this section, than holding savings accounts. On the other hand, the group of poor savers represented the remaining respondents who indicated that they are poor savers in both terms, neither holding savings accounts nor using other methods for saving.

As indicated in table 9.6, the variable (v3) "family is protected in emergencies" received the highest rate of importance by both groups. This finding confirmed that this motivation is considered by Iraqi consumers as a very important motivation for saving. Also, it confirmed the fact that most Iraqi savers seek keeping surplus money they can spare for future use, especially at the times of emergencies. This method of keeping money for future use is more common than any type of insurance in Iraq. Also, it gives consumers the flexibility to use the money they saved to face different types of emergencies rather than investing it in one type of insurance and facing emergency in other aspects of life in particular when consumers have little or some money to spare.

Variables	Group me good savers	ean rating poor savers _.	t value	2-tail prob.
vl meet living expenses	1,73	2.09	-4.35	.000
v3 family is protected in	1.34	1.61	-3.97	.000
v4 build-up a retirement	1.77	2.10	-3.86	.000
v6 buying a property	1.89	2.35	~4.67	.000
vll feel own part of the	2.87	3.39	-4.61	.000
v14 use other services	2.89	3.23	-3.26	.001
v15 higher interest	2.41	2.78	-3.52	.000
vl6 method to keep surplus money	1.71	1.96	-3.14	.002
v30 feel important as	2.27	2.65	-3.65	.000
v42 availability of	1.96	2.42	-4.37	.000
v45 design of building	2.04	2.44	-4.07	.000
v48 bank is well-known	2.05	2.36	-3.12	,002
v52 AD on TV	2.06	2.40	-3.44	.001
v56 personal contact by bank	2.44	2.86	-4.35	.000
v57 full-range of services	1.87	2.15	-3.49	.001
v60 promotions	1.74	2.01	-3.03	.003

Table 9.6: The results of t-test to compare (good vs. poor savers)

Note: The ratings are based on a five-point scale, where l= very important and 5= not important at all.

The group mean values for other variables, as presented in table 9.6, indicated that they received generally high rates of importance by the group of good savers. While the group of poor savers generally attached less or some importance to the same group

of variables. The prime reason for that is the fact that good savers are actual savers not because they indicated that they are good savers, but because they are using savings accounts or other methods for saving. However, most good savers are holders of savings accounts therefore they have more positive attitudes towards the idea of saving in general and holding savings accounts at commercial banks in particular.

The results of the t-test shown in table 9.6 suggested that good savers and poor savers differ along sixteen characteristics. Thus, the group of good savers in comparison with the group of poor savers tended to give significantly higher importance to all the motivations and reasons for saving, such as "meeting living expenses", "family is protected in emergencies", "building-up a retirement fund" and "buying a property". Also, they tended to give significantly more importance to other savings accounts attributes, such as "higher interest rate" and "promotion". In addition, they attached more importance to other bank attributes, such as "fullrange of services", "availability of parking facilities", and "bank is well-known".

In summary, the group of good savers in comparison with poor savers are more concerned about the benefits they are gaining from being good savers. Also, they are more concerned about the quality of the services they received from the bank and about some of the bank attributes which may influence their saving decision and selecting a particular bank to deal with.

Finally, the classification results of the previous section (section 9.2.1) indicated that out of the sample of 621 respondents, there were 265 respondents "banked with savings accounts" and 356 "non-banked". By using the same sample with using variable (v61) "I

an a good saver" as the classification variable, i.e., good vs. poor savers, the classification results have changed. Thus, the results of the t-test indicated that out of the 621 respondents there were 403 respondents who indicated that they are good savers. That means, the actual savers, i.e., people who save surplus money for future use, are not only the 265 respondents who have savings accounts but another 138 respondents who save by other means. In other words, 403 respondents indicated that they are good savers while only 265 of them save through savings accounts. This indicates that the remaining 138 good savers are using other methods to hold their savings. If this is the case with a sample of 403 respondents, i.e., good savers, where only 68 per cent save through savings accounts while the remaining 32 per cent save through other methods, what about the entire population of Iraq? In other words, one can conclude that about 32 per cent of savers in Iraq are using other methods to hold their surplus money rather than depositing it at a bank. Such a conclusion is in fact the real situation in Iraq for a number of reasons: Firstly, there is a popular distrust of banks in Iraq and people are reluctant to deposit their money with them. Secondly, some people are illiterate and find it hard to deal with banks. They believe in cash in their hands and tend to keep it in secure places in their homes in the form of money, gold or jewelry. Thirdly, the banks in Iraq are largely to blame for this situation since they provide poor services in terms of quality, quantity and efficiency. For some consumers dealing with banks is a non-preferable activity. Iraqi banks, as described in chapter two, suffer from many problems which influence their performance and ability to serve consumers. Therefore, Iraqi consumers became reluctant to deal with banks. Fourthly, the interest rate paid by

banks is very low, in fact lower than the rate of inflation. Therefore, people are not willing to take the risk of depositing their money in a bank for a very low return. However, it should be mentioned here, following the above conclusions, that some savers perhaps defined themselves as good savers whilst the amount of money they save or they hold in the form of savings is not significant. They prefer to maintain their money under close control in a secure place of their choice rather than in a bank. They do not consider it worthwhile to deposit money in a bank given the difficulties mentioned above.

9.2.2.2 Respondents' Age Group

The t-test also used for comparing respondents according to their age groups: Group (A) respondents aged 44 years and under and Group (B) respondents aged 45 years and over. The results of the ttest, as shown in table 9.7, showed that younger and older respondents differ along four characteristics. Thus, the t-test confirmed that there were statistically significant differences between younger and older respondents in Iraq on certain characteristics. The group mean ratings together with the t-values confirmed that younger respondents attached significantly higher importance to three attributes. These are: "bank personnel are knowledgeable", "length of waiting time" and "location of branches".

Variables	Mean r Group A	ating Group B	t value	2-tail prob.
vll feel own part of the bank	3.15	2.86	2.48	.01
v21 bank personnel are knowledgeable	1.97	2.16	-2.26	.02
v32 length of waiting time	2.37	2.64	-2.44	.01
v4l location of branches	1.96	2.15	-2.22	.02

Table 9.7: The results of t-test to compare respondents according to their age group

Notes:

1. The ratings are based on a five-point scale, where 1 = very important and 5 = not important at all.

2. Group A = respondents aged 44 years and under.

3. Group B = respondents aged 45 years and over.

These findings indicated that younger respondents are more concerned about aspects of convenience a bank can provide to them in terms of location and time. Also, they seemed to be more pleased in dealing with bank personnel who reflect good knowledge about the services on offer by the bank. Therefore, it can be concluded that younger respondents indicated that they view convenience in terms of time, location and people to deal with as the prime important requirements to feel satisfied with the bank.

Older respondents attached higher importance to feeling owning part of the bank than younger respondents. This difference can be interpreted as one of the characteristics of older people, in particular the oldest segment of this group. Thus, they are mostly retired people and they deposited their lifetime savings at the bank they are entitled to feel they are owning part of the bank bearing in mind that most of a bank's assets are its depositors money.

9.2.2.3 Respondents' Family Size

When respondents were compared according to the size of their families the results of the t-test, as presented in table 9.8, indicated that they differ along five characteristics. Thus, the results indicated that respondents of group (B), i.e., family of six and over, attached significantly higher importance to the five characteristics shown in table 9.8 than respondents of group (A), i.e., family of five and under.

Table 9.8: The results of t-test to compare respondents according to their family size

Variables	Mean Group A	rating Group B	t value	2-tail prob.
vl0 easily withdraw savings	2.25	2.04	2.10	.03
vl8 advice from family members	2.43	2.16	2.49	.01
v35 accuracy of transactions	1.71	1.47	3.14	.00
v36 confidentiality of transactions	1.87	1.58	3.38	.00
v37 speed of processing transaction	1.97	1.76	2.52	.01

Notes:

The ratings are based on a five-point scale, where 1 = very important and 5 = not important at all.
 Group A = respondents their family size = 5 and under.
 Group B = respondents their family size = 6 and over.

, droup b - respondents their ramity size - o and over.

The mean ratings together with the t-values indicated that respondents of group (B) are more concerned about quality of transaction variables in particular accuracy and confidentiality of transactions. This finding indicated that respondents of larger families are very keen about their money, may be more so than the average person, but perhaps because this segment of people feel that they have heavier social responsibility towards the welfare of their large families. Also, respondents of larger families attached more importance to advice from a family member. The influence of the family on consumer behaviour in societies like the Iraqi society is very strong. However, it is even stronger in large families.

9.2.2.4 Respondents' Income Group

The test was used in this section in order to investigate the dissimilarities between respondents of two income groups in Iraq. Group (A) consisted of respondents their income group is 2999 I.D. and under, and Group (B) consisted of respondents their income group is 3000 I.D. and over. The results of the t-test, as presented in table 9.9, indicated that the two group differ along four characteristics.

Table 9.9: The results of t-test to compare respondents according to their income group

Variables	Mean ra Group A	ating Group B	t value	2-tail prob.
vl meet living expenses	1.77	1.96	-2.37	.01
v6 buying a property	1.95	2.19	-2.46	.01
v15 higher interest	2.39	2.74	-3.36	.00
v37 speed of processing transactions	1.87	1.72	2.03	.04

Notes:

- 1. The ratings are based on a five-point scale, where 1 = very important and 5 = not important at all.
- 2. Group A = respondents their income group = 2999 I.D. and under.
- 3. Group B = respondents their income group = 3000 I.D. and over.

Table 9.9 shows that respondents of lower income (group A) attached significantly higher importance to three variables. These

are "meeting living expenses", "Buying a property" and "higher interest". These findings indicated that respondents of lower income motivated to save and open a savings accounts by the desire to improve their standard of living and keeping some present financial assets to face future demands such as buying a property. Respondents of higher income attached less importance to these motivations for savings than their counter-group because their income is higher and they manage to deal with their future demands more easily than people with low income.

Higher interest on savings received a lower rate of importance by both groups among all other variables included in table 9.9. The obvious reason for that is the fact that the rate of interest paid by banks in Iraq is low and no match the rate of inflation. However, the group of lower income attached somewhat higher importance to higher interest on savings than the group of higher income.

The group of higher income attached significantly higher importance to one of the service attributes, i.e., speed of processing transactions. This finding indicated that this group are more frequent users for the services provided by banks than the group of lower income.

9.2.2.5 Respondents' Educational level

In this section an attempt was made to compare respondents according to their educational level. Thus, the aim was to investigate the influence of consumers' educational level on their preferences. The results of the t-test, as presented in table 9.10, suggested that respondents of lower and higher educational level differ along five characteristics.

Variables	Mean a Group A	rating Group B	t value	2-tail prob.
vl meet living expenses	1.74	2,05	-3.92	.00
vl2 start a new business	1.90	2.25	-3.90	.00
v15 higher interest	2.43	2.70	-2.51	.01
vl7 advice from friend	2.60	2.85	-2.31	.02
v18 advice from family members	2.10	2.42	-3.49	.00

Table 9.10: The results of t-test to compare respondents according to their educational level

Notes:

1. The ratings are based on a five-point scale, where 1 = very important and 5 = not important at all.

2. Group A = respondents their educational level = Secondary School and under.

3. Group B = respondents their educational level = University and over.

The t-values together with the mean ratings indicated that respondents of lower educational level, i.e., graduates of secondary school and under. attached higher importance to the five characteristics shown in table 9.10 than respondents of higher educational level, i.e., graduates of university and over. Thus, respondents of lower educational level attached more importance to variables, such as "starting a new business" which underlined the fact that most people with some or no formal education In Iraq tend to work in the private sector, in particular, establishing their own businesses. Also, since banks are reluctant to provide financial assistance to people starting a new business, especially small businesses, people learnt to depend on their savings to finance their new businesses.
In addition, people with low or no formal education in Iraq usually that segment of people with low income tend to give higher importance to aspects related to achieving some gains from saving such as "higher interest".

Finally, respondents of lower or no formal education indicated that they are more dependent on advice provided by friends and family members in making their decisions than respondents of higher educational level. This finding explains the fact that lack of education made people more dependent on others in making their decisions especially when it comes to matters of some complication such as banking.

9.3 Comparison with other Studies of Consumers' Saving Behaviour :

The results of eight prior studies related to the aspects of consumers' banking behaviour in general and consumers' saving behaviour in particular, which were discussed earlier in Chapter Three, may be compared to the present investigation's results.

Claycamp (1965) developed a model to compare between three groups of savers: (a) savers holding accounts at Commercial Banks; (b) savers holding accounts at Savings and Loan Associations; and (c) savers holding accounts in both of them. In other words, Claycamp, in his study, investigated the group of savers only and used only some motivations, socio-economic and psychological characteristics to investigate the differences between the three groups of savers. Thus, the study was limited to one segment of consumers, i.e., savers, and only a few related variables were used in the analysis, while consumers attitudes towards other related aspects to consumers saving decisions were neglected.

However, the present investigation confirmed the finding of Claycamp's study that psychological variables did prove to be as important in discriminating between consumers. In addition, the present study was an attempt to provide further understanding for consumers' saving behaviour by investigating the differences between consumers according to their: (a) saving decisions; (b) saving habits; (c) age; (d) income; (e) family size; and (f) educational level. Furthermore, the present study investigated the attitudes of savers, in particular, towards the variables related to their saving behaviour and the role of their socio-economic and demographic characteristics in influencing their saving decisions.

Claycamp concluded that only psychological variables were important as discriminating variables. The present study confirmed that and concluded in addition to it that consumers' saving behaviour can be affected by their attitudes, motivations, socioeconomic, demographic and psychographic characteristics.

Allaire (1972) developed a model to determine the choice of financial organisations by consumers in which to open savings accounts. The findings of the present study confirmed the results of Allaire's study that convenience of location was given great importance by consumers as the prime reason for selecting a financial organisation to deal with. In addition to that, the present study confirmed that other aspects of bank-customer relationship, competition, savings accounts attributes and other facilities provided by the chosen financial institution were of importance in influencing savers decisions.

In his study Allaire concluded that the term convenience, which consists of convenience of location, logical place and convenient hours of operation, is perceived by consumers as of

significant importance. Thus, consumers' perception of a bank depends largely on the convenience it provides to customers in terms of time, place, services and other facilities.

Mason and Mayer (1974) attempted to determine the reasons for maintaining savings accounts with Banks, Savings and Loan Associations and Credit Unions. The results of the present study confirmed the findings of Mason and Mayer, that is, that convenience of location was evaluated by consumers as of significant importance in influencing their decision to have a savings account with a particular financial institution. In addition, the reputation of the bank and its image were found, as is the case in Mason and Mayer's study, of significant importance in influencing consumers' saving decisions and was even more important than the rate of interest.

Furthermore, the present study and Mason and Mayer's study confirmed that consumers want to achieve more than financial gains by choosing a particular bank, they want convenience, good treatment and respect.

Reynolds and Wells (1978) in their analysis of consumers life style concluded that there are differences in the life style of three groups of consumers: non-savers, infrequent savers and frequent savers. However, the present study confirmed that there are differences particularly between savers, i.e., banked with savings accounts, and non-savers, i.e., non-banked, not only in terms of their life style, but also in terms of their attitudes, motivations, socio-economic, demographic and psychographic characteristics.

In addition, the Reynolds and Wells study was limited to male respondents only, while the present study investigated both male and female respondents' saving behaviour. Furthermore, the model developed by Reynolds and Wells compared between respondents

according to their saving decisions only, while in this investigation respondents were compared according to other criteria, such as age, income, education and family size, as well as to their saving decisions and habits. In other words, the findings of this study provided more insight into consumers' saving behaviour in terms of the variables which may influence consumers behaviour in addition to consumers life style.

As is the case with other studies of consumers' saving behaviour which were carried out in the U.S.A, Arora, Cavusgil and Nevin (1985) examined a number of factors which are important in consumers' decisions to choose between Banks and Savings and Loan Associations.

In their study the authors concluded that Banks and Savings and Loan Associations share many similar characteristics in the eyes of consumers. However, their study did not investigate the differences between the customers of Banks and Savings and Loan Institutions. Also, it did not investigate the effect of consumers' socio-economic and demographic characteristics in their choice and behaviour.

However, the present study confirmed the findings of Arora, Cavusgil and Nevin's study that a number of common variables were of significant importance in influencing consumers decisions to open a savings account with a particular financial institution, such as: convenience of location, full-range of services, friendliness of bank personnel etc.

With regard to a model of investigating a particular consumer segment's attitudes Goudge and Green (1986) investigated children and their parents' attitudes towards savings accounts in the U.K. In other words, Goudge and Green attempted to investigate the role of

the family in influencing their children's saving behaviour. In fact, they concluded, as is the case in the present study, that the family, and the parents in particular, have a great influence on their children's saving behaviour and decisions.

Langrehr (1986), as most researchers of consumers' saving behaviour in the U.S.A, attempted to investigate the competition between Banks and Savings and Loan Institutions in the U.S.A. Langrehr concluded that there were significant differences in consumers' evaluation of Banks and Savings and Loan Institutions in terms of the products (types of services) they provide. Thus, Langrehr findings concluded that Banks were evaluated as better than Savings and Loan Institutions by consumers.

However, the present study confirmed Langrehr's finding that Banks were in the top list of choices in the eyes of consumers mainly because banks provide a full-range of services which are not provided by other types of financial institutions.

The last consumer saving behaviour study which can be impared to the present investigation was conducted by Goodfelow 987). Goodfelow attempted to gain insights into the basic tivations, attitudes and beliefs of savers. Goodfelow also tempted to investigate consumers' perceptions and attitudes wards savings and investments. The findings of Goodfelow's study evealed that at a superficial level the terms savings and investment were perceived to define different areas, but in-depth analysis confirmed that there was no meaningful differentiation between the two terms. Thus, both terms were perceived as terms for money which is not spent, but is tied up for future use.

However, the present study findings share with Goodfelow's findings that stage in life cycle, which contains a combination of

age, family status, education and perhaps occupation, can be identified as a significant factor in influencing attitudes and behaviour of consumers towards savings.

But. Goodfelow, in his investigation. reported the differences between saving and investing and the effect of the consumers' life cycle stage in formulating their financial behaviour only. In other words, Goodfelow's study was limited to one segment of consumers, i.e., savers, therefore, a sample of 25 savers only was used in his study. In fact, such a small sample size can not be statistically representative for a large population such as savers. In addition, Goodfelow's study did not explain the interrelationship between factors affecting consumers' saving behaviour. Also, it did not extend the investigation to cover the differences between consumers' saving behaviour according to their, for example, saving decision, saving habits, income, age, educational level, occupation and family size.

Finally and generally speaking, the present investigation was an attempt to extend and develop the models of investigating consumers' banking behaviour in general and consumers' saving behaviour in particular. That was done through recruiting all the possible socio-economic, demographic, psychographic, motivational and attitudinal variables which may influence consumers' saving behaviour.

In addition, attempts were made in the present investigation in order to provide comparison, i.e., differences, between consumers' behaviour according to six major criteria, these were: (a) Consumers' saving decisions "banked with savings accounts" vs. "non-banked"; (b) Consumers' saving habits (good vs. poor savers); (c) Consumers' age groups; (d) Consumers' family size groups; (e)

Consumers' family income group; and (f) Consumers' educational level.

Also, the relationships between consumers' saving decisions, i.e., to be "banked with savings accounts" or "non-banked", and their socio-economic and demographic characteristics were investigated. In other words, the present investigation attempted to provide an overlook to the most important variables which may influence consumers' saving decisions and behaviour.

9.4 Conclusions:

In this chapter the attempts were made to discriminate between consumers' attitudes towards having a savings account at a commercial bank and their saving behaviour according to six criteria (dimensions), as follows:

Firstly, Discriminant Analysis and T-Test were used in order to achieve the stated objective of this investigation. Thus, the findings of the DA and T-Test confirmed that there were dissimilarities in consumers ratings for the importance of the variables related to the idea of saving and the service of savings accounts provided by commercial banks in Iraq, if they are compared according to their saving decisions, savings habits, age, income, education and family size.

the methods and findings of the Secondly. present investigation confirmed most of the findings of previous investigations of consumers' saving behaviour and attitudes towards the related variables. The present study in comparison saving and with previous studies was more comprehensive not in terms of sample size and statistical techniques of analysis used, but in terms of its attempts to investigate the problem of consumers' saving

behaviour through all its possible dimensions. In other words, all the relevant consumers' socio-economic and demographic characteristics to consumers' saving behaviour were used in order to investigate the differences between different groups of consumers.

.

.

.

CHAPTER TEN

MULTIPLE CLASSIFICATION ANALYSIS FINDINGS

.

10.1 INTRODUCTION.

10.2 MULTIPLE CLASSIFICATION ANALYSIS FINDINGS FOR: 10.2.1 CONSUMERS' SOCIO-ECONOMIC CHARACTERISTICS. 10.2.2 CONSUMERS' DEMOGRAPHIC CHARACTERISTICS.

10.3 TESTING RESEARCH HYPOTHESES.

10.4 CONCLUSIONS.

10.1 Introduction:

The main objective of this chapter is to examine the relationship between consumers' saving behaviour (decisions), i.e., their decision to hold or not to hold a savings account, and their profile characteristics.

The term "profile characteristics" is used to mean selective socio-economic characteristics, such as income level, educational level, and occupation; and demographic characteristics, such as sex of respondent, marital status, age, and family size. Such profile characteristics were used in this investigation, because, it was thought to be the most related characteristics in influencing consumers' saving behaviour (decisions) in Iraq.

However, in order to achieve this objective a computer program titled Multiple Classification Analysis (MCA) was used for the purpose of data analysis. The MCA statistical technique is one of the dependent methods of multivariate data analysis. Thus, in this investigation the purpose was to examine the relationship between the single dependent variable (pl), i.e., hold / or not hold a savings account, and a number of independent categorical variables (predictors). The variable (Pl) represents consumers' saving decisions to hold or not to hold a savings account. In other words, the main objectives for using the MCA statistical technique in this study are the following:

(a) To determine the ability of all predictors (independent variables) in explaining the proportion of variance in the dependent variable (bank customers' saving decisions);
(b) To measure the ability of a predictor variable in explaining variation in the dependent variable (bank

customers' saving decisions) after adjusting for the effects of all other predictor variables; and

(c) To determine whether a predictor variable has a significant portion (effect) in explaining the variation of the dependent variable (bank customers' saving decisions to hold or not hold a savings account at a commercial bank).

The MCA computer program has no constraint on the type of data (i.e., in terms of data measurement). Thus, the predictors are treated as sets of categories whatever their measurement type, i.e., nominal, ordinal, interval, or ratio. The MCA computer program output consists of the grand mean of the dependent variable, and a table of category means for each predictor variable expressed as a deviations from the grand mean. Also, the MCA output contains the following statistics:

1. The Eta statistic: This is the correlation ratio, and represents the ability of the predictor variable to explain the variation in the dependent variable. Eta statistic is similar to Beta statistic but it is based on the unadjusted mean rather than on the adjusted mean. However, the Eta statistic is very useful for the purpose of calculating the value of F-test for each predictor. Thus, the Eta statistic is useful in calculating the sum of squares, which is based on the unadjusted deviation for each predictor. This sum of squares is essential in calculating the value of F-test. In other words, it is essential for the purpose of testing the research ypotheses for all predictors taken together, and each predictor

2. The Beta statistic: This statistic indicates the ability of the predictor, i.e., the independent variable, in explaining the

variation in the dependent variable, after adjusting for the effect of all other predictors.

3. The Multiple Coefficient of Determination $\binom{2}{R}$: This coefficient indicates the ability of all predictors taken together in explaining the variation in the dependent variable.

4. The F-ratio: This ratio allows to test the significance of a predictor variable in explaining the proportion of variance in the dependent variable. However, the F-ratio is not the F-test, thus, the MCA output does not calculate the F-test. As mentioned in item one above, the required statistics for the purpose of calculating the F-test are part of the MCA output, such as: the sum of squares for each predictor, and the total sum of squares for all predictors taken together.

Based on the above assumptions, and in order to achieve the stated objective from applying the MCA technique, the categorical data gathered from 621 respondents were analysed. In the following section an interpretation for MCA results is provided:

10.2 Multiple Classification Analysis Findings: Consumers' Profile Characteristics:

As mentioned earlier, the main objective of this chapter was to examine the effects of consumers' profile characteristics in influencing consumers' saving behaviour (or their decisions to open or not to open savings accounts at a commercial bank). Therefore, the MCA computer programme output, as one of the dependent method, provides the required statistical results to achieve this objective.

However, table 10.1 describes the main results of the MCA computer program, which consist of the grand mean for the dependent variable, the class mean for each category in each predictor

variable, the Beta statistic for each predictor, and the F-ratio for each predictor.

Table 10.1 shows that the coefficient of determination (R) is equal to .30 for all predictor variables taken together. This statistic suggests that the seven consumers' profile characteristics contributed in explaining 30 per cent of the variation in consumers' saving behaviour, i.e., the dependent variable (p1) (hold / not hold a savings account). While, 70 per cent of the variation in variable (pl) remained unaccounted for by the seven consumers' profile characteristics. By referring to the findings of chapters eight and nine of this study, which investigated consumers' attitudes towards the service of savings accounts, the remaining unexplained 70 per cent of the variation in variable (pl) can be explained by the 65 attitudinal variables used to discriminate between respondents according to their saving decisions. Thus, the variation in variable (p1), i.e., consumers saving decisions, can be influenced by two sets of variables. These are consumers attitudes towards the service of savings accounts and their profile characteristics. In chapters eight and nine the influence of consumers' attitudes on their saving decisions was investigated and reported. In this chapter, and in order to complete the investigation, the influence of consumers' socio-economic and demographic characteristics is to be analysed.

By comparing the contribution of this research consumers' profile characteristics in explaining the variation in the dependent variable with other attempts using the same statistical technique and type of categorical data it can be concluded that 30 per cent is not bad (see for example, Anderson and Nevin, 1975).

One of the aims of this analysis was to determine the ability of all predictors (independent variables) taken together in

explaining the proportion of variance in the dependent variable (consumers' saving decisions). This can be achieved, in addition to maximising the value of $(R)^2$, by examining the overall value of F-ratio and its significance for the independent variables. Thus, because the F-ratio is an indicator for the significance of the independent variables in explaining the variation in the dependent variable (consumers' saving decisions).

Table 10.1: The (MCA) Output for the Relationship Between Consumers' Saving Behaviour and Consumers' Profile Characteristics.

No	Variable Categories	N	2	Class Mean	Beta	F-Ratio
xl	Sex:	621	100		.09	2.982 Ъ
	Male	504	81	.59		
	Female	117	19	.51		
x2	Marital Status:	621	100		.37	13.351 a
	Married	535	86	.55		
	Single	86	14	.71		
x3	Age Group:	621	100		.24	11.388 a
	24 years and under	72	12	.81		
	25-44 years	335	54	.55		
	45-64 years	165	26	.46	1	
	65 years and over	; 49	8	.76	1	2 -
x4	Family Size Group:	621	100		.36	4.579 a
	Single	· 87	14	.71	-	-
	Couple without child	51	8	.53	-	2
	Comple with three	214	35	.49		2 8 7
	children and under	:	1	:	- 	
*	Couple with four	269	43	.61		9
	children and over			- -	51	-
x5	Family Income Group:	621	100	# - E1 - B111	.13	29.315 a
1	Under 2000 I.D.	203	33	.71	# £	
5 • 2	2000-2999 I.D.	, ≣ 15€	5 = 25	.74	9 E	÷
•	3000-3999 I.D.	: 138	3 - 22	.41	-	-
	4000 I.D. and over	124	20	.31		87 1 12

Table 10.1 (Continued)

No	Variable Categories	N	z	Class Mean	Beta	F-Ratio			
x6	Household Educational Level:	621	100		.11	5.578 a			
	Less than secondary	242	39	.70					
	school.								
	Secondary school	140	23	.53					
	graduate.								
, ; ;	University graduate.	198	32	.49					
	Higher education	41	6	.38					
	graduate (Msc, Ph.D).								
x7	Household Occupation:	621	100		.41	10.467 a			
	General manager and	25	4	. 20					
	higher positions.				1				
	Manager	49	8	.29					
	Teacher	55	9	.52					
	Clerk	150	24	.72					
	Farmer	49	8	.82					
	Labourer	89	14	.89	*				
	Pensioner	63	10	.63					
	Owner of private	94	15	.17					
	business								
	Housewife	23	4	.52					
	Student	24	4	.54		1			
Tota	1 population = 621 (265)	5 "banl	ked w	ith savings a	account	ts" and			
356	"non-banked").								
Gran	d = .57	2							
Coefficient of determination $(R) = .30$									
Explained sum of squares = 46.473 has F-ratio = 11.440									
	which has sign	nifica	nce 1	evel = .000					
Residual sum of squares = 105.444									
Tota	Total sum of squares = 151.916								
a = significant at less than .01 significance level									
b = not significant									

As shown in table 10.1, the overall value of F-ratio for all independent variables taken together is equal to (11.440), which is significant at (.000) significance level. This finding suggests that consumers' socio-economic and demographic characteristics explain the variation in the dependent variable (p1), i.e., consumers' saving decisions. In other words, the seven socio-economic and demographic variables, taken together, are explaining why some consumers decide to be savers, i.e., banked with savings accounts, and others non-savers, i.e., non-banked.

After achieving that, the second aim of this analysis was to measure the ability of each predictor variable in explaining the variation in the dependent variable, i.e., consumers' saving decisions, after adjusting for the effects of all other predictor variables. In other words, the aim was to identify the ability of each predictor in explaining, why consumers take their decisions either to be savers (banked with savings accounts) or non-savers (non-banked).

The third aim was to determine whether a predictor variable has a significant portion (effect) in explaining the variation in the dependent variable, i.e., consumers' saving decisions. In other words, the aim was to measure the significance of each predictor in explaining why some consumers decided to be savers and others not.

In order to achieve the last two above mentioned aims, the MCA output provided two types of statistics (Beta and F-ratio), which can best describe the ability of each predictor in doing this. These statistics are shown in table 10.1 and will be discussed for each predictor separately in the following sections of this chapter.

addition to that, table 10.2 provides descriptive In statistics for each predictor variable, which help to provide more insights in demonstrating the importance of each predictor in explaining consumers' saving decisions. Table 10.2 consists of a number of descriptive statistics. The first column is the variable code. The second column is the variable name and the categories it contains. The third column (N) represents the number of respondents represented in each category of each predictor. Thus, the total sample was 621 respondents, and each predictor consists of a number of categories. For example, the predictor sex of respondents consists of two categories: Male and Female. Therefore, the total sample of 621 respondents consists of 504 males, and 117 females, and so on for other predictors while the remaining four columns represent the number of savers and non-savers and their percentage for each predictor in terms of the categories it consists of.

No	Variable Categories	N	Banked	z	Non-banked	z
xl	Sex:					
	Male	504	208	41	296	59
	Female	117	57	49	60	51
x2	Marital Status:					
	Married	5 35	240	45	295	55
	Single	86	25	29	61	71
x 3	Age Group:					
	24 years and under	72	14	19	58	81
	25 – 44 years	335	150	45	185	55
	45 – 64 years	165	89	54	76	46
	65 years and over	49	12	24	37	76
x4	Family Size Group:					
2 1 2 2	Single	87	25	29	62	71
	Couple without child	51	24	47	27	53
	Couple with three	214	110	51	104	49
	children and under					
-	Couple with four	269	106	39	163	61
:	and over			1		
x 5	Family Income Group:					
	Under 2000 I.D.	203	58	29	145	71
	2000 - 2999 I.D.	156	40	26	116	74
	3000 - 3999 I.D.	138	81	59	57	41
	4000 I.D. and over	124	86	69	38	31

Table 10.2: Rate of Occurrence of the Consumers' Profile Characteristics for All and Each Group of Respondents.

Table 10.2 (Continued)

No	Variable Categories	N	Banked	z	Non-banked	2
x6	Household Educational					
	Level:					
	Less than secondary	242	72.	30	170	70
	School.					
	Secondary school	140	66	47	74	53
	graduate.					
	University graduate.	198	101	51	97	49
	Higher education	41	26	63	15	37
	(Msc., Ph.D.).					
x7	Household Occupation:					
	General manager and	25	20	80	5	20
	higher positions.					
	Manager	49	35	71	14	29
	Teacher	55	26	48	29	52
	Clerk	150	42	28	108	72
	Farmer	49	9	18	40	82
	Labourer	89	10	11	79	89
	Pensioner	63	23	37	40	63
	Owner of private	94	78	83	16	17
	business.					
	Housewife	23	11	48	12	52
	Student	24	11	46	13	54
[FOTAL	621	265	43	356	57

10.2.1 Consumers' Demographic Characteristics:

Four demographic predictors were used in this analysis, and they are: Sex of respondent, Marital Status of respondent, Age Group of respondent, and Family Size Group of respondent.

Predictor x1: Sex of respondent:

The findings of the MCA shown in table 10.1 indicated that sex of respondents has a Beta statistic of .09 and a F-ratio of 2.982. By comparing the Beta value and the F-ratio of this variable with the values of other independent variables, as shown in table 10.1, it can be concluded that sex of respondents found to be the variable of the lowest ability to explain the variation in the dependent variable. In other words, sex of respondents does not explain why Iraqi consumers decide to be "banked with savings accounts" and others not.

In addition, table 10.2 and figure 10.1 show that out of the 621 respondents, 504 are males, 208 or 41% of them are "banked with savings accounts", and 269 or 59% are "non-banked". While, out of the 621 respondents, 117 are females, 57 or 49% of them are "banked with savings accounts", and 60 or 51% are "non-banked". This numerical comparison indicated that males are marginally likely to be "non-banked" rather than "banked with savings accounts". With regard to females respondents the results indicate that females can be banked or non-banked since the difference between banked females and those non-banked is very small (only 2%). Therefore, it can be concluded that the sex of respondents (in this case of study) is of no significant importance in explaining the variation between consumers' decisions to hold or not to hold a savings account.



FIGURE 10.1: SEX OF RESPONDENTS

SEX

SAVERS

NON-SAVERS

Predictor x2: Marital Status of respondents:

Table 10.1 shows that marital status of respondents has a Beta statistic of .37, which is the second highest value. This means that the marital status of respondents has the ability to explain the variation in the dependent variable (i.e., hold / not hold a savings account).

Furthermore, table 10.1 indicates that the marital status of respondents has a F-ratio of 13.351, which is significant at less than .01 significance level. This means that the predictor "marital status of respondents" has a significant portion (effect) in explaining the variation in the dependent variable (i.e., hold / not hold a savings account).

In addition, table 10.2 and figure 10.2 show that out of the 621 respondents, 535 are married, 240 or 45% of them are "banked with savings accounts" and 295 or 55% are "non-banked". While, out of the 621 respondents, 86 are single, 25 or 29% of them are "banked with savings accounts" and 61 or 71% are "non-banked". This numerical finding concluded that married people are likely to be "non-banked" rather than "banked with savings accounts", by a 10 per cent of the total sample. On the other hand, single people are more likely to be "non-banked" rather than "banked with savings accounts". This is mainly because single people represent the young, earning less, and are not so concerned about their financial situation because they do not have dependents.



FIGURE 10.2: MARITAL STATUS OF RESPONDENTS

SAVERS

NON-SAVERS

Predictor x3: Age Group of respondents:

Table 10.1 indicates that age group of respondents has a Beta statistic of .24. This means that age group of respondents has the fourth rank in terms of its ability to explain the variation in the dependent variable (i.e., hold / not hold a savings account).

Furthermore, table 10.1 shows that age group of respondents has a F-ratio of 11.388, which is significant at less than .01 significance level. This means that, the predictor "age of respondents" has a significant effect in explaining the variation in the dependent variable (i.e., hold / not hold a savings account).

In addition, table 10.2 and figure 10.3 show that, at the age of 24 years and under the percentage of "banked with savings accounts" 19% is lower than the percentage of "non-banked" 81%. At the age 25-44 years the percentage of "banked with savings accounts" 45% is still lower than "non-banked" 55%, but, the difference is smaller (only 10%). At later age stages the pattern is changed, so at age 45-64 years the percentage of "banked with savings accounts" 54% is higher than "non-banked" 46%, but still the difference is small 8%. While, at late age stages (i.e., age group 65 and over), the pattern returns to the situation where the percentage of "banked with savings accounts" 24% is lower than "non-banked" 76%.

This concludes that the percentage of "banked with savings accounts" is mainly high between the middle age groups (i.e., 25-64 years old) which represents the main earning period in most peoples' life cycle. While, the percentage of "non-banked" is higher in early and late life stages (i.e., 24 years and under, and 65 years and over). This is because, at early age stages most persons are in their early life cycle such as students or in early employment. Which means they have less income and are less likely to be "banked with savings accounts". On the other hand, the older groups are mainly retired people living on pensions or savings. These conclusions may help bank marketing management to target the most important age segment of consumers of banking services in Iraq, i.e., the group of consumers aged between 25 to 64 years.





Predictor x4: Family Size Group:

Table 10.1 indicates that family size group of respondents has a Beta statistic of .36. This means that, this predictor has the third rank in terms of its ability to explain the variation in the dependent variable (hold / not hold a savings account).

Furthermore, table 10.1 shows that family size group of respondents has a F-ratio value of 4.579, which is significant at less than .01 significance level. This means that this predictor has a significant portion (effect) in explaining the variation in the dependent variable (i.e., hold / not hold a savings account).

In addition, table 10.2 and figure 10.4 explain that the percentage of "banked with savings accounts" 29% between single people (i.e., people have no dependents at least financially) is lower than the percentage of "non-banked" 71%. The percentage of "banked with savings accounts" between couple without children is increased to 47%, while, the percentage of "non-banked" is decreased to 53% but still the percentage of "banked with savings accounts" is relatively lower than the percentage of "non-banked".

However, for the family size group of a couple with three children and under, the percentage of "banked with savings accounts" 51% and "non-banked" 49%, are about equal (only 2% difference). While, for families consisting of a couple with four children and over, the percentage of "banked with savings accounts" 39% is lower than "non-banked" 61%.

Thus, in the case of single persons (no dependents), as previously mentioned, this group mainly consists of young single people, earning less, and less concerned about their future financial status because they do not have dependents.

However, by having children the results, as indicated in table 10.2 and figure 10.4, showed that the more the number of children a family have, the less the chance to be able to save. Thus, the percentage of "banked with savings accounts" decreases when the family size increases. Table 10.2 shows conclusively that the more children there are in a family, the less likely is the opportunity for the household to have a savings account.

10.2.2: Consumers' Socio-economic Characteristics:

Three socio-economic predictors were used in the analysis, and they are the following:

Predictor x5: Family Income Group:

Table 10.1 shows that the predictor family income group has a Beta statistic of .13. This means that it ranks fifth in terms of its ability in explaining the variation in the dependent variable (i.e., hold / not hold a savings account).

In addition, table 10.1 indicates that this predictor has a F-ratio of 29.315, which is significant at less than .01 significance level. In other words, the predictor family income group has a significant portion (effect) in explaining the variation in the dependent variable, i.e., hold / not hold a savings account.

Table 10.2 and figure 10.5 show that the percentage of "banked with savings accounts" 29% for the first group income (under 2000 I.D.) is lower than the percentage of "non-banked" 71%. The same conclusion can be made about the second income group (2000 - 2999 I.D.), i.e., 26% "banked with savings accounts" and 74% "non-banked".



FIGURE 10.4: FAMILY SIZE GROUP OF RESPONDENTS





In the case of the third income group (3000 - 3999 I.D.) the percentage of "banked with savings accounts" 59% is higher than the percentage of "non-banked" 41%. While, for the fourth income group (4000 I.D. and over) the percentage of "banked with savings accounts" 69% is far higher than the percentage of "non-banked" 31%. Figure 10.5 shows the changes in the percentage of "banked with savings accounts" and "non-banked" according to their income group.

However, the above results indicated that the higher the family income the more likely they are to have a savings account. On the other hand, the lower the family income the less chance they have a savings account.

Predictor x6: Household Head's Educational Level:

Table 10.1 shows that the predictor household educational level has a Beta statistic of .11 which ranks as sixth in terms of its ability in explaining the variation in the dependent variable (i.e., hold / not hold a savings account).

Also, table 10.1 shows that this predictor has a F-ratio of 5.578 which is significant at less than .01 significance level. Thus, to say that the predictor household educational level has a significant portion (effect) in explaining the variation in the dependent variable.

In addition, table 10.2 and figure 10.6 show that people of higher education are more likely to be "banked with savings accounts", and people with less education are likely to be "nonbanked". For example, in the case of people with less than secondary school education 30% are "banked with savings accounts" and 70% are "non-banked". In the case of people with secondary school education, the percentage of "banked with savings accounts" increased to 47%, while, the percentage of "non-banked" decreased to 53%.

In the case of people with university education the percentage of "banked with savings accounts" increased to 51%,

while, the percentage of "non-banked" decreased to 49%. Finally, in the case of people with higher education (M.Sc. and Ph.D) the percentage of "banked with savings accounts" increased further to 63%, while, the percentage of "non-banked" decreased to 37%.

Predictor x7: Household Head's Occupation:

Table 10.1 shows that this predictor is the predictor of the highest Beta statistic value .41. Which means, it is the predictor variable of the highest ability in explaining the variation in the dependent variable (i.e., hold / not hold a savings account).

Also, table 10.1 indicates that the predictor household head's occupation has a F-ratio of 10.467, which is significant at less than .01 significance level. In other words, it has a significant portion (effect) in explaining the variation in the dependent variable, i.e., hold / not hold a savings account.

In addition, table 10.2 and figure 10.7 explain that people with better jobs are more likely to be "banked with savings accounts" than people with poor jobs or no jobs at all. For example, 83% of the owners of private business are "banked with savings accounts", while only 11% of labourers are "banked with savings accounts". And, 80% of general managers and people in higher positions are "banked with savings accounts", while, only 28% of people in lower clerical jobs are "banked with savings accounts".

On the other hand, the differences between the percentage of "banked with savings accounts" and "non-banked" in case of other jobs are relatively marginal. For example, 48% of housewives are "banked with savings accounts", and 52% are "non-banked". And, 46% of students are "banked with savings accounts", and 54% are "nonbanked". And, the same thing applies in the case of teachers 48% are "banked with savings accounts" and 52% are "non-banked".



FIGURE 10.6: EDUCTIONAL LEVEL OF RESPONDENTS

FIGURE 10.7: OCCUPATIONS OF RESPONDENTS





10.3 Testing Research Hypothesis:

In this chapter the below mentioned research hypothesis, as stated in its null form, is to be tested by using the parametric Ftest. The idea was to underline the significance of all independent variables taken together, which consists of consumers' profile characteristics, i.e., socio-economic and demographic, in explaining the variation in the dependent variable, i.e., hold / not hold a savings account. After achieving that, the next stage was to test the significance of each predictor (independent variable) used in the analysis in explaining the variation in the dependent variable, i.e., hold / not hold a savings account.

Hypothesis HO: Consumers' profile characteristics, i.e., the seven socio-economic and demographic variable taken together or separately, have no significant influence on consumers' saving decisions (hold / not hold a savings account) in Iraq.

 <u>1. Testing the significance of all predictors variables taken</u> together.

In order to test the significance of the seven predictors variables taken together in explaining the variation in consumers' saving decisions the value of F-test was calculated as follows:

46.473 / (30 - 7)

F = _____

105.444 / (621 - 30 + 7 - 1)

F(23,597) = 11.439

Tabulated F-value (23,597) = 1.84 at p = .01 significance level. Thus, the calculated F-value exceeds the tabulated F-value, i.e., it is significant at p = 1ess than .01. Therefore, the null hypothesis

is rejected in favour of the alternative hypothesis, that is consumers' profile characteristics have a significant effect in explaining the variation in consumers' saving decisions (hold / not hold a savings account). Thus, the seven socio-economic and demographic characteristics of consumers can influence their saving decisions.

2. Testing the significance of each predictor variable taken separately.

In order to test the significance of each predictor variable in explaining the variation in consumers' saving decisions the value of F-test was calculated for each predictor variable as follows: <u>Sex of Respondents</u>

The calculated value of F-test for the predictor variable "sex of respondents" is equal to:

.6228 / (2 - 1)

F = _____

151.916 - .6228 / (621 - 2)

F(1,619) = 2.548

Tabulated F-value (1,619) = 3.84 at p = .05 significance level. Thus, the tabulated F-value exceeds the calculated F-value, i.e., it is not significant at all. Therefore, the null hypothesis is accepted, i.e., sex of respondents has no significant effect in explaining the variation in consumers' saving decisions (hold / not hold a savings account). Thus, respondents' sex is not of significant importance in explaining why some respondents in Iraq decide to be savers and others not.

Marital Status of Respondents

The calculated value of F-test for the predictor variable "marital status of respondents" is equal to:

1.8996 / (2 - 1)

151.916 - 1.8996 / (621 -2)

F(1,619) = 7.838

Tabulated F-value (1,619) = 6.63 at p = .01 significance level. Thus, the calculated F-value exceeds the tabulated F-value, i.e., it is significant at p = 1ess than .01 significance level. Therefore, the null hypothesis is rejected in favour of the hypothesis that, marital status of respondents has a significant effect in explaining the variation in consumers' saving decisions. This finding indicate that marital status of respondents in Iraq can influence their saving decisions.

Age Group of Respondents

The calculated value of F-test for the predictor variable "age group of respondents" is equal to:

8.0466 / (4 - 1)

F =

151.916 - 8.0466 / (621 - 4)

F(3,617) = 11.503

Tabulated F (3,617) = 3.78 at p = .01 significance level. Thus, the calculated F-value exceeds the tabulated F-value, i.e., it is significant at p = less than .01 significance level. Therefore, the null hypothesis is rejected in favour of the hypothesis that age group of respondents has a significant effect in explaining the variation in consumers' saving decisions, i.e., it can explain why some respondents decide to be savers and others not.

Family Size of Respondents:

The calculated value of F-test for the predictor variable " family size of respondents" is equal to:
3.5868 / (4 - 1)

151.916 - 3.5868 / (621 - 4)

F(3,617) = 4.973

Tabulated F (3,617) = 3.78 at p = .01 significance level. Thus, the calculated F-value exceeds the tabulated F-value, i.e., it is significant at p = less than .01 significance level. Therefore, the null hypothesis is rejected in favour of the hypothesis that, family size group of respondents has a significant effect in explaining the variation in consumers' saving decisions. In other words, the size of the family can determine respondents saving decisions in Iraq.

Income Group of Respondents

The calculated value of F-test for the predictor variable "income group of respondents" is equal to:

20.4024 / (4 - 1)

F = ----

151.916 - 20.4024 / (621 - 4)

F(3,617) = 31.906

Tabulated F (3,617) = 3.78 at p = .01 significance level. Thus, the calculated F-value exceeds the tabulated F-value, i.e., it is significant at p = 1ess than .01 significance level. Therefore, the null hypothesis is rejected in favour of the hypothesis that, family income group of respondents has a significant effect in explaining the variation in consumers' saving decisions (hold / not hold a savings account). This finding explain that income of respondents has an influence on their saving decisions, since, as it is known, income plays an important role in consumer saving behaviour, thus, the higher the income the more opportunity to save.

Educational Level of Respondents

The calculated value of F-test for the predictor variable "educational level of respondents" is equal to:

151.916 - 7.246 / (621 - 4)

$$F(3,617) = 10.301$$

Tabulated F (3,617) = 3.78 at p = .01 significant level. Thus, the calculated F-value exceeds the tabulated F-value, i.e., it is significant at p = less than .01 significance level. Therefore, the null hypothesis is rejected in favour of the hypothesis that, educational level of respondents has a significant effect in explaining consumers' saving decisions (hold / not hold a savings account). Thus, respondents' educational level has an influence on their saving behaviour (decisions).

Occupation of Respondents

The calculated value of F-test for the predictor variable "occupation of respondents" is equal to:

44.2491 / (10 - 1)

F = _____

151.916 - 44.2491 / (621 - 10)

F(9,611) = 27.902

Tabulated F (9,611) = 2.41 at p = .01 significance level. Thus, the calculated F-value exceeds tabulated F-value, i.e., it is significant at p = less than .01 significance level. Therefore, the null hypothesis is rejected in favour of the hypothesis that, occupation of respondents has a significant effect in explaining the variation in consumers' saving decisions (hold / not hold a savings

account). In other words, occupation of respondents can influence their decisions to be savers or non-savers.

10.4 Conclusions:

The main objective of this chapter was to examine the relationship between consumers' saving decisions (hold / not hold a savings account), and consumers' profile characteristics such as: income, educational level, occupation, sex, age, marital status, and family size.

Therefore, in order to achieve this objective, Multiple Classification Analysis (MCA) was used in the analysis. The output of MCA provided several types of statistics, which helped in examining the relationship between consumers' saving decisions and consumers' profile characteristics.

However, the principal aims of this chapter were: (a) to examine the extent to which consumers' profile characteristics (the seven predictors taken together) contributed in explaining the variation in consumers' saving decisions; and (b) to determine the significance of consumers' profile characteristics in explaining the variation in consumers' saving decisions. The coefficient of determination (R)² value equal to (30) per cent, and the F-ratio value of (11.440) which it is significant at p = less than .01 level confirmed that there is a relationship between consumers' profile characteristics and consumers' saving decisions.

In other words, the coefficient of determination $R^2 = 30$ per cent refers to the fact that the seven profile characteristics were able to explain the variation in consumers' saving decisions. Thus, consumers' decisions are usually affected by different factors such as: consumers' personal characteristics, consumers' attitudes

towards the object, attributes of the object itself, social factors, motivations and advantages can be gained, and many other factors.

However, all the above mentioned factors have their contribution in explaining the variation in consumers' decisions. Therefore, the finding of this investigation is that consumers' profile characteristics contributed by 30 per cent in explaining the variation in consumers' saving decisions referred to two facts:

Firstly, in addition to consumers' personal characteristics there are some other consumers' attitudinal factors and service attributes, which can play an important role in explaining the variation in consumers' saving decisions, $R^2 = 70$ per cent such as: consumers' perceptions, attitudes, psychological characteristics, and lifestyle. Secondly, in spite of that, consumers' profile characteristics seemed to be of significant importance in explaining the variation in consumers' saving decisions.

The second set of aims was to examine the ability, and the significance of each predictor in explaining the variation in consumers' saving decisions. The output of MCA programme provided the required statistics to achieve such aims.

Firstly, the Beta values of the predictors used in the analysis concluded that: occupation of respondents came first in terms of its ability in explaining the variation in consumers' saving decisions which was followed by marital status (second), family size (third), age (fourth), income (fifth), educational level (sixth) and sex (seventh or the lowest Beta value).

Secondly, the F-ratio values of the same predictors used in the analysis concluded that six predictors were of significant importance in explaining the variation in consumers' saving decisions, and they are: Marital status, family size, age, family

income, educational level, and occupation. The sex of respondents, which has the lowest Beta value was found to be insignificant in explaining the variation in consumers' saving decisions.

The above conclusions were again confirmed by using the Ftest in testing the stated research hypotheses. Thus, all predictors taken together and each predictor taken separately were found to be of significant effect in explaining the variation in consumers' saving decisions, with only one exception being that the sex of respondents was found to be of no significance.

Sex of respondents in this study (i.e., the Iraqi society) was found to be of no significant influence on consumers saving decisions. However, it is possible to suggest that the main reasons for such a finding could be the following: (a) as in many Middle Eastern societies women play a minor role and men are dominant in the society; (b) women generally have less financial independence, or in other words, family financial affaires are conducted by men within the family; and (c) perhaps the lack of female participation in this survey (only 19% women) was one of the reasons for such a conclusion. As is well known in many Middle Eastern societies, female participation in marketing research surveys is one of the major problems facing many marketing researchers. The findings in relation to the influence of other consumers' characteristics such income, marital status, education, family size and as age. occupation on consumers' saving behaviour (decisions) are comparable to the findings of similar studies carried out in different environments (countries), as mentioned earlier in chapters three and nine of this investigation.

Finally, based on the main findings of the MCA output the main types of Iraqi consumers and their profile characteristics can be summarised as described in table 10.3.

Table 10.3 Main groups and characteristics of consumers.

Group Name	Group Characteristics
Good saver, or likely to be saver.	 Male, married, 45-64 years old has: a. three or less children, b. a very high income (4000 I.D. and over), c. higher education (University graduate and over), and d. better occupation in terms of earning.
Saver, or can be a saver.	 Male, married, his age (24-44), has: a. three or less children, b. a good income (3000-3999 I.D.), c. a good educational level (University graduate and over, and d. upper class occupation in terms of earning.
Saver / non-saver, or can be saver or non- saver depends on his / her situation.	 Male or female, married or single, his / her age is 24 and over, has or has not children, has: a. an average income (2000-2999 I.D.), b. an average education (Secondary school or University graduate), and c. middle class occupation in terms of earning.
Non-saver / or likely to be non-saver.	Male or female, married or single, very young (24 and under) or very old (65 and over), has: a. no children or too many of them (four children and over), b. a very low income (less than 2000 I.D.), c. a very low education (less than Secondary school), and d. lower class occupation in terms of earning.

CHAPTER ELEVEN

CONCLUSIONS

- 11.1 RESEARCH CONCLUSIONS.
- 11.2 RESEARCH CONTRIBUTIONS.
- 11.3 RESEARCH LIMITATIONS.
- 11.4 RECOMMENDATIONS FOR FURTHER RESEARCH.

.

Ł

11.1 Research Conclusions:

Banks activities depend on their deposit taking system. In other words, banks accept money (deposits) from depositors and pay them interest on it in order to lend it to other customers who need it to finance their businesses. Therefore, at the end of the day banks will gain the difference between the lower interest they pay and the higher interest and other service charges they charge as a profit. Thus, in this system, as explained in figure 11.1, there are three main players: The first player is the bank who makes profit out of it as mentioned above. The second player is the depositor who gains interest on their savings and security to their money. The third player is the borrower who pays interest and other service charges to the bank for giving him / her a loan, then investing the loan into businesses and gaining higher return on it than the paid interest.



Figure 11.1: The money circulation between banks and customers.

Thus, given the importance of the deposits to banks as explained in figure 11.1, commercial (retail) banks found that in order to encourage consumers to deposit their money at the bank, they should adopt a new marketing approach, which should be more consumer-oriented rather than product-oriented.

However, this thesis is the result of a study of consumers' saving behaviour carried out through an empirical investigation of consumers' attitudes towards commercial banking services with special reference to savings accounts in Iraq. The main reasons for conducting this investigation can be summarised in the following: (a) There is an urgent demand to provide an understanding of consumers' behaviour and their needs and wants in matters related to banking in Iraq; (b) It is hoped that this investigation will marketing about literature of bank the contribute to the underdeveloped countries which is very limited, and in some countries, such as Iraq, there is no single attempt carried out to investigate this subject neither by banks nor by researchers; (c) Help in improving the quality of banking services, especially savings accounts, through effective marketing; (d) Establishing the extent to which the comprehensive models of consumer behaviour are applicable in investigating consumers' saving behaviour in Iraq; and (e) Relating consumers' attitudes, motivations, psychographic, socio-economic and demographic variables to their saving behaviour in Iraq. Therefore, a number of research objectives and related hypotheses were stated to guide this research's empirical part as mentioned in chapter one.

It was concluded in chapter four of the present investigation that the main purposes of bank marketing are creating and delivering satisfactory services to consumers at a profit to the bank, as

described in figure (11.1). The marketing concept for banks means that banks must recognise that bank customers are changing in terms of their wants, needs, problems and behaviour. Therefore, banks should meet these changes through customer oriented marketing strategies. Market research in banking is a method of carrying-out empirical investigations about aspects of problems related to the marketing concept in banking.

Through reviewing earlier similar and comparative marketing research studies, i.e., those studies which are related specifically to consumers' saving behaviour, the following conclusions were derived: (a) Consumers' saving behaviour is subject to great influence by their attitudes, motivations, socio-economic, demographic and psychographic characteristics; and (b) Modern consumers seek not only a good return on their savings, but they seek convenience, respect, personal relationships with the bank and availability of modern banking facilities.

However, it was concluded in the present investigation that no attempt was made by previous studies to comprehensively investigate consumers' saving behaviour as a separate concept. In other words, most previous studies concentrated on one set or a few more sets of variables which may influence consumers' saving decisions, but no attempt was made to include all the possible sets of variables to do the job. Also, most previous studies analysed consumers! saving decisions through one dimension, such as discriminating between customers of Banks and Savings and Loan Associations. No attempt was made to analyse consumers according to other dimensions than their bank selection such as: consumers' saving decisions, saving habits, age, income, family size. educational level etc.

In Iraq, as is the case in most underdeveloped countries, the hanking system is still underdeveloped if compared with those of the developed countries. Most of the problems facing the present Iraqi system are the creation of nationalisation, i.e., banking transforming a bank ownership from the private to the public sector. Thus, with banks in Iraq being nationalised in 1964, the number of banks operating in the country has decreased from seventeen in its peak before nationalisation to only seven at present time, including the Central Bank of Iraq and other specialist banks. At the present time there are only two commercial (retail) banks in Iraq.

In such a banking system, where banks are state-owned, the first victims were consumers of banking services who were badly treated, neglected and poorly served and unfortunately for them, they have no other options but to deal with the existing unwanted banks. Therefore, one of the benefits of the present investigation is underlining the present problems and shortcomings in the performance of the commercial (retail) banks in Iraq, as discussed in chapter two of this study. Then, attempting to understand Iraqi consumers in terms of their wants, needs, desires and problems in order to improve the banking services for them in the future.

It was concluded in the discussions of the comprehensive models of consumer behaviour and the multi-attribute models of consumers attitudes in the present investigation that consumers' saving behaviour can be related to these models. Therefore, based on the models of consumer behaviour a model of consumer banking behaviour in Iraq was developed, as presented in figures 4.4, 4.5, 4.6 and 5.1 of chapters four and five of this investigation. This study's theoretical model of consumer banking behaviour was used as the base to establish the empirical part of the research, i.e., the

collection of data and its analysis. The empirical finding of the present investigation provided statistically tested results proving that a method of investigating consumer saving behaviour (decisions) can be achieved.

The design of the present investigation was based on the faceted design of the universe of content under investigation. Therefore, all the possible facets, sub-facets and items (variables) which are applicable to the problem under investigation, i.e., consumers' saving behaviour in Iraq, were included in the faceted research design. Thus, by adopting the faceted design as a research design for the present investigation it has been concluded that this study is:

A study of the cognitive aspects of individual consumers as perceived by them at the present time with regard to the aspects of, Context of Consumers' Saving Behaviour in Iraq, which are the following:

D1 Consumers' attitudes towards savings accounts.

D2 Bank attributes.

D3 Consumers' financial psychographic characteristics. Towards the study's chief objective of investigating consumers' saving behaviour (decisions) in Iraq empirically, with special reference to opening savings accounts at commercial banks in Iraq.

However, in order to conduct the empirical part of this investigation a research questionnaire was designed to collect the primary data from respondents. A total sample of 800 respondents was selected from Iraq using two methods of probability sampling, area and systematic sampling. An overall response rate of 77.63 per cent was secured.

The collected data was then analysed according to the research objectives by using three major statistical techniques of data analysis available in the SPSS-X statistical package, these are: Factor Analysis, Discriminant Analysis and Multiple Classification Analysis. In addition, three statistical tests, the Kolmogorov-Smirnov Goodness-of-Fit One Sample Test, the F-Test and the T-Test were used to test the significance of the relevant research hypotheses.

The primary objective of using Factor Analysis in the present investigation was to identify a set of factors which are capable in explaining the following: How SAVERS, i.e., consumers who are banked with savings accounts, perceived the importance of 60 attitudinal variables in influencing their decisions of having savings accounts at commercial banks in Iraq with a minimal loss of information.

In order to achieve objective one of this investigation, Factor Analysis was carried out. The findings of Factor Analysis successfully identified thirteen factors which are capable of explaining SAVERS, i.e., respondents who are banked with savings accounts, attitudes towards the variables related to the service of savings accounts provided by commercial banks in Iraq with a minimal loss of information. In fact, a number of 57 variables, out of the original 60 attitudinal variables used in the analysis, were loaded with at least one of the factors extracted by Factor Analysis.

The results of the Kolmogorov-Smirnov test confirmed that respondents who are banked with savings accounts, in general, share the same attitudes towards the service of savings accounts provided by commercial banks in Iraq. In other words, they attached similar rates of importance to the variables which may influence their saving decisions.

order to achieve objective two of this investigation, Tn Two-Group Discriminant Analyses was carried out to discriminate between respondents according to their attitudes towards the 65 attitudinal variables used in the analysis. The discrimination between respondents was based on their saving decisions, i.e., banked with savings accounts vs. non-banked. The findings of Two-Group Discriminant Analysis confirmed that there are differences between consumers if divided according to their saving decisions in they attached to the variables terms of the rates of importance related to the service of savings accounts provided by commercial banks. Thus, the discriminant function extracted 18 variables of significant importance in discriminating between the two groups of respondents in terms of their attitudes towards the idea of saving and having savings accounts at commercial banks in Iraq.

of the t-test confirmed that there are The findings differences between consumers if divided according to their saving habits, i.e., good vs. poor savers, in terms of their attitudes towards the variables related to the service of savings accounts provided by commercial banks in Iraq. Thus, the results of the ttest indicated that 16 variables were of significant importance in underlining the dissimilarities between good and poor savers. In addition, the classification results confirmed that not all good savers are saving through savings accounts in Iraq. Thus, out of the 403 respondents who said that they were good savers, only 265 of them hold savings accounts, while the remaining 138 good savers appeared to be using other methods of saving, such as: keeping money in a secure place at home or transforming money into gold and jewelry instead of keeping it at a bank. In fact, such consumer behaviour in a country like Iraq is the result of many factors such

as: public distrust in banks because of political and economical instability, poor banking services, consumers dissatisfied with the treatment they receive from bank personnel, and a high level of illiteracy especially in matters related to banks and banking services.

The findings of the t-test confirmed that there are differences between respondents of two age groups (group A: 44 years and under vs. group B: 45 years and over) in terms of the rates of importance they attached to the attitudinal variables related to the service of savings accounts. A number of 4 variables identified by the t-test as of significant importance in underlining the dissimilarities between the two age groups of respondents.

The findings of the t-test confirmed that there are differences between respondents of two family size groups (group A: families of 5 and under vs. group B: families of 6 and over) in terms of their attitudes towards the idea of saving and having savings accounts at commercial banks. The t-test identified a number of 5 variables of significant importance in explaining the dissimilarities between the two family size groups of respondents.

The findings of the t-test confirmed that there are differences between respondents of two income groups (group A: 2999 I.D. and under vs. 3000 I.D. and over) in terms of their attitudes towards the service of savings accounts provided by commercial banks in Iraq. The results of the t-test identified 4 variables of significant importance in explaining the differences in the rates of importance attached by the two income groups of respondents to the variables related to the service of savings accounts provided by commercial banks in Iraq..

Finally, the finding of the t-test confirmed that there are differences between respondents of two educational level groups (group A: graduates of Secondary schools and under vs. group B: graduates of Universities and over) in terms of their attitudes towards the idea of saving and having savings accounts at commercial banks in Iraq. In fact, the t-test extracted 5 variables of significant importance in underlining the dissimilarities in the attitudes of the two educational level of the groups of respondents.

When a comparison was made between the findings of the present investigation's Discriminant Analysis and T-Test with the findings of previous investigations the following conclusions were made: Firstly, the methods and findings of the present investigation confirmed most of the findings of comparative previous investigations. Secondly, the present investigation if compared with previous investigations was more comprehensive in terms of the size of the sample used and the advanced statistical methods of data analysis employed. In addition, it was the first attempt made to study the problem of consumers' saving behaviour and decisions in a Thus, the relevant attitudinal, comprehensive approach. a11 motivational, psychographic, socio-economic and demographic variables which are applicable to the Iraqi consumers' were used in order to investigate the problem of consumers' saving behaviour in Iraq.

Finally, in order to achieve the third objective of the present study, the Multiple Classification Analysis (MCA) was used to examine the relationship between consumers' saving decisions (hold / not hold a savings account) and seven of their socioeconomic and demographic characteristics, such as: Sex; Marital Status; Age; Family size; Income; Educational level; and Occupation.

the MCA confirmed that there 'l'he findings of are relationships between consumers' saving decisions and their socioeconomic and demographic characteristics when taken together and separately, with one exception that sex of respondents was found of no significant importance in influencing consumers' saving decisions.

In addition, the results of the F-test for the relevant research hypotheses confirmed that:

1. Sex of respondents has no significant effect in explaining the variation in consumers' saving decisions (hold / not hold a savings account) in Iraq.

2. Respondents' marital status, age, family size, income, educational level and occupation have significant effect in explaining the variation in consumers ' saving decisions (hold / not hold a savings account) in Iraq.

11.2 Research Contributions:

The present investigation has some theoretical, methodological and practical contributions to the existent knowledge in bank marketing:

1. The Theoretical and Methodological Contributions:

1. A theoretical model of consumer banking behaviour is developed, as presented in figures 4.4, 4.5, 4.6 and 5.1 of this research. In comparison with most general models of consumer behaviour this study's model is developed to deal specifically with the problem of understanding consumer banking behaviour, especially in the underdeveloped countries. Also, it includes the role of some bank and banking services attributes, consumers' financial psychographic, socio-economic and demographic characteristics on consumer behaviour

in banking. Then, the model was empirically used and tested through the process of data collection and analysis, as described in the findings section of this research.

2. The use of marketing research in banking in comparison with other industries, is a relatively recent development. However, most bank marketing researches were conducted in the developed countries, especially in the United States of America. Therefore, one of the theoretical contributions of this investigation is extending the use of bank marketing research techniques in the underdeveloped countries. This study is the first bank marketing research study carried out about Iraq, as one of the underdeveloped countries where the concept of banking and advancement in banking facilities differ significantly from that in the developed countries. Thus, this study's theoretical model and its empirical application in Iraq can be generalised and be used in other underdeveloped countries similar to Iraq in order to provide better understanding for consumer banking behaviour in the underdeveloped world.

3. The present investigation extended the application of Facet Theory, as a systematic description for the universe of content under investigation, in the area of bank marketing research in general and in investigating consumer banking behaviour in Iraq, as one of the underdeveloped countries, in particular.

4. This investigation is a method to study consumers' saving behaviour in a comprehensive approach. In other words, the present investigation used a combination of some attitudinal, psychographic, socio-economic and demographic variables to study consumers' saving behaviour more comprehensively in comparison with previous comparative studies. Thus, more consumer banking behaviour variables

were used in the present study than previous comparative studies in order to provide better understanding of consumer saving behaviour. 5. The present investigation defined the differences between consumers' according to six dimensions of which the most important of them is consumers! saving habits. It provided some new information about why some people, especially in the underdeveloped countries where political and economical instability created public distrust in the banking system, save through methods of savings other than having savings accounts, such as: transferring their savings into gold and other jewelry or keeping them simply in a secure place at their homes.

6. The present investigation provided new discussions about the problems facing many underdeveloped countries who nationalised their banking system in the past when the wave of nationalisation affected most underdeveloped countries. In addition, suggestions were made to help overcome the problems facing the present nationalised (stateowned) Iraqi banking system in terms of what should be done to created repair the damage bγ the nationalisation and if privatisation is the possible solution to the problems.

7. The present study defined the variables which may influence actual savers, i.e., consumers who are banked with savings accounts. Then, a comparison was made between consumers' according to six dimensions, these are: (1) Consumers' saving decisions; (2) Consumers' saving habits; (3) Consumers' age group; (4) Consumers' family size; (5) Consumers' income group; and (6) Consumers' educational level.

8. The present study investigated the concept of consumers' saving behaviour, as a separate concept, and analysed accordingly. This was done by examining the relationship between consumers' saving

decisions and their attitudes, motivations, financial psychographic, socio-economic and demographic characteristics.

9. The present investigation employed a hybrid method of sampling through using a combination of two methods of probability random sampling, area and systematic sampling, to derive a representative sample of the population under investigation. In addition, because the case in many underdeveloped countries, in Iraq, as is statistical data about population of a city is impossible to obtain, the locally selected representative of district records was used as a base for the purpose of selecting the sample from each district. This is the only acceptable way by which reliable data can be obtained about population in a country like Iraq. People in Iraq keep contact with local representatives because they need them to certify that they are living in that particular district when they process any formal transactions with the authorities. In addition, the decision was made to divide the City of Baghdad into 20 districts and the prime consideration in making the division was the standard of living in each district in order to include wealthy, middle class and poor districts. Then, from each district, 40 respondents were selected of which 20 were "banked with savings accounts" and 20 were "non-banked". This study is distinct among other marketing research studies in terms of sampling procedures. Dividing a big city into districts according to residents' standard of living, then selecting an equal number of respondents from each district of which half should be "banked with savings accounts" and the other half "non-banked" created a new approach in solving the problem of sampling in the underdeveloped countries (for more details see chapter six).

10. The findings of the present investigation provided more information about consumer saving behaviour in comparison with previous comparative studies, as follows: (a) It identified the main factors which are perceived as important by actual savers in influencing their decision to hold a savings account at a commercial bank; (b) It identified the similarities between actual savers in terms of their attitudes towards the service of savings accounts provided by commercial banks on one hand, and the dissimilarities between them and non-savers on the other hand; (c) It identified some financial psychographic characteristics which have an influence on consumers saving decisions; and (d) the relationship between a set of consumers socio-economic and demographic characteristics and their saving decisions was analysed and reported. Therefore, it can be stated that the present study provided a better and more comprehensive understanding of consumer saving behaviour, especially in the underdeveloped countries.

11. The present investigation is the first to use the Multiple Classification Analysis computer programme as a method to examine the relationship between consumers' saving decisions and their socio-economic and demographic characteristics.

2. The Practical Contributions:

1. It is hoped that the discussions, issues, proposals and findings made in the present investigation will contribute to efforts to overcome the problems and poor performance facing the Iraqi banks and the entire Iraqi banking system. Also, because the problems which are facing the Iraqi banking industry are common factors in many underdeveloped countries, it is hoped that other countries

which are similar to Iraq will benefit from the conclusions made in the present investigation.

2. The findings of the present study provide some benefits in regard to encouraging the idea of saving and mobilisation of finance in Iraq. Thus, in Iraq, as is the case in many underdeveloped countries, public distrust in the banking system led to less use of saving facilities provided by banks and other financial institutions. Therefore, the practical findings of the present research have three levels of benefit:

a. At the national level: Defining consumers attitudes, notivations and characteristics is the first step in improving the services relating to saving facilities provided by banks and encouraging consumers to save. Thus, this study provides some information about the banking behaviour of Iraqi consumers. This can help the Iraqi state-owned banks to perform better and gain more customers. This will help the government in its efforts to achieve greater mobilisation of finance, and better use of national financial resources within the national economy. b. At the banking system level: Improving the service of savings accounts and its related facilities by banks means more savers who want to open savings accounts at the banks. In other words, more deposits means increasing the lending ability of the banks to provide loans for other customers

and gaining more profits out of it.

c. At the consumers level: The findings of the present investigation defined the most important variables which may effect consumers saving decisions. Therefore, it is the banks duty then to improve their services which at the end

of the day will benefit consumers and encourage them to be savers for their own benefit.

3. The present study can be used as a practical reference by the banks, especially in Iraq, in establishing how to conduct a bank marketing research to solve the existing and possible future banking problems related to consumers.

11.3 Research Limitations:

The present investigation is subject to the following limitations:

1. As the present study was conducted in an underdeveloped country, i.e., Iraq, the selection of facets, sub-facets, and items (variables) from the universe of content according to the research faceted design was affected by the level of advancement in the banking sector, i.e., the banking services and facilities available to consumers. Also, it was affected by Iraqi respondents overall understanding and knowledge of the variables under investigation. In other words, some variables were excluded because Iraqi respondents have no knowledge of them because of their nonexistence in Iraq, such as: variable related to the automatic banking services.

2. Consumers' saving behaviour is a very complex subject and there are a huge number of variables which may affect it. However, the selection of the variables in the present investigation was limited to the facets defined by this research's facet design. In other words, it was limited to the effect of consumers' attitudes, motivations, psychographic, socio-economic, and demographic variables, while other aspects such as, for example, government regulations and the effect of political stability on consumer behaviour, were excluded from this investigation because they are out of the scope of this study and require separate investigation.

3. The study was an empirical investigation of consumers' attitudes towards commercial banking services, with special reference to savings accounts in Iraq, and because banks, banking services and consumers banking knowledge in Iraq are underdeveloped the present research was affected largely by this fact.

4. This study is a bank marketing research carried out in a U.K. university, but the empirical investigation is about an underdeveloped country, i.e., Iraq. Therefore, as is the case with most studies carried out about underdeveloped countries, the present study suffers from the following technical limitations:

a. The selection of the sample of respondents for the purpose of data collection was a major difficulty in this investigation. Thus, as is the case in many underdeveloped countries, there are no up-to-date accurate statistics about the population and, if it exists access to them is impossible to obtain. For example, in the present investigation data about population socio-economic and demographic characteristics was not available and only a map of the city was available. Therefore, the decision was made to divide the city into blocks, chunks, units and houses, then respondents were selected and interviewed in their houses to collect the required data.

b. Respondent error was another difficulty in this investigation with the result that although 800 questionnaires were delivered to respondents personally (hand-to-hand) assistance and was provided to most respondents, only 621 usable questionnaires were secured. c. The above mentioned technical difficulties caused costs in time, effort and money.

d. Female participation in the present investigation was low, only 18.8 per cent of the total sample being female. In fact, this is a problem in most Middle Eastern Countries where female participation in surveys is affected by the traditions of that part of the World.

5. The present research attitudinal (ordinal) data is measured by a 5-Point Likert Scale. Therefore, the selection of statistical techniques of data analysis was affected by the limitations of using the Likert Scales. However, most researchers treat the data from such scales as if they were equal to interval in nature, since the results of most standard statistical techniques, such as Factor Analysis and Discriminant Analysis, are not affected by small deviation from the interval requirements.

6. Finally, the undertaking of this investigation was affected by the unhappy events in Iraq during the Gulf Crisis, which affected the study in terms of obtaining further data about the problem under investigation and in terms of financial resources available to carry on the investigation.

11.4 Recommendations for Further Research:

This thesis was a result of a study of consumers' saving behaviour in Iraq. In addition, the present investigation was an extension to the applications of consumer behaviour theories into the area of consumers' banking behaviour in general and consumers' saving behaviour in particular. Therefore, more research is needed to cover other aspects and dimensions related to this subject in order to provide better understanding of consumers' banking behaviour. However, it has been felt that one of the duties of this research is to recommend areas for further research in consumers' banking behaviour:

1. More research is needed to investigate consumers' saving behaviour and if possible in a more comprehensive approach. That can be done through employing not only consumers' attitudes, motivations and other personal characteristics, but, employing other possible variables which may have an effect on consumers' saving behaviour such as:

a. The role of government and banks regulations on consumers' saving behaviour.

b. The effect of consumers' life style and beliefs, such as religion, on consumers' saving behaviour.

c. The role of World and each nation's political and economical situation, i.e., stability and instability, on consumers saving decisions.

2. More research is needed to investigate the differences between different segments of savers. For example, the differences between savers who save through savings accounts and others who save through keeping their money at home or transferring it into gold and other jewelry.

3. Further research can be done to investigate the differences between savers and investors. In other words, why some people choose to save at a bank while others choose investment by themselves or again through a bank and taking the risk, i.e., low risk takers (savers) vs. high risk takers (investors).

4. Further research can be done to extend the methods used in the present investigation to investigate the use of other financial and banking services.

5. In the present investigation the relationships between consumers' saving decisions (hold / not hold a savings account) and their socio-economic and demographic characteristics were investigated.

But, further research is needed to examine the relationships, for example, between consumers' saving habits (frequent vs. infrequent savers or good vs. poor savers, and so on) and their socio-economic and demographic characteristics.

6. More research is needed to investigate the psychology of consumers' of banking services in terms of what their wants and needs are and how they think about the banking services provided for them.

7. More research can be done to investigate the role of the new banking technology on consumers' perception of banks and the services they provide especially in those countries who have that technology.

8. More bank marketing research should be done in the underdeveloped countries in order to reach consumers and understanding them. Thus, consumers in most underdeveloped countries are neglected, badly treated and left unsatisfied but they have no option but to deal with the existing banks.

9. More research can be done to investigate the differences between consumers' of financial services in terms of their attitudes, wants, needs and opinions and according to their socio-economic and demographic characteristics.

10. The present investigation can be extended to investigate consumers of other institutions who provide some financial services, such as: Building Societies and Post Offices savings accounts facilities. Also, it can be extended to be carried out in other environments, i.e., countries, especially the underdeveloped countries.

References

- Alessandra, A. J., Knierim, L. H. and Yavas, U. (1986), "Some insights into the banking behaviour of older consumers: An empirical study", <u>Management Research News</u>, vol. 4, no. 2, pp. 4-7.
- Al-Kaysee, F. (1967), <u>The Banking System in Iraq after</u> <u>Nationalisation</u>, Baghdad, Iraq.
- Allaire, Y. (1972), "Some marketing thoughts on competition between Banks and near-Banks", <u>Institute of Canadian Bankers</u> <u>Review</u>, vol. 5, part 1, pp. 5-8.
- Al-Shamaa, K. M. H. (1978), <u>A Study of the Iraqi Banking System</u>, Management Research No. 8, Iraq.
- Anderson, D. R. and Nevin, J. R. (1975), "Determinants of young married life insurance purchasing behaviour: An empirical investigation", <u>Journal of Risk and Insurance</u>, vol. 42, pp. 375-387.
- Andrews, F. M., Morgan, J. N. and Sonquist, J. A. (1973), <u>Multiple</u> <u>Classification Analysis: A Report on a Computer Programme</u> <u>for Multiple Regression Using Categorical Predictors</u>, Ann Arbor: Institute for Social Research, The University of Michigan.
- Arora, R., Cavusgil, S. T. and Nevin, J. R. (1985), "Evaluation of Financial Institutions by Banks versus Savings and Loan Customers: An Analysis of Factor Congruency", <u>International</u> <u>Journal of Bank Marketing</u>, vol. 3, no. 3, pp. 47-55.
- Baker, M. J. (1977), "Bank Marketing", <u>The Scottish Bankers</u> <u>Magazine</u>, October, pp. 106-121.
- Baran, R. (1976), <u>Practical Bank Marketing Research</u>, Bank Marketing Association, The U.S.A.

- Berger, R. M. and Patchner, M. A. (1988), <u>Implementing the</u> <u>Research Plan: Guide for the Helping Professions</u>, SAGE Publications.
- Borg, I. (1982), "Some basic concepts of Facet Theory", <u>In: Key Text</u> <u>in Multidimensional Scaling</u>, Edited by Davies and Coxon, Heinemann Educational Ltd., London, pp. 193-235.
- Brien, R. H. and Stafford, J. E. (1967), "The myth of marketing in banking", <u>Business Horizons</u>, vol. 10, Spring, pp. 71-78.
- Canter, D. (1982), "A multivariate model of housing satisfaction", <u>International Review of Applied Psychology</u>, vol. 31, pp. 185-208.
- Canter, D. (1983), "The potential of Facet Theory for Applied Social Psychology", <u>Quality and Quantity</u>, vol. 17, pp. 35-67.
- Capaldini, L. A. (1974), "Bank marketing: The past and the present", <u>In: Marketing for the Bank Executive</u>, Ed: Berry, L. L. and Capaldini, L. A., Leviathan House, London, pp. 4-5.
- Churchill, G. A., Jr. (1987), <u>Marketing Research: Methodological</u> <u>Foundations</u>, Fourth Edition, The Dryden Press.
- Churchill, G. A., Jr. (1988), <u>Basic Marketing Research</u>, The Dryden Press.
- Claycamp, H. J. (1965), "Characteristics of owners of thrift deposits in Commercial Banks and Savings and Loans Associations", <u>Journal of Marketing Research</u>, vol. II, May, pp. 163-170.
- Clowe, K. A. and Diina, A. N. (1978), <u>Basics in Bank Marketing</u> <u>Research</u>, Bank Marketing Association, The U.S.A.
- Crosse, H. D. and Hempel, G. H. (1980), <u>Management Policies for</u> <u>Commercial Banks</u>, Prentice-Hall, Inc., New Jersey. Donnelly, J. H. and Ivancevich, J. M. (1969), "How marketing-

oriented are Commercial Banks?", <u>Banking</u>, The U.S.A, vol. 61, February, pp. 59-60.

- Earl, R. L. and Rose, P. (1976), "Marketing studies on U.S. Banks: What do they tell?, <u>The Canadian Bankers and ICB Review</u>, vol. 83, part 2, pp. 32-35.
- Edris, T. (1989), <u>The Use of Food Nutrition Labelling in the U.K.</u>, Ph.D Thesis, The University of Sheffield, The U.K.
- Elizur, D. and Guttman, L. (1967), "The structure of attitudes toward work and technological change with an organisation", <u>Administrative Science Quarterly</u>, vol. 21, pp. 611-622.
- Engel, J. F. and Blackwell, R. D. (1982), <u>Consumer Behaviour</u>, Fourth Edition, The Dryden Press, Japan.
- Erol, C. and El-Bdour, R., (1989), "Attitudes, behaviour and patronage factors of bank customers towards Islamic banks", <u>International Journal of Bank Marketing</u>, vol. 7, no. 6, pp. 31-37.
- Evans, R. H. and Beckman, M. D. (1974), "Psychographic analysis: An aid for bank management", <u>The Magazine of Bank</u> <u>Administration</u>, vol. 50, January, pp. 32-35.
- Farley, J. U., Howard, A. and Ring, L. W. (1974), <u>Consumer</u> <u>Behaviour: Theory and Applications</u>, Boston, Allyn and Bacon.
- Fishbein, M. (1975), "Attitudes, attitudes change and behaviour: A theoretical review", <u>In: Attitudes Research Bridges the</u> <u>Atlantic</u>, Philip Levine (ed.), Chicago: American Marketing Association, pp. 3-16.
- Fitts, R. L. (1975), <u>Predictions of Usage Rates for Selected</u> <u>Commercial Bank Services: A Cross-Sectional Approach</u> Utilising Socio-economic, Demographic and Attitudinal

<u>Variables</u>, DBA Dissertation, University of Alabawa, The U.S.A.

- Foa, V. C. (1965), "New Development in Facet Design and Analysis", <u>Psychological_Review</u>, vol. 72, pp. 262-274.
- Frank, E. and Green, P. E. (1967), <u>Quantitative Methods in</u> <u>Marketing</u>, Englewood Cliffs: Prentice-Hall, Inc.
- Frank, R. E. and Massy, W. F. (1963), "Innovation and brand choice: The Folger's invasion", <u>In: Toward Scientific Marketing</u>, ed. Stephen A. Greyser, Chicago: American Marketing Association, pp. 99-100.
- Goodfelow, J. H. (1987), "Consumer perceptions and attitudes towards savings and investments", <u>International Journal of</u> <u>Bank Marketing</u>, vol. 5, no. 3, pp. 32-49.
- Goudge, P. and Green, F. (1986), "Child's play? Developing a saving account for kids", <u>Market Research Society Annual</u> <u>Conference</u>, 28th Conference, 1985-1986, pp. 97-109.
- Green, P. E. and Tull, D. S. (1978), <u>Research for Marketing</u> Decisions, Fourth Edition, Prentice-Hall, Inc.
- Griwm, J. L. (1974), <u>Institutional and Demographic Characteristics</u> and Patronage Motives of Consumer Demand Depositors, D.B.A. Dissertation, Kent State University, The U.S.A.
- Guttman, L. (1954), "An outline of some new methodologies for social research", <u>Public Opinion Quarterly</u>, vol. 18, pp. 395-404.
- Guttman, L. (1959), "A structural theory for inter-group beliefs and action", <u>American Sociological Review</u>, vol. 24, pp. 318-328.
- Harby, G. (1985), <u>A Non-Metric Multidimensional Scaling Analysis</u> of Consumers' Attitudes to Life Insurance in Egypt, Ph.D Thesis, The University of Sheffield, The U.K.

- Harman, H. H. (1967), <u>Modern Factor Analysis</u>, University of Chicago, University Press.
- Hawkins, Del. I., Best, R. J. and Coney, K. A. (1986), <u>Consumer</u> <u>Behaviour: Implications for Marketing Strategy</u>, Third Edition, Business Publications, Inc.
- Hindle, T. (1985), <u>Pocket Banker</u>, Basil Blackwell Publisher Ltd. and The Economist Magazine, The U.K.
- Hodges, L. H., Jr. and Tillman, R., Jr. (1968), <u>Bank Marketing: Text</u> and Cases, Addison-Wesley Publishing, Co.
- Howard, J. A. (1977), <u>Consumer Behaviour: Applications of Theory</u>, McGraw-Hill, New York.
- Hussain, S. K. (1982), <u>The Suggestions on Re-organising the Iraqi</u> Banking System, Baghdad University, Iraq.
- Hussain, S. K. (1988), <u>Nationalisation a Case Study on Nationalising</u> <u>the Commercial Banks in Iraq</u>, MPA Dissertation, Liverpool University.
- Jain, A. K., Pinson, C. and Malhorta, N. K., (1987), "Customer loyalty as a construct in the marketing of banking services", <u>International Journal of Bank Marketing</u>, vol. 5, no. 3, pp. 49-72.
- Joy, A., Kim, C. and Laroche, M. (1991), "Ethnicity as a Factor Influencing Use of Financial Services", <u>The International Journal of</u> <u>Bank Marketing</u>, vol. 9, no. 4, pp. 10-16.
- Kaiser, H. F. (1974), "An index of factorial simplicity", <u>Psychometrika</u>, vol. 39, pp. 31-36.
- Kaynak, E., Kucukemiroglu, O. and Odabasi, Y. (1991), "Commercial Bank Selection in Turkey", <u>The International Journal of</u> <u>Bank Marketing</u>, vol. 9, no. 4, pp. 30-39.

- Kaynak, E. and Yavas, U. (1986), "Segmenting the banking market by account usage: An empirical investigation" <u>Journal of</u> Professional Services, vol. 1, nos. 1 and 2, pp. 177-187.
- Kerlinger, F. N. (1973), <u>Foundations of Behavioural Research</u>, 2nd Edition, New York: Holt, Rinehart and Winston.
- Khalaf, F. H. (1977), <u>The Banking Credit and Its Role in The Iraqi</u> <u>Economy</u>, Al-Mustansiriyah University, Iraq.
- Kimball, R. C. (1990), "Relationship Versus Product in Retail Banking", <u>Journal of Retail Banking</u>, vol. 12, no. 1, Spring, pp. 13-25.
- Kinnear, T. C. and Taylor, J. R. (1987), <u>Marketing Research: An</u> <u>Applied Approach</u>, Third Edition, McGraw-Hill, Marketing Series.
- Klecka, W. R. (1980), <u>Discriminant Analysis</u>, A SAGE University Paper, Series no. 19.
- Langrehr, F. W. (1986), "The changing perceptions of financial institutions", <u>Journal of Professional Services Marketing</u>, vol. 1, nos. 1 and 2, pp. 169-175.
- LeBlanc, G. and Nguyen, N., (1988), "Customers' perceptions of service quality in financial institutions", <u>International</u> <u>Journal of Bank Marketing</u>, vol. 6, no. 4, pp. 7-17.
- Lewis, B. R. (1984), "Marketing bank services", <u>The Service</u> <u>Industries Journ</u>al, vol. 4, no. 3, November, pp. 61-67.
- Loudon, D. L. and Della Bitta, A. J. (1984), <u>Consumer Behaviour:</u> <u>Concepts and Applications</u>, Second Edition, McGraw-Hill, Co.

Markin, R. J., Jr. (1974), Consumer Behaviour: A Cognitive

Orientation, MacMillan Publishing, Co., Inc., New York. Martin, P. (1967), "Savings and loans in new submarkets: Search for predictive factors", <u>Journal of Marketing Research</u>, vol. IV, May, pp. 163-166.

- Mason, J. B. and Mayer, M. L. (1974), "Differences between high and low income savings and checking account customers", <u>The Magazine of Bank Administration</u>, vol. 50, June, pp. 48-52, 65.
- Massy, W. F. (1971), "What is Factor Analysis?", <u>In: Multivariate</u> <u>Analysis in Marketing</u>, Ed. D. A. Aaker, Wadsworth Publishing Co., Inc., P. 241.
- Meidan, A. (1974), <u>Non-Metric Multidimensional Analysis of Export</u> <u>Marketing Management</u>, Ph.D Thesis, The University of Bradford, The U.K.
- Meidan, A. (1975, A), "Facet Design: The implication for market research", <u>Der Market</u>, vol. 35, January, pp. 9-12.
- Meidan, A. (1975, B), "When to use Non-Metric Multidimensional Techniques in Market Research", <u>European Research</u>, pp. 58-65.
- Meidan, A. (1984), Bank Marketing Management, MacMillan Education.
- Morgan, N. and Piercy, N. (1990), "Marketing in Financial Services Organisation: Policy and Practice", <u>In: Managing and</u> <u>Marketing Services in the 1990s</u>, ed. R. Teare, L. Moutinho and N. Morgan, Cassell Educational Ltd., pp. 120-136.
- Morrison, D. F. (1976), <u>Multivariate Statistical Methods</u>, McGraw-Hill, New York.
- Moutinho, L. (1982), <u>An Investigation of Vacation Tourist Behaviour</u> <u>in Portugal</u>, Ph.D Thesis, The University of Sheffield, The U.K.
- Mukherjee, B. N. (1965), "A Factor Analysis of some qualitative attributes of Coffee", <u>Journal of Advertising Research</u>,

vol. 5, pp. 35-38.

- Norusis, M. J. (1985), <u>SPSS-X Advanced Statistics Guide</u>, SPSS Inc., Chicago.
- Payne, R. (1976), "Organised climate and job satisfaction: A conceptual synthesis", <u>Organisational Behaviour and Human</u> <u>Performance</u>, vol. 16, pp. 45-82.
- Perez, R. C. (1983), <u>Marketing Financial Services</u>, Prdeger Publishers.
- Person, M. (1964), <u>Marketing and Banking: An Evaluation of Marketing</u> <u>Program Development in Selected Commercial Banks</u>, D.B.A. Dissertation, Indiana University, The U.S.A.
- Peter, J. P. and Olson, J. C. (1987), <u>Consumer Behaviour: Marketing</u> <u>Strategy Perspectives</u>, Richard D. Irwin, Inc.
- Peterson, R. A. (1982), <u>Marketing Research</u>, Business Publications, Inc., Texas, The U.S.A.
- Pezzullo, M. A. (1989), <u>Marketing for Bankers</u>, American Bankers Association, The U.S.A.
- Ramanadham, V. (1984), <u>Public Enterprise and the Developing World</u>, Croom Helm.
- Reed, E. W., Cotter, R. V., Gill, E. K. and Smith, R. K. (1984), <u>Commercial Banking</u>, Prentice-Hall, Inc., New Jersey.
- Reekie, W. D. (1972), "Marketing in Banking", <u>The Bankers Magazine</u>, The U.S.A, August, pp. 56-60.
- Reynolds, F. D. and Wells, W. D. (1978), "Life style analysis: A dimension for Future-oriented bank research", <u>Journal of Bank Research</u>, vol. 9, Autumn, pp. 181-185.
- Robertson, D. H. and Bellenger, D. N., (1977), "Identifying bank market segments", <u>Journal of Bank Research</u>, Winter, pp. 276, 283.

- Ross, W. O. (1968), <u>Marketing in Commercial Banks: How to Proceed</u>, Masterco Press, Inc.
- Runkel, P. and McGarth, J. (1972), <u>Research on Human Behaviour: A</u> <u>Systematic Guide to Method</u>, Holt Rinehart and Winston, New York.
- Sayadian, R. (1989), <u>A Comparative Analysis of Marketing Life</u> <u>Insurance Policies in the U.K.</u>, Ph.D Thesis, The University of Sheffield, The U.K.
- Shankleman, E. (1979), "The clearing banks' slow struggle to be corporate marketers", <u>Marketing</u>, vol. 4, part 2, pp. 156-162.
- Sheth, J. N. (1970), "Multivariate Analysis in Marketing", <u>Journal</u> of <u>Advertising Research</u>, Vol. 10, February, pp. 29-39.
- Sheth, J. N. (1971), "The Multivariate Revolution in Marketing Research", <u>Journal of Marketing</u>, vol. 35, January, pp. 13-19.
- Shye, S. (1978), <u>Theory of Construction and Data Analysis in</u> <u>Behavioural Sciences</u>, Holt Rinehart and Winston, Inc., New York.
- Siegel, S. (1956), <u>Non-Parametric Statistics for the Behavioural</u> <u>Sciences</u>, McGraw-Hill Ltd.
- Simon, J. L. (1969), <u>Basic Research Methods in Social Sciences: The</u> Art of Empirical Investigation, New York: Random House.
- Stevens, S. S. (1946), "On the theory of scales of measurements", Science, no. 103, pp. 677-680.
- Stone, R. (1982), <u>Social Change in Israel: Attitudes and Events</u> (1967-1979), Prager Publishers, New York.
- Tan, C. T. and Chua, C. (1986), "Intention, attitude and social influence in bank selection: A study in an Oriental
Culture", International Journal Of Bank Marketing, Vol. 4,

no. 3, pp. 43-53.

- The Banking Act No. 43, 1947, Iraq.
- The Banking Act No. 72, 1956, Iraq.
- The Banking Act No. 64, 1964, Iraq.
- The Banking Act No. 100, 1964, Iraq.
- The Central Bank of Iraq, Yearly Bulletins, 1971-1989.
- The Central Bank of Iraq, Economic Review, 1972.
- The Iraqi Constitution, Article No. 108, 1924.
- The Iraqi Ministry of Finance, <u>The Banking Acts and Regulations</u>, Iraq, 1976.
- The Iraqi Ministry of Finance, Economic reviews, 1980-1989.
- Tull, D. S. and Hawkins, D. I. (1990), <u>Marketing Research:</u> <u>Measurements and Methods</u>, Fifth Edition, MacMillan Publishing Company.
- Walters, C. G. (1978), <u>Consumer Behaviour: Theory and Practice</u>, Third Edition, R. D. Irwin, Inc., Illinois.
- Wells A. and Sheth, J. (1971), "Factor Analysis in Marketing Research", <u>In: Multivariate Analysis in Marketing</u>, (ed.) D. A. Aaker, California: Wadsworth Publishing Co., Inc., p. 223.
- Williams, K. C. (1981), <u>Behavioural Aspects of Marketing</u>, Heinemann, London.
- Yavas, U., (1988), "Banking Behaviour in an Arabian Gulf Country: A Consumer Survey", <u>International Journal of Bank Marketing</u>, vol. 6, no. 5, pp. 40-48.

<u>Bibliography</u>

.

.

- Arbeit, S. and Sawyer, A. G., "Applying psychographic analysis to a retail banking market", <u>Magazine of Bank Administration</u>, vol. 50, no. 11, November 1974, pp. 39-45.
- Asher, J., "Effective marketing at work 11 case studies", <u>Banking</u>, The USA, vol. 66, August 1973, pp. 22-24, 68-70.

Asher, J., "New corporate savings accounts", <u>Banking</u>, The USA, vol. 68, May 1976.

- Ashforth, T., "Increasing competition between banks and life insurance companies: A clash of giants or a non-event?", <u>International Journal of Bank Marketing</u>, 6, 1, 1988, pp. 5-13.
- Athanasoulies, C. C., <u>Marketing of Bank Services</u>, Ph.D Thesis, University of Aston, 1976.
- Awh, R. Y. and Waters, D., "A discriminant analysis of economic, demographic and attitudinal characteristics of bank charge-card holders: A case study", <u>Journal of Finance</u>, vol. 29, June 1974, pp. 973-980.
- Badoc, M., "Banks new marketing wave", <u>European Business</u>, vol. 42, Autumn, 1974, pp. 47-57.
- Baker, M. J., "The marketing of services", <u>Scottish Bankers</u> <u>Magazine</u>, vol. LXIX, no. 273, 1977, pp. 6-18.
- Bannister, D., Brown, A. P. and Dormand, B., "The effectiveness of the major clearing banks in attracting new accounts: A case study", <u>Managerial Finance</u>, 5, 3, 1980, pp. 277-285.
- Barton, T. H., "How to make marketing work in your bank?", <u>Banking</u>, The USA, September 1965, pp. 65-72.

Bartman, P., "The corporate banking customer", European Research,

March, 1975.

- Bellenger, D. N. and Goldstucker, J. L., "Discrimination against women: An opportunity for bank marketing", <u>Journal of Bank</u> <u>Research</u>, vol. 6, part 2, Summer 1975, pp. 155-159.
- Bellenger, D. N., Robertson, D. H. and Greenberg, B. A., "Female attitudes toward the use of credit vs. cash", <u>Journal of</u> <u>Bank Research</u>, vol. 10, Spring 1979, pp. 54-57.
- Berry, L. L., "Banking, marketing and the image of concept", <u>Arizona Business Bulletin</u>, vol. 15, no. 9, November 1968, pp. 239-244.
- Berry, L. L., "Services marketing is different", <u>Business</u>, May / June, 1980, pp. 24-29.
- Berry, L. L., "How bank marketers view their jobs?", <u>The Bankers</u> <u>Magazine</u>, The USA, November 1980.
- Berkson, J., "Application of the logistic function to bio-assy", Journal of the American Statistical Association, vol. 39, September 1944, pp. 357-365.
- Bettinger, C., "Community bank strategies: The importance of magic in bank marketing", <u>Bank Marketing</u>, The USA, December 1985.
- Blake, A. J., "What's new in bank incentive promotions", <u>Banking</u>, vol. 66, March, 1974.
- Booz, A., "Tyros in the marketing game", <u>Business Week</u>, September 1973.
- Boyd, H. W., Jr., Frank, R. E., Massy, W. F. and Zoheir, M, "On the use of marketing research in the emerging economies", <u>Journal of Marketing Research</u>, vol. 1, November, 1964, pp. 20-23.
- Briggs, G., "New thrusts and directions in bank marketing", Bankers Monthly Magazine, vol. 88, part 8, August 1971,

pp. 12-19.

- Brody, R. P. and Cunningham, S. M., "Personality variables and the consumer decision process", <u>Journal of Marketing Research</u>, vol. V, February 1968, pp. 50-57.
- Buswell, D., "Measuring the quality of customer service", <u>The Market</u> <u>Research Society</u>, 1984, pp. 1-10
- Carey, T. P. A., "Strategy formulation for banks: Banks and marketing", <u>International Journal of Bank Marketing</u>, vol. 7, no. 4, 1989, pp. 8-13.
- Casey, D. C., "Marketing research and bank planning", <u>Bankers</u> <u>Monthly Magazine</u>, The USA, November, 1970, pp. 36-37, 50.
- Channon, D., "Banking through the looking Glass", <u>Cambridge Seminar</u>, 1987.
- Charlton, C. F., "You can measure advertising performance", <u>Banking</u>, vol. 64, part 3, September, 1971.
- Christopher, M., " Non-metric scaling: The principles and marketing possibilities", <u>European Research</u>, May, 1973, pp. 108-114.
- Clausen, J. A. and Ford, R. N., "Controlling bias in mail questionnaires", <u>Journal of the American Statistical</u> <u>Association</u>, vol. 42, no. 240, December 1947, pp. 497-511.
- Clawson, C. J., "Fitting branch locations, performance standards and marketing strategies to local conditions", <u>Journal of</u> <u>Marketing</u>, vol. 38, January 1974, pp. 8-14.
- Collins, G., "On Methods", <u>Journal of Advertising Research</u>, vol. 24, no. 2, Summer 1960, pp. 28-33.
- Cooley, J. L., "Banking goes to market", <u>Banking</u>, The USA, August 1965, pp. 43-45.
- Cox, D., <u>Success in Elements of Banking</u>, Third Edition, Richard Clay (The Chaucer Press) Ltd., The UK, 1986.

- Cummings, T., "A bank president looks at bank's Ads", <u>Banking</u>, The USA, October 1965, pp. 70 -71.
- Davies, A. H., The Practice of Marketing Research, 1983.
- Day, A., "Marketing in action", <u>International Journal of Bank</u> Marketing, vol. 5, no. 5, 1987, pp. 6-29.
- Day, A. and Wills, G., "Critical issues in bank marketing today: An overview", <u>International Journal of Bank Marketing</u>, 3, 4, 1985, pp. 3-8.
- Doyle, P., "Non-metric multidimensional scaling: A user's guide", <u>European Journal of Marketing</u>, vol. 7, no. 2, pp. 82-89.
- Dudley, S. C., Young, G. F. and Powers, R. L., "A study of factors affecting individual's banking performance", journal of <u>Professional Services Marketing</u>, vol. 1, nos. 1 and 2, Fall 1985 / Winter 1985-1986, pp. 163-168.
- Edwards, R. D., "The changing culture of marketing", <u>The Banker</u>, The USA, November, 1985, pp. 12-16.
- Eisenberg, R., "Changing out the financial supermarkets", <u>Money</u>, May 1983, pp. 141-147.
- Ennew, C., Watkins, T. and Wright, M., <u>Marketing Financial Services</u>, Heinemann Professional Publishing Ltd., 1990.
- Evans, R. H., "Bank selection: It all depends on the situation", Journal of Bank Research, Winter 1979, pp. 242-245.
- Farley, J. U., "Why does "brand loyalty" vary over products?", <u>Journal of Marketing Research</u>, vol. 1, November 1964, pp. 9-14.
- Fitch, T. P., "The War for the Consumer Dollar", <u>The United States</u> <u>Banker</u>, vol. 97, part 3, pp. 11-18, 71.
- Fitts, R. L. and Karson, M. J., "Canonical correlation in a market segmentation study of commercial bank services", <u>Journal of</u>

Economics and Business, vol. 29, part 2, Winter 1977, pp. 116-121.

- Flory, W. N., "Bank marketing: The state of the art", <u>Bankers'</u> <u>Monthly Magazine</u>, The USA, vol. 90, July 1973, pp. 16-18.
- Francel, E. G., "Mail-administered questionnaires: A success story", <u>Journal of Marketing Research</u>, vol. III, February 1966, pp. 89-92.
- Frank, R. E., Massy, W. F. and Morrison, D. G., "Bias in Multiple Discriminant Analysis", <u>Journal of Marketing Research</u>, vol. II, August 1965, pp. 250-258.
- Friars, E. M., "Applying retail principles in Retail banking", Journal of Retail Banking, vol. 7, part 3, 1985, pp. 43-51.
- Furlong, C. B. and Brent Ritchie, J. R., "Consumer concept testing of financial services", <u>International Journal of Bank</u> <u>Marketing</u>, 4, 1, 1986, pp. 3-17.
- Gabay, J. J., "Financial services and direct marketing: An act in good faith", <u>Direct Response</u>, 9, 6, June 1988, pp. 50-51.
- Gee, N., "Banks tap the women's market", <u>Public Relations Journal</u>, vol. 31, August 1975.
- Grasing, R. E. and Hessick, M. H., <u>Achieving Quality in Financial</u> <u>Service Organisations</u>, Robert E. Nolan Co., Inc., 1988.
- Greenway, G. S. and Southgate, P. W., "Quality of service in the Customer-banker relationship: Match or mis-match?", <u>Marketing Research Society</u>, Annual Conference, 28th Conference, 1985, pp. 83-95.
- Grossack, M., "Humanised banking new approach to new markets", <u>Banking</u>, vol. 63, part 3, September 1970, pp. 52, 103.
- Haas, R. W., "A marketing approach to bank advertising", <u>Bankers</u> <u>Monthly Magazine</u>, October 1970.

- Haas, R. W., "Bank marketing in the age of consumerism", <u>"The Bankers</u> Magazine, The USA, vol. 155, 1972, pp. 39-44.
- Haas, R. W., "Marketing the bank card to users and non-users", <u>Banking</u>, The USA, vol. 65, part 3, September, 1972, pp. 23, 26, 90.
- Haas, R. W., "The missing link in bank marketing", <u>Atlantic</u> <u>Economic Review</u>, vol. 24, January / February 1974, pp. 35-39.
- Hall, J. G., "Commercial finance: Stop selling and start marketing", <u>The Secured Lender</u>, vol. 43, no. 2, March / April 1987, pp. 38-41, 57.
- Handscombe, R., <u>Bankers' Management Handbook</u>, McGraw-Hill Book Co., 1976.
- Haugh, L. J., "Bank marketing's error: not knowing customer", <u>Advertising Age</u>, May 31, 1976, p. 10.
- Hedges, R. B., Jr., "Managing your customer relationship portfolio", <u>Commercial Lending Review</u>, vol. 1, part 2, 1986, pp. 3-20.
- Heinze, D. C., "The marketer's information search: A bayesian approach", <u>Akron Business and Economic Review</u>, vol. 5, Summer, 1974, pp. 43-45.
- Hirsch, H. D., "new approaches in bank marketing", <u>Banking</u>, The USA, October 1969, pp. 45-46.
- Howcroft, B. and Lavis, J., "Image in retail banking", <u>International</u> <u>Journal of Bank Marketing</u>, 4, 4, 1986, pp. 3-13.
- Hughes, G. D., "Predicting bank deposits and loans", <u>Journal of</u> <u>Marketing Research</u>, vol. VII, February 1970, pp. 95-100.
- Johnson, E. M., "How to measure your bank's image", <u>Banking</u>, The USA, December 1969, pp. 63-66.

Jones, G. L., "Family finance centre? It could be a service

corporation", <u>Savings and Loan News</u>, vol. 95, September 1974, pp. 68-69.

- Judd, R. C., "The case for re-defining services", <u>Journal of</u> <u>Marketing</u>, vol. 28, January 1964, pp. 58-59.
- Judd, R. C., "Similarities or differences in product and service retailing", <u>Journal of Retailing</u>, vol. 43, no. 4, Winter 1968.
- Kardashian, G. R., "Applying market research to new services", <u>Banking</u>, The USA, May 1965, pp. 46, 134.
- Kaynak, E., "Difficulties of undertaking marketing research in the developing countries", <u>European Research</u>, vol. 6, no. 6, November 1978, pp. 251-259.
- Kohn, S. J., "Bank marketing strategy for the 70's", <u>The Bankers'</u> <u>Magazine</u>, The USA, vol. 154, 1971, pp. 41-48.
- King, B., "Consumer demanding expertise and care", <u>The Financial</u> <u>Post</u>, 17 December, 1983.
- King, J. O., <u>An Analysis of the Current Marketing Activities of</u> <u>Commercial Banks: A Cross Sectional Approach Utilising</u> <u>Socio-economic, Demographic and Attitudinal Variables</u>, DBA Dissertation, Mississippi State University, The USA, 1971.
- Kirk, R. A., "Co-ordinating the marketing effort", <u>Burroghs Clearing</u> <u>House</u>, vol. 17, June 1969, pp. 26-27, 67-74.
- Kirkby, D., "Finance through factoring", <u>Managerial Finance</u>, vol. 2, no. 3, 1976, pp. 211-228.
- Koch, J. H., "For profitable marketing in smaller banks", <u>Banking</u>, The USA, vol. 63, January 1971.
- Kotler, P., "How to anticipate consumerism's coming threat to banking", <u>Banking</u>, The USA, vol. 65, January 1973.

- Kress, G. J., "Bank marketing by student researcher", <u>Bankers</u> <u>Magazine</u>, The USA, vol. 156, Summer 1973.
- Landon, L., "Sales management up-date", <u>Bank Marketing</u>, The USA, December, 1985.
- Laroche, M. and Manning, T., "Consumer brand selection and categorisation processes: A study of bank choice", <u>International Journal of Bank Marketing</u>, 2, 3, 1984, pp. 3-21.
- Laroche, M., Rosenblatt, J. A. and Manning, T., "Service used and factors considered important in selecting a bank: An investigation across divers demographic segments", <u>International Journal of Bank Marketing</u>, 4, 1, 1986, pp. 35-55.
- Lauden, F. K., "The great name change", <u>Banking</u>, vol. 66, April 1974, pp. 29-31, 111.
- Laurie, S., "More than just money", The Banker, May 1988, pp. 42-44.
- Lefkowith, E. F., "Planning communications under changing conditions", <u>Burroughs Clearing House</u>, vol. 58, March 1974, pp. 18, 62.
- Levy, S. J., "Consumer views of bank services", <u>Journal of Bank</u> <u>Research</u>, vol. 4, Summer 1973, pp. 100-104.
- Liegh, J. H. and Martin, C. R., "Don't know item non-response in a Telephone survey: Effects of question form and respondent characteristics", <u>Journal of Marketing research</u>, vol. 24, November 1987, pp. 418-424.
- Ligaya, J. F., "A computer-based planning model for the economic evaluation of proposed investments in new bank branch sites", <u>Cost and Management</u>, vol. 51, part 1, 1977, pp. 30-33.

- Lingoes, J. C., "An IBM-7090 Program for Guttman-Lingoes Smallest Space Analysis", <u>Behavioural Science</u>, vol. 10, May 1965, pp. 183-184.
- Lipis, A. H., "EFTS- some directions for the future", <u>The Magazine</u> of Bank Administration, vol. 52, June 1976, pp. 38-42.
- Littlefield, J. E., "On the art of locating new banks", <u>The Bankers</u> <u>Magazine</u>, The USA, Autumn 1968, pp. 94-102.
- Loudon, D. L., "A note on marketing research in Mexico", <u>Journal of</u> <u>Business Research</u>, vol. 4, no. 1, February 1976, pp. 69-74.
- Ma, J. C., "Accounts receivable financing", <u>Credit and Finance</u> <u>Management</u>, vol. 76, December 1974, pp. 28-31.
- MaCallum, J. S., "Marketing social security direct deposit service", <u>The Magazine of Bank Administration</u>, vol. 51, September 1975.
- Mahalanobis, P. C., "On the generalised distance in statistics", <u>Proceedings of the National Institute of Science</u>, India, vol. 2, 1936.
- Manasian, D., "Banks fight for consumers", <u>Marketing</u>, 14 May, 1980, pp. 24-26.
- Martenson, R., "Consumer choice criteria in retail bank selection", <u>International Journal of Bank Marketing</u>, 3, 2, 1985, pp. 64-74.
- Mason, J. B. and Mayer, M. L., "Analysing factors that affect banking patronage", <u>The Magazine of Bank Administration</u>, vol. 50, July 1974, pp. 42-45.
- Massy, W. F., "Discriminant analysis of audience characteristics", <u>Journal of Advertising Research</u>, vol. 5, 1965, pp. 39-48.
- Mathews, H. L. and Slocum, J. W., "Correlatives of commercial bank credit card use", <u>Journal of Bank Research</u>, vol. 2, Winter

1972, pp. 20-27.

- Mathis, M. G., "Bankers' consensus: Now accounts here to stay", Banking, vol. 67, August 1975.
- McColl, H. L., "The dramatic changes on banking's horizon", <u>The Magazine of Bank Administration</u>, vol. 49, August 1973.
- Meidan, A., "Smallest space analysis of students' attitudes on marketing education", <u>European Journal of Marketing</u>, vol. 11, no. 1, 1977.
- Miller, R. B., "Bank marketing- a symposim", <u>The Bankers' Magazine</u>, Autumn 1973, pp. 17-21.
- Miller, W. R., Jr., "The new retail banking poses a marketing challenge", <u>Banking</u> (The Journal of the American Bankers Association), April 1977, pp. 21-22.
- Monroe, J. S., "Can your bank compete in tomorrow's market?", <u>The Magazine of Bank Administration</u>, vol. 50, June 1974, pp. 17-21.
- Moutinho, L. and Brownlie, D. T., "Customer satisfaction with bank services: A multidimensional space analysis", <u>International Journal of Bank Marketing</u>, vol. 7, no. 5, 1989, pp. 23-27.
- Moran, D. E., "Improving branch office performance", <u>Management</u> <u>Focus</u>, May-June 1979, pp. 25-31.
- Moskowitz, L., "A new view of factoring credit financial management", vol. 76, January 1974, pp. 27-29.
- Myers, J. G., "Determinants of private brand attitude", <u>Journal of</u> <u>Marketing Research</u>, vol. IV, February 1967, pp. 73-81.
- Myers, R. H., "Marketing opportunities", <u>Business Horizons</u>, February 1972.

Myers, J. G. and Nicosia, F. M., "On the study of consumer

typologies", <u>Journal of Marketing Research</u>, vol. V, May 1968, pp. 182-193.

- Nadler, P. S., "Making your bank's stock more attractive", <u>Bankers</u> <u>Monthly Magazine</u>, vol. 90, November 1973.
- Neidell, L. A. and Teach, R. D., "Measuring bank images: A comparison of two approaches", <u>Journal of the Academy of</u> <u>Marketing Sciences</u>, vol. 2, Spring 1974, pp. 374 390.
- Newton, L. F., "Research challenges banking myths", <u>Banking</u>, The USA, April 1965, pp. 54-56.
- Niese, L., "Customer service packages- a status report", <u>The</u> <u>Magazine of Bank Administration</u>, vol. 50, July 1974.
- Olsen, L. M. and Lord, J. D., "Market area characteristics and branch bank performance", <u>Journal of Bank Research</u>, Summer 1979, pp. 102-110.
- Partel, F. J., Jr., "What the American consumer wants and what the responsive banking institution can do about it?", <u>Journal</u> <u>of Retail Banking</u>, vol. 6, part 1 and 2, 1984, pp. 1-7.
- Patty, G. R., "A primer on market research for small banks", <u>Banking</u>, The USA, January 1965, pp. 65-67.
- Perry, M., "Discriminant analysis of relations between consumers' attitudes, behaviour and intentions", <u>Journal of</u> <u>Advertising Research</u>, vol. 9, no. 2, 1969, pp. 34-39.
- Person, R. J., "Losing customers, keeping customers", <u>Banking</u>, The USA, October 1975, pp. 92-96.
- Plummer, J. T., "Life style patterns: a new constraint for mass communications research", <u>Journal of Broadcasting</u>, vol. XVI, no. 1, Winter 1971-1972, pp. 79-89.
- Pool, A. A., "Application of discriminant analysis in formulation of promotional strategy for Cash-Dispensing Machines",

Journal of Bank Research, vol. V, part 1, Spring 1974, pp. 13-19.

- POO1, A. A., "Attitudes toward consumer banking packages: an empirical analysis", <u>Journal of Bank Marketing</u>, vol. 7, Spring 1976, pp. 88-92.
- Pottruck, D. S., "Eight keys to improving customer loyalty and retention through direct marketing", <u>International Journal</u> of Bank Marketing, 6, 1, 1988, pp. 14-19.
- Poza, E. J. and Fuchs, C. J., "Improving customer service in banks: a case history", <u>Personnel</u>, The USA, 64 / 5, May 1987, pp. 58-61.
- Raphel, M., "Ad techniques: move customers up the loyalty ladder", <u>Bank Marketing</u>, vol. 12, part 11, 1980, pp. 37-38.
- Rayburn, L. G., "Cost standards for bank advertising", <u>Banking</u> Magazine, The USA, vol. 161, part 4, 1978, pp. 33-38.
- Redhead, H. M., "Selective marketing research: a key to retail banking", <u>Banking</u>, The USA, June 1965, pp. 47, 106.
- Reekie, W. D., "Marketing in Banking (Part 2)", <u>Bankers' Magazine</u>, The USA, August-September 1972.
- Regan, W. J., "The service revolution", <u>Journal of Marketing</u>, vol. 27, July 1963, pp. 57-62.
- Renaud, E., "As bank marketing widens its horizons", <u>Bankers Monthly</u> <u>Magazine</u>, vol. 91, September 1974.
- Reuss, H. S., "Shaping banking's future", <u>Credit and Finance</u> <u>Management</u>, vol. 77, September 1975.
- Riday, J. W., "Market research techniques", <u>Banking</u>, The USA, February 1967, pp. 75-78.
- Riggall, J., "What counts with bank customers?", <u>Banking</u>, The American Bankers Association, May 1979,

pp. 116-119.

- Riggall, J., "A new study: how newcomers select banks", <u>Banking</u>, The American Bankers Association, July 1980, pp. 92-95.
- Roberts, C. S., "A changing world requires new marketing concepts", The Commercial and Finance Chronicle, vol. 192, no. 5994.
- Robertson, T. S. and Kennedy, J. N., "Prediction of consumer innovators: application of Multiple Discriminant Analysis", vol. V, February 1968, pp. 64-69.
- Rossier, J. A., "The coming crises in bank management (part 2)", Bankers' Magazine, The USA, vol. 156, August 1973.
- Roter, R., "NatWest's fight to overcome the fear barrier", <u>Campaign</u>, 23 March, 1979, pp. 33, 36.
- Sarker, A. K. and Saleh, F. A., "The buyer of professional services: an examination of some key variables in selection process", <u>Journal of Purchasing and Materials Management</u>, vol. 10, 1974, pp. 22-32.
- Schneider, B. and Bowen, D. E., "Employee and customer perceptions of service in bank: replication and extension", <u>Journal of</u> <u>Applied Psychology</u>, vol. 70, no. 3, 1985, pp. 423-433.
- Sentell, G. D. and Philpot, J. W., "A note on evaluating the representativeness of samples taken in less developed countries", <u>International Journal of Bank Marketing</u>, vol. 1, part 1, 1984, pp. 81-84.
- Shanmugam, B., "Marketing of financial services in a developing country: the Malaysian experience", <u>International Journal</u> <u>of Bank Marketing</u>, vol. 7, no. 4, 1989, pp. 33-39.
- Shirilla, R. M., "A transaction product that focuses on broad consumer wants", <u>Bank Marketing</u>, November 1985, pp. 38-44. Stanley, T. J., Berry, L. L. and Danko, W. D., "Personal service

versus convenience perception of the high-income consumer",

The Journal of Retail Banking, June 1979, pp. 54-61.

- Stafford, D. C., <u>Bank Advertising and the Personal Saving Market</u>, Ph.D Thesis, University of Exeter, The UK, 1984.
- Szatrowski, T., "Bank marketing with a difference", <u>The Bankers</u> Magazine, vol. 158, no. 2, Spring 1975.

Thomas, K. H., "Framework for branch performance evaluation",

Journal of Retail Banking, vol. 4, part 1, 1982, pp. 24-34.

- Thompson, D. W., "Reading customer behaviour, does your sales force really know the ABCs?", <u>Bank Marketing</u>, November 1985, pp. 46-52.
- Tyler, W., "Organising for marketing", <u>Banking</u>, vol. 66, August 1973.
- Valentine, S. and Mason, S., <u>Teach Yourself Banking</u>, Hodder and Stoughton Educational, The UK, 1984.
- Volckhausen, G., "Market research is for you", <u>Burrogh Clearing</u> <u>House</u>, vol. 57, August 1973, pp. 31, 70-73.
- Wasem, G. M., "The triple dilemma of bank marketing", <u>The Bankers'</u> <u>Magazine</u>, The USA, vol. 152, 1969, pp. 65-69.
- Wasem, G. M., "Emerging trends in bank marketing", <u>Bankers' Monthly</u> <u>Magazine</u>, vol. 86, August 1969.
- Wasem, G. M., "Marketing and campaigns (part 1)", <u>Bankers Monthly</u> <u>Magazine</u>, vol. 88. December 1971.
- Wasem, G. M., "Marketing and campaigns (part 2)", <u>Bankers Monthly</u> <u>Magazine</u>, vol. 89, January 1972.
- Wasem, G. M., "The marketer and profit", <u>Bankers Monthly Magazine</u>, vol. 90, February 1973.
- Wasem, G. M., "Position description: the marketing officer", <u>Bankers Monthly Magazine</u>, vol. 90, June 1973, pp. 33-37.

- Wasem, G. M., "Government- the forgotten market segment", <u>Bankers</u> <u>Monthly Magazine</u>, vol. 90 December 1973.
- Wasem, G. M., "Bank marketing ethics", <u>Bankers Monthly Magazine</u>, vol. 91, January 1974.
- Wasem, G. M., "The banking education of student X", <u>Bankers Monthly</u> <u>Magazine</u>, vol. 91, July 1974, pp. 16-19.
- Wasem, G. M., "A marketing look ahead", <u>Bankers Monthly Magazine</u>, vol. 91 December 1974.
- Wasem, G. M., "The marketer and his city" <u>Bankers Monthly Magazine</u>, vol. 92, January 1975.
- Watne, D. A., "Cross-selling the bank customer", <u>Journal of Bank</u> Research, vol. 10, Autumn 1979, pp. 165-172.
- Wetterhuus, A., "Cashless, checkless society: on its way?", <u>Computers and Automation</u>, November 1972.
- Weyer, D. V., "The marketing of financial services", <u>Journal of</u> <u>General Management</u>, vol. 1, no. 1, 1973, pp. 16-23.
- Whitehead, G. and Hoyle, J., <u>Elements of Banking</u>, Heinemnann Professional Publishing Ltd., The UK, 1987.
- Wilson, A., "The Marketing of professional services", <u>Marketing</u> <u>Forum</u>, November-December 1972.
- Yavas, U., "Marketing Research in an Arabian Gulf Country", <u>Journal of Market Research Society</u>, vol. 29, no. 4, 1987, pp. 458-463.
- Yavas, U. and Kaynak, E., "Current status of marketing research in developing countries: problems and opportunities", <u>Journal</u> <u>of International Marketing and Marketing Research</u>, vol. 5, no. 2, 1980, pp. 79-89.
- Yavas, U. and Kaynak, E., "Key dimensions of the bank patronage decision", <u>Management Research News</u>, vol. 2, 1982, pp.22-25.

APPENDIX A

THE RESEARCH QUESTIONNAIRE

.

.

QUESTIONNAIRE

Dear respondent:

I am a Doctoral student at the University of Sheffield in the United Kingdom, and I am currently undertaking a research study into Consumers' Attitudes towards Commercial Banking Services in Traq, with particular reference to Savings Accounts.

I would be grateful if you could assist me by answering the enclosed questionnaire. Any information and responses provided will be confidential and analysed in aggregate.

The results of this study will hopefully help the Traqi banks to identify your needs and wants in order to satisfy them, and to provide better banking services for you and other Traqi bank customers in the future.

Thank you very much for your cooperation.

Sincerely,

Salim K. Hussain Ph.D research student

Part 1: Allitudes towards savings accounts.

In this part, I would like you to rate the importance of each of the following statements on the basis of their importance to you in your decision to open a savings account at a commercial bank:

.

.

i	very important	important	some importance	not important	not important at all
1. Savings help to mee increasing living expenses.	t 1	2	3	4	5
2. Savings help to beat inflation.	t 1	2	3	4	5
3. Savings help to mak sure the family is protected in case of emergencies.	te 1 f	2	3	4	5
4. Savings help to bui up a retirement fur	ld 1 nd.	2	3	4	5
5. Savings help to inc the over all standa living.	rease 1 rd of	2	3	4	5
6. Savings help in buy property for family accommodation.	ying 1	2	3	4	5
7. I feel my money is	safe. 1	2	3	4	5
8. Savings are an investment.	1	2	3	4	5
9. I can easily deposit savings at the ban	2 1 k.	2	3	4	5
10. I can easily withdo money from the ba	raw 1 ank.	2	3	4	5
11. I feel I own part the bank.	of 1	2	3	4	5

-	very important	important	some importance	not important	not important at all
12. Savings help to start a new business.	1	2	3	4	5
13. Savings help to cover marriage expenses.	1	2	. 3	4	5
14. I already use other services at this bank.	1	2	3	4	5
15. Bank pays higher interest rate on savings.	1	2	3	4	5
16. A savings account is a good method to kee surplus money.	1 p	2	3	4	5
17. Because of advice from friends.	1	2	3	4	5
18. Because of advice from family members.	1	2	3	4	5
19. Own decision.	1	2	3	4	5
20. Bank personnel are concerned about each of their customers.	1	2	3	4	5
21. Bank personnel are knowledgeable.	1	2	3	4	5
22. Bank managers listen to their customers' needs.	1	2	3	4	5
23. There is a good interaction between staff and customers.	1	2	3	4	5
24. Bank personnel are friendly.	1	2	3	4	5
25. Bank personnel give advice to customers.	1	2	3	4	5
26. Bank personnel are willing to explain the services offered.	1	2	3	4	5

	:	very important	important	some importance	not important	not important at all
27.	Bank management is always available for consultation.	1	2	3	4	5
28.	Friendliness of tellers.	1	2	3	4	5
29.	Complaints are dealL with.	1	2	3	4	5
30.	Bank personnel make me feel important as a customer.	1	2	3	4	5
31.	Length of time requir to process transaction	red 1 on.	2	3	4	5
32.	Length of waiting tin before being served.	ne 1	2	3	4	5
33.	Convenient hours of operation for savers.	1	2	3	4	5
34.	Special lines for savings accounts holders.	1	2	3	4	5
35.	Accuracy of bank transactions.	1	2	3	4	5
36.	Confidentiality of bank transactions.	1	2	3	4	5
37.	Easily of processing bank transactions.	1	2	3	4	5
38.	Bank is located near shopping centre.	1	2	3	4	5
39.	Bank is located near home.	1	2	3	4	5
40.	Bank is located near place of employment	. 1	2	3	4	5
41.	Bank has many bran locations.	ch 1	2	3	4	5

	im 	very portant	important	some importance	not important	not important at all
42.	, Bank has parking facilities for cuslomers.	1	2	3	4	5
43.	Bank is located near public transportation (bus stop, taxi stand).	1	2	3	4	5
44.	Easy entrance to and exit from bank buildin	1 g.	2	3	4	5
45.	Bank building has modern design.	1	2	3	4	5
16.	Availability of wailing areas inside the bank for cuslomers.	1	2	3	4	5
47.	Bank has large building.	1	2	3	4	5
48.	Bank is well-known.	1	2	3	4	5
49.	Bank has been well- established for many years.	1	2	3	4	5
50 .	Bank has a good reputation.	1	2	3	4	5
51.	Bank has many branch	es. 1	2	3	4	5
52.	Bank advertises on Television.	1	2	3	4	5
53.	Bank advertises on Radio.	1	2	3	4	5

		very important	important	some importance 	not important 	not important at all
54. Ba	ank advertises in newspapers.	1	2	3	4	5
55. Ва л	ank advertises in nagazines.	1	2	3	4	5
56. Pe b	ersonal contact by ank personnel.	1	2	3	4	5
57.Ba	ank has full-range f bank services.	1	2	3	4	5
58. Ba si b	ank provides regula tatements of saving alance for customer	r 1 s s.	2	3	4	5
59. Ba se (i of	ank has competitive ervice charges interest on loans, ther charges).	1	2	3	4	5
60. Ba P sa d ir 10	ank provides different romotions for avers (free prize raws, free life nsurance up to 000 I.D., etc.).	ent 1	2	3	4	5

.

Part 2: Psychographic characteristics regarding banking and

financial matters.

i.

In this part, I would like you to indicate your degree of satisfaction with each of the following statements, by circle the number that best describes your feelings about that statement:

		strongly agræ	agree	neither agree nor disagree	disagree	strongly disagree
1 1	am a good saver.	1	2	3	4	5
1	l like to pay cash for everything I buy.	1	2	3	4	5
3. 1	To buy anything, other than a house on cred is unwise.	· 1 it	2	3	4	5
4. <i>7</i>	A person can save a lo of money by looking fo the lowest possible prices when shopping.	t 1 or	2	3	4	5
5. I	t is good to have savings account, becau il is useful.	1 Ise	2	3	4	5

Part 3: Profile data on respondents:

The questions in this part deals with your personal profile data. Please read each question carefully and place an "x" in the box that best describes your situation:

1. Do you presently have a savings account at a commercial bank ?

	Yes	No
2. Are you	Male	¦ Female
3. Are you	ll Single	II Married

4. Please indicate your age group:

 1
 24 and under
 1
 25-44

 1
 1
 45-64
 1
 1
 65 and over

5. Please indicate your family size group:

Single

Couple without children

Couple with three children and under

Couple with four children and over

6. Please indicate your household yearly income range. If you are the only member of your household, please indicate your income range:

Under 2000 I.D. 2000-2999 I.D.

7. Please indicate the best description of your household head's educational level:

Less than Secondary School

Secondary School graduate

University or College graduate (B.Sc., B.A.)

Higher education graduate (Ph.D., M.Sc., M.A.)

- 8. Please indicate the best description of your household head's occupation:
 - General manager and higher positions
 - ____ Manager
 - Teacher (at School or University)
 - Clerk (Office-worker)

Labourer

- Pensioner
- : Owner of private business
- Farmer
- Housewife
- Student

Thank you very much for your cooperation.

استمارة استطلاع رأي

أخي المواطن

اني طالب دكتوراه في جامعة شفيلد في المملكة المتحدة، واني حالباً اجري دراسة الـدكتـوراه المـوسـومـة «مـواقف المستهلكين من الخـدمات المصرفية التجارية مع اشارة خاصة لحسابات التوفير». واني ساكـون شاكـراً لمسـاعـدتكم الطيبة لي من خلال اجابتكم على الاسئلة الواردة في استمارة استطلاع الرأي هذه وتأكدوا بأن كل معلومة اواجابة موف تعطونها ستكون لغرض الدراسة فقط وليس مطلوب منكم اعطاء الاسم ار العنوان.

انه لمن المؤمل ان تساهم نتائج هذه الدراسة في تطوير الخدمات المصرفية في العراق من خلال مساعدة مؤسساتنا المصرفية في معرفة وتحديد متطلب اتكم ورغب أتكم وماهي السبل المتوفرة للمصارف لغرض تحقيق ذلك لتوفير خدمات مصرفية افضل لك ولكل مواطن .

واخيراً شكراً جزيلًا لتعاونكم . . .

المخلص الطالب سالم كريم ح طالب دكتوراه



.

مهم جداً=١ مهم=٢ بعض الاهمية=٣ غيرمهم=٤ غيرمهم علىالاطلاق=٥

الاسئلة

غيرمهم على الاطلان	غير مهم	. بعض الاهمية	مهم	مھم جدآ
0	٤	٣	۲	1
0 0	٤ ٤	.٣ ٣	r r	1
0	٤	٣	٢	١
0	٤	٣	۲	١

الاجابات

الاسئلة

	مه	مهم	بعض	غير	غيرمهم
	جد		الاهمية	مهم	على الاطلان
بساعدعلى شراء بيت للعائلة .	1	۲	٣	٤	0
بان المبالغ المودعة في	1	۲	٣	٤	٥
حساب التوفير مضمونة .					
ن التوفيركنوع من الاستثمار.	1	۲	٣	٤	٥
ايداع مبالغ في الحساب	1	۲	٣	٤	٥
فمتوح في المصرف .					
ة سحب مبالغ من الحساب	1	۲	۳	٤	٥
لمتوح في المصرف ."					
بأني املك جزء من المصرف	N	۲	٣	٤	٥
ن اموالي فيه .					
ريساعد في بدء اعمال	1	۲	٣	٤	٥
بة جديدة .					
ريساعد في تغطية نفقات	1	۲	٣	٤	٥
ياج .					
مل خدمات اخرى في المصرف	N 1	۲	٣	٤	٥
ضافة لحساب التوفير .					
رف يدفع فائدة عالية على	N	۲	٣	٤	٥
بالغ المودعة في حساب التوفير.					
مساب توفير طريقة جيدة لحفظ	1	Y	٣	٤	٥
ئض من الدخل .					
ءة الاصدقاء مهمة في قرار فتح	1	۲	٣	٤	٥
باب توفير .					
عة افراد العائلة مهمة في قرار	1	۲	٣	٤	٥
حساب توفير .					
الشخصي مهم في قرار فتح	N I	٢	٣	٤	٥
اب توفير.	I				

ات	٦	'حا	יע	
		•	_	

الاسئلة

	مهم	مهم	بعض	غير	غيرمهم
	جداً		الاهمية	مهم	على الاطلاق
٢٠ موظفي المصرف يهتمون بكل زبون	1	۲	٣	٤	0
من زبائنهم .					
٢٢ موظفي المصرف متفهمين في	١	۲	٣	٤	٥
مجال عملهم .					
٢١ ادارة المصرف تصغي لرغبات	1	۲	٣	٤	٥
كل الزبائن.					
٢٢_ هناك علاقة ودية مُتبادلة بين	1	۲	٣	٤	٥
الموظفين في المصرف والزبائن .					
٢٤ موظفي المصرف طيبين .	N	۲	٣	٤	0
٢٠_موظفى المصرف يقدمون النصائح	N	۲	٣	٤	٥
للزبائن عند الحاجة .	-				
٢٠ موظفي المصرف لهم رغبة في	<u>,</u>)	۲	٣	٤	0
شرح الحُكمات للزبائين.					
۲۱_ مسؤولي المصرف حاضرين دائماً	١	۲	٣	٤	٥
لاجابة استفسارات الزبائن .					
/٢_ امناء الصندوق يتعاملون	١	۲	٣	٤	٥
بلطف.					
۲۰ شکاوی الزبائن یستجاب لھ	١	۲	٣	٤	٥
٣٠ موظفي المصرف يجعلوني اشعر	١	۲	٣	٤	0 .
كزبون مهم للمصرف .			•		
٣٠_ طول الفترة الزمنية لقضاء	١	۲	٣	٤	٥
المعاملة.					
٣٠_ طول الفترة الزمنية للانتظار	١	۲	٣	٤	٥
قبل ان تؤدي لي الخدمة .					
۳۲_ اوقات دوام المصرف ومدى	۱.	۲	٣	٤	٥
مناسبتها للزبائن .					

الاسئلة

الاجابات

	مهم جداً	مهم	بعض الاهمية	غير مهم	غير مهم على الاطلان
٢_ وجود خطوط سريعة خاصة لتمشية	\	<u>۲</u>	٣	٤	0
معاملات اصحاب حسابات التوفير .		·		:	
٢_ دقة المعاملات المصرفية .	>	۲	٣	٤	٥
٢_ موثوقية المعاملات المصرفية .	۱ I	۲	٣	٤	٥
٢- انسيابية المعاملات المصرفية .	1	۲	٣	٤	٥
٢_ قرب المصرف من مراكز التسوق .	1	۲	٣	٤	٥
٢_ قرب المصرف من السكن .	N	Ŷ	٣	٤	٥
٤_ قرب المصرف من مكان العمل .	١.	۲	٣	٤	٥
s_ المصرف له عدد من الفروع مواقعها	1	۲	٣	٤	٥
قريبة .					
s_ المصرف لديه موقف سيارات خاص)	۲	٣	٤	٥
للزبائن .					
s_ قرب المصرف من المواصلات العامة	۱ I	۲	٣	٤	٥
ناطق وقوف حافلات نقل الركاب الخ) .					
ا_سهولة الدخول الي والخروج من	١	۲	٣	٤	٥
بناية المصرف .					
ا بناية المصرف حايثة وجميلة .)	۲	٣	٤	٥
ا ـ توفر مكانات مريحة للانتظار داخل	١	۲	٣	٤	٥
المصرف.					
- بناية المصرف كبيرة .	1	۲	٣	٤	٥
: ـ المصرف معروف للجميع (مشهور) .	١	۲	٣	٤	٥
ا المصرف تأسس منذ سنوات طويلة	1	۲	٣	٤	٥
(قديم وچيد ومعروف) .					
ا المصرف ذي سمعة طيبة .	١	۲	٣	٤	٥
ا المصرف لديه عدد كبير من الفروع .	۱ ۱	۲	٣	٤	٥

ت	با	جا	Y	۱
	-	•		

	مھم جدا	مهم	بعض الاهمية	غير مهم	غيرمهم على الاطلان
۰۲ ۵ ـ الاعلان عن خدمات المصرف على شاشة التلفنيدن	١	۲	٣	٤	٥
التشريون . ٥٢_ الاعلان عن خدمات المصرف عبر محطات الاذاعة	١	۲	٣	٤	٥
٤ ٥- الاعلان عن خدمات المصرف في الصحف المدمية	١	۲	٣	٤	0
٥٥ - الاعلان عن خدمات المصرف في المحلات	Ň	۲	٣	٤	٥
٥٦ اتصال مسؤولي المصرف بالزبائن ١١ مار مارت المصرف مرز بائن	١	٢	٣	٤	٥
ممرويج فلحنامات المصروية. ٥٧- المصرف يقوم بتقديم كافة انواع الخدمات الرم ف قرالن الز	١	۲	٣	٤	٥
الحديثات المصرفية للربان . ٥٨- المصرف يقوم بأرسال شهادة رصيد المددي	١	۲	٣	٤	٥
المودعين شهريا للربائن . ٥٩- المصرف يفرض فائدة واجور قليلة ما المعادلات المدينة تالما تعانية	١	۲	٣	٤.	٥
على المعاملات المصرفية المحتلفة . ٦٠- المصرف يقدم محفزات مختلفة	١	۲	٣	٤	0
للمودعين (سحبة ياتصيب، تأمين مجاني على الحياة الخ) .					

الاسئلة

. _

موافق بقوة = ١ موافق = ٢ لا موافق ولا غير موافق=٣ غير موافق = ٤ غير موافق بقوة = ٥ الاسنلة

غيرموافق	غيرموافق	موافق ولاغير	موافق= لا	ا موافق بقوة	
بقوة =ه	<u>٤=</u>	وافق= ٣	<u>۲</u>	1= .	
٥	٤	٣	۲	Υ	۱ ـ اني مدخر جيد .
٥	٤	٣	۲	1	۲_اني ادفع نقداً لکل شيء
					اشتريه .
٥	٤	٣	۲	1	٣_ ان الاقتراض لشراء اي شيء عدا
					بيت للسكن غير عقلاني .
٥	٤	٣	۲	N	٤_ الشخص يمكن ان يوفر مبالغ من
					خلال البحث عن اوطأ الاسعار لشراء
					حاجياته.
٥	٤	٣	۲	1	٥_ امتلاك حساب توفير في المصرف
					فكرة نافعة وجيدة .
				I	•



APPENDIX B

The D-Values of the Kolmogorov-Smirnov Goodness-of-Fit Test

Variable	Category	Fo (x)	- Sn (x)	= D-value	
٧1	1 2 3 4 5	.200 .400 .600 .800 1.000	.430 .750 .927 .957 1.000	.230 <u>.350</u> .327 .157 .000	
v2 [·]	1 2 3 4 5	.200 .400 .600 .800 1.000	.298 .618 .875 .947 1.000	.098 .218 <u>.275</u> .147 .000	
٧3	1 2 3 4 5	.200 .400 .600 .800 1.000	.732 .909 .984 .992 1.000	<u>.532</u> .509 .384 .192 .000	
v4 .	1 2 3 4 5	.200 .400 .600 .800 1.000	.468 .728 .939 .977 1.000	.268 .328 <u>.339</u> .177 .000	
v5	1 2 3 4 5	.200 .400 .600 .800 1.000	.464 .664 .868 .943 1.000	.264 .264 .268 .143 .000	
v 6	1 2 3 4 5	.200 .400 .600 .800 1.000	.404 .634 .883 .943 1.000	.204 .234 .283 .143 .000	
٧7	1 2 3 4 5	.200 .400 .600 .800 1.000	.608 .876 .967 .990 1.000	.408 .476 .367 .190 .000	
Variable	Category	Fo (x)	- Sn (x) =	= D-value	
-------------	----------	--------	------------	-------------	--
Q	1	200	272	n72	************
¥0	2	.200	• 2 7 2	202	
	2	.400	.000	200	
	4	800	940	140	
	т 5	1 000	1 000	.140	
		1.000	1.000	+000	
v9	1	.200	.366	.166	
	Z	.400	.687	.287	
	3	.600	.801	.201	
	4	.800	.980	.180	
	C	1.000	1.000	.000	
v10	1	.200	.400	.200	
	2	.400	.725	<u>.325</u>	
	3	.600	.887	.287	
	4	.800	.955	.155	
[5	1.000	1.000	.000	
v11	1	.200	.174	.026	
	2	.400	.359	.041	
	3	.600	.560	.040	
	4	.800	.775	.025	
	5	1.000	1.000	.000	
v12	1	.200	.434	.234	
	2	.400	.679	.279	
	3	.600	.921	<u>.321</u>	
	4	.800	.985	.185	
	5	1.000	1.000	.000	
v13	1	.200	.445	.245	
	2	.400	.728	.328	
	3	.600	.879	.279	
	4	.800	.947	.147	
	5	1.000	1.000	.000	
v14	1	.200	.140	.060	(19, 19, 29, 29, 29, 29, 29, 29, 29, 29, 29, 2
	2	.400	.351	.049	
	3	.600	.596	.004	
•	4	.800	.800	.000	
	5	1.000	1.000	.000	
v15	1	. 200	. 257	.057	
	2	. 400	.423	.023	
	3	.600	.695	.095	
: ; ;	4	.800	.888	.088	
	5	1.000	1.000	.000	
;	1 -	1			

Variable	Category	Fo (x)	- Sn (x)	= D-value	
v16	1 2 3 4 5	.200 .400 .600 .800 1.000	.507 .786 .922 .979 1.000	.307 <u>.386</u> .322 .179 .000	
v17	1 2 3 4 5	.200 .400 .600 .800 1.000	.260 .449 .676 .857 1.000	.060 .049 <u>.076</u> .057 .000	(
v18	1 2 3 4 5	.200 .400 .600 .800 1.000	.291 .616 .831 .937 1.000	.091 .216 <u>.231</u> .137 .000	
v19	1 2 3 4 5	.200 .400 .600 .800 1.000	.709 .898 .962 .988 1.000	<u>.509</u> .498 .362 .188 .000	
v20	1 2 3 4 5	.200 .400 .600 .800 1.000	.313 .528 .830 .917 1.000	.113 .128 <u>.230</u> .117 .000	
v21	1 2 3 4 5	.200 .400 .600 .800 1.000	.355 .680 .937 .975 1.000	.155 .280 <u>.337</u> .175 .000	
v22	1 2 3 4 5	.200 .400 .600 .800 1.000	.302 .532 .860 .947 1.000	.102 .132 <u>.260</u> .147 .000	
v23	1 2 3 4 5	.200 .400 .600 .800 1.000	.253 .502 .808 .944 1.000	.053 .102 <u>.208</u> .144 .000	

Variable	Category	Fo (x) -	Sn (x)	= D-value	
¥24	1 2 3	.200 .400	.283	.083 .162 .253	
	4 5	.800	.959	.159	
v25	1	.200	.366	.166	****
	2 3 4	.400 .600	.623 .880	.223 .280 152	
	5	1.000	1.000	.000	
v26	1 2	.200 .400	.309 .573	.109 .173	
	3 4	.600	.815	<u>.215</u> .113	
v97	5 1	200	. 272	.000	., . ,
	23	.400	.604	.204	
	4 5	.800 1.000	.932 1.000	.132 .000	
v28	1 2	.200	.362	.162	
	3 4	.600 .800	.876	<u>.276</u> .148	
v79	5	. 200	.340	.140	
, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	23	.400	.634	<u>.234</u> .219	
	4 5	.800 1.000	.917 1.000	.117 .000	
v30	1 2	.200 .400	.279 .524	.079 .124	1999-999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 19
	34	.600	.807	<u>.207</u> .116	
v31	5	1.000	.302	.000	
	23	.400	.544	.144 .189	
	4 5	.800 1.000	.876 1.000	.076 .000	

•

į

I

Variable	Category	Fo (x) -	Sn (x)	= D-value	
	1	200	2 i 2	110	
VSZ		.200	• 212	•113	
		.400	• 588 014	.188	
	3	.600	.814	<u>•214</u>	
	4	.800	.886	.086	
	5	1.000	1.000	.000	
v33	1	.200	.472	.272	
	2	.400	.778	.378	
	3	.600	.940	.340	
	4	.800	.974	.174	
	5	1.000	1.000	.000	
v34	1	.200	.506	.306	
	2	.400	.721	. 321	
	3	.600	.876	.276	
	4	. 800	948	.148	
	5	1 000	1 000	000	
		1.000	1.000	• 000	
v35	1	.200	.611	.411	
	2	.400	.886	.486	
	3	.600	.977	.377	
	4	.800	.994	.194	
	5	1.000	1.000	.000	
w36	1	200	528	378	
430	2	.200	953	453	
	2	600 600	.033	374	
		.000	• 27 4	• J / 4 107	
	4	.000	.997	•197	
	5	1.000	1.000	.000	
v37	1	.200	.438	.238	
	2	.400	.819	.419	
	3	.600	.947	.347	
	4	.800	.989	.189	
	5	1.000	1.000	.000	
v 38		200	473	223	
7 5 6	2	400	721	221	
	4	.400	.002	202	
	A	\$000 \$00	055	155	
	7 5	1 000	1 000	000	
		1.000	T+000	• 000	
v39	1	.200	.362	.162	
:	2	.400	.690	.290	
:	3	.600	.860	.260	
	4	.800	.962	.162	
	5	1.000	1.000	.000	

Variable	Category	Fo (x) -	- Sn (x)	= D-value	
v40	1	.200	.415	.215	
	2	.400	.694	. 294	
	3	.600	.890	.290	
	4	800	981	. 181	
	5	1 000	1 000	000	
	. J	1.000	1.000	.000	
v41	1	.200	.377	.177	
	2	.400	.736	<u>.336</u>	
	. 3 '	.600	.920	.320	
	t 4 1	.800	.984	.184	
	5	1.000	1.000	.000	
v 47	· 1 :	.200	. 419	.219	-
		400	.728	. 328	
	·	600	856	256	
	Å	2000	016	116	
	· E :	1 000	1 000	.110	
	с 	1.000	1.000		
v43	1	.200	.453	.253	
	2	.400	.774	.374	
	3	.600	.910	.310	
	· 4 *	.800	.970	.170	
	; 5	1.000	1.000	.000	
v 4 4	: 1	. 200	.513	. 313	Lenn of t
•••	2	400	830	430	
	3	600	243	343	
	а. А.	2000	•273 088	199	
	4 E	1 000	. 700	.100	
	- C	1.000	1.000	-000	÷
v45	1	.200	.362	.162	
	2	.400	.668	.268	
	3	.600	.860	.260	
	. 4	.800	.954	.154	
	; 5	1.000	1.000	.000	
vAA	- <u> </u>	200	517		
UT V	2	•200 #nn	2017 2017	40A	
	2	.400 KNN	•004 651	251	
		.000 600	1,521 205	.JJI 105	
		- 000	. 900	.100	
		T.000	T.000	. UUU	
v47	· 1 :	.200	.336	.136	
	2	.400	.589	.189	
	3	.600	.831	.231	
	4	.800	.944	.144	
	5	1.000	1.000	.000	
	. D	T.000	1.000	- UUU	

Variable	Category	Fo (x)	- Sn (x)	= D-value	
v 48	1 2 3 4	.200 .400 .600 .800	.362 .682 .883 .936	.162 .282 <u>.283</u> .136	
-	5	1.000	1.000	.000	
v49	1 2 3 4 5	.200 .400 .600 .800 1.000	.370 .649 .845 .951 1.000	.170 <u>.249</u> .245 .151 .000	
v50	1 2 3 4 5	.200 .400 .600 .800 1.000	.626 .852 .950 .976 1.000	.426 .452 .350 .176 .000	
v51	1 2 3 4 5	.200 .400 .600 .800 1.000	.415 .751 .817 .874 1.000	.215 <u>.351</u> .217 .074 .000	
v52	1 2 3 4 5	.200 .400 .600 .800 1.000	.366 .619 .830 .913 1.000	.166 .219 <u>.230</u> .113 .000	
v53	1 2 3 4 5	.200 .400 .600 .800 1.000	.215 .547 .762 .905 1.000	.015 .147 <u>.162</u> .105 .000	
v54	1 2 3 4 5	.200 .400 .600 .800 1.000	.264 .558 .807 .909 1.000	.064 .158 <u>.207</u> .109 .000	
v55	1 2 3 4 5	.200 .400 .600 .800 1.000	.223 .532 .804 .917 1.000	.023 .132 <u>.204</u> .117 .000	

Variable	Category	Fo (x) -	Sn (x)	= D-value	
v56	1	.200	.192	.008	
:	2	.400	.494	• • 094	
•	3	.600	.781	<u>.181</u>	
1	4	.800	.936	.136	
f	5	1.000	1.000	.000	
v57	1	.200	.370	.170	
	2	.400	.789	<u>.389</u>	
	3	.600	.955	.355	
	4	.800	.985	.185	
1	5	1.000	1.000	.000	
v58	1	.200	.366	.166	
ı	2	.400	.664	.264	-
•	3	.600	.853	.253	
	4	.800	.913	.113	
:	5	1.000	1.000	.000	
v59	1	.200	.279	.079	
	2	.400	.604	.204	
•	3	.600	.823	.223	
Ŧ	4	.800	.944	.144	
	5	1.000	1.000	.000	
v60	1	.200	.479	.279	
•	2	.400	.792	<u>.392</u>	
k .	3	.600	.924	.324	
•	4	.800	.954	.154	
	5	1.000	1.000	.000	

APPENDIX (C)

.

.

THE COMPLETE ROTATED FACTOR MATRIX

.

FACTOR		• 0 5 1 9 9	12051	966/D	160400	-04890-	01809	01587	60020 ·	42601 v.	20210-	10000	.08090	38504	-00925	01921	-02573	.03040	-02086	1/100.	-01585	• • • • • • • • • • • • • • • • • • •	.19951	.10357	.08472	00913	03960.	P7660.	:05716	• 1 5 4 9 0	•1L398	6/678°	20061.	5 0 B 7 0 K	• 11850	01294	696E0.	08410	.22116	.09626	10216	2/960.	008A0	- 04175	÷-00311	17326	.25332	.01514	.04691	10401 • ·	11611.	272724	.65131		.68910
FACTOR 7		-07085	5/620 - ·		16171	11129	- 27642	02122.	01397	41200.	11200.	11906	12711	00585	.12251	.19002	.06696	.13270	•03567	.06505	+6241•	0124U-	13016	03555	00977	•09416	01610-	.23061	•04699	1-03054		10030	02240	• 02 24 3	. 14737	•02040	•07366	.16499	.15386	.52679	04640*	54162	EEVIV.	.09960	•01432	9,660-	-:01766	1660;	•10992	07690*	70700.	7072°	.16068	001490	.10209
FACTOR 6		622ED*	55650.	04440	.10170	05784	12980	00736	64090. 57151.	69650.	.00722	.12268	• 02453	01766	10085	07335	.08804	-10372	60611.	10761.	.02226	09480	03227	.06725	•14739	.15863	.15983	.20108	-09754	06580	60606. 1989e.	10015- CEVEC-	. 79199	71861.	.12400	•05306	.05061	• 02 5 8 3	• • • • • •	E0600	11956	06413	- 01805	• 06283	. 00076	.22235	.0300.	•03084	EB/20.		26160	EU21E.	.05958	01860	.15006
FACTOR 5		16246	.05063	06211.	.10230	-04357	12858	610CO.	11615	01154	19020	00313	-02184	11388	.01019	00000	•00955	02877		3260	•02145	00,00.	.01062	•00305	-04673	04273	02204	09335	2/640*-	12060.	9E910-	05611	-10248	.15060	16169.	•85739	.01601	69169.	4707C• ·	22064.	12223.	06713	-00327	.06837	.05500	-•00597	.12509	50800			00000°	00068	•05653	02220	•04540•
FAC 10R 4	00250°-	.10693	.05745	10011.	·11405	101.00		nr 271.	.105.15	5,00	.05070	00010.	11920	.02251	.07438	12051.	16541.		200101	104.02	.08120	90661.	• 000 2 8	00727	02346	00172	20100	01080.	66000 -	24240	14548	03026	16030	.17154	.10310	.03278	26260.	5010. 57190-	31.710	.15072	. 1 2 6 1 7	19110	· 5 8 0 7 9	. 76451	. 17620	11100.	1704.	220010		.19674	.16623	11961	112 40.	E1 0E 1 •	-tu/0
FACTOR 1	. 7 4 4 59	.64764	- 4 7 7 8 7	16165.	. 73220	26276.		.13965	.17172	.10062	.43013	• 15362	•1221•	96(62.	(6/61•	22051.	06591.	13500	00150	.07678	.06359	•05234	02357	02120	•10156 03232	34460.	17530	19550-		-05441	• 0 2 1 7 3	.04570	.09029	12240.	.07632	2720U-	66600°-	.17903	•04152	86460.	-03106	03022	.15070	00000.	11/201	10/21-	51000	16590.	.02945	10120.	°01554	.00537	01355	401EO.	61000.
r ACIDR 2	07755	03150	-10670	20951.	CC/CD-	19/30-	.16700	- COB-7-2	.07247	06141.	01446	19561.		0/000.	5551.	01779	01062	.03356	05211	.12523	01240°	10000	15540.			5 1 1 1 0 0 ° 1	.09156	91010.	06235	-00640	01056	-10179	0(690.	66/10	- 004 90	14220	11091	.13705	.07508	.00936	•15250	.10530	- 27572	11752	201392	20272-	.07511	- 20574	.92096	.87495	• 4 9059	166331	02 17 1 .	2/690.	- 1 1 6 4 5
FALTOR 1	646101	.06734		1 1 2 1 2 1	05971	11001.	55600.	11:01 2 .	167 56.		92.610.	60102.	.25562	05151.	.16420	.17229	10000.	.67702	. 20785	01221.0	. 744.35	50057 -	000111	221102	. 76527	. 26024	. 60707	.27660	.21870	• 28565	- 50260	51/66.	51/64.	56103. 063301	11620	04527	.07536	13662.	:02773	61200.	46601.	.10269	201100 -	.06760	1 60 50 .		-01682	66760.	-06790	21/20.	13572	.06170	61(33. All(0.	00100	
:	5	v r > >		5 2	۲6 ۲	۲۷	02	5 × 5				414	V15	V16	V 1 7	017	۲۱۶	V 20	127	227	202	222	V 2 6	V 2 7	V 2 8	V 2 9	0 E A	167	2 E A	667	547		~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	067	V 37	0% ۸	۲،۱	2 4 2 	£ % A		242	0 V X	0 5 7	V 4.9	V 5 0	۲۶۱	V52	V5.)	V54	202	, av	222	V 59	017	

אוואורה ראכווית אאוגואי

.

	FACTOR 9	FACTOR 10	FACTOR 11	FACTOR 12	PACTOR 13
71	04645	0577	14100-	-10230	01921
~ ~ ~	26117	.16633	10175	1000	.09698
	.22521	11805	-20688	. 05565	10/10
۷5	•07544	.10318	00 C / D .		10082
۲6 ۱	193847	• 1 4 6 0 3 0 1 8 1 4	03800	*****	02027
~~~	17481	01010°	.29814	.27530	.03416
8 0 2	.12229	.13629	.12017	• 74248	•07226
	-11046	•09825	-06624	.68137	
117	02903	•16668	•64993	.18596	91052
V12	.45263	.20603	.34063	.12619	
EIV	.23163	.37423	• 05033	• 24118	
V14	.05368	-05494	.73240	-*02393	C7/70*-
	.15344	.00343	.40211	.15696	CO+/ T *
V16	.61138	•12565	.23728	•07880	14680.
V17	•02658	• 79098	•17033	•11368	
V18	.22561	.78183	•07936	06190*	
V19	26767 <b>.</b>	•05977	19122	9674T.	
V20	.05633	.00448	•24368	.15502	
V 2 1	.20314	.07317	.16981	•11340	14140.
V22	15031.	.11250	•06242	8567D.	CC770
627	00119	.05747	02384	• 10322	- 00513
V 2 4	04358	03713	•04775	\$1151. 51151	
V 25	.14066	.10294	15623	• 1 4 5 4 B	
V 2 6	.05210	•13778	02891	04105	
1 C C Z	.07861	•07376	05833	.01612	+0770*
124	01674	- 05996	•04690	.07353	7/060*
877		28550	•07309	.00921	.15574
677			82112-	.01584	.11094
V30	167/0"-		86660	• 02337	.78528
V31	00220°-	40460 -	47 5 6 1	.05465	.77579
V32	4/670*		18501	07284	.18666
567	40CTO.	+0140*L	10193	01996	.28310
V34			100120	.02058	.01340
V 35	c6100	6,000.	05410	-0508	•09032
V36	•08389	44470°		-11463	•07559
V37	•04490	04342		18610	18222
9EV	0861.9	•13452	10110-	. 04715	02306
439	•01288	04809	+0+10		-06111
V40	°03095	•00+6+00	/4680.		10666
141	.12167	•05492	1/260*		12021
V42	10096	• 0 5 3 5 0	•07660		16078
C43	01102	.14865	11631	6201 <b>1</b> .	0000
V44	.11353	02641	•04153	00460	08446
V 4 5	.25760	.08271			.06352
V46	.15178	•16279 •1521	C+400	15458	00008
747	e12/0-	10011 -	00500	04711	-•06547
V48	90170	10700 -	64690-	.19334	-:02522
V49	02420*		97610-	06624	- 64640
V50	90107	13000°	05907	.15369	04444
157	- 03554	-17754	-01394	•01334	-+00083
267		.05958	0000	01357	02768
502	77260	19040.	.04103	.03852	04151
+ C >	005300	-01660	.01219	.04415	•01199
CCA	- 07880		08923	•04405	°04703
007		00848	01977	.12112	.07481
	15964	•17909	01363	• 00204	44141 <b>.</b>
252	.03112	-•17507	•08680	.08607	000TA*
V60	.10469	.06558	.04692	09716	n+nn)•

(continued)	
1.1	
xhibit :	

The research was qualitative and based only on group discussions. There was no prepared list of questions before and during the discussions. No statistical analysis was carried out.	Using the same sample and the same list of questions over two periods has some disadvantages such as: time consuming, costly and can be effected by other social and economical changes.	The original sample contained all age groups therefore only 11% rate of response was achieved. No advanced statistical methods of data analysis were used.	The overall sample was small (250 respondents) in particular the sub- sample of frequent users (only 58 respondents. The findings did not mention the interaction between the demographic variables and the dependent variable, i.e., the usage rate.
Children and parents attitudes towards saving and savings accounts were positive and very encouraging. Thcentives provided by the bank to children were warmly welcomed by both parents and children.	With regard to product evaluation the study concluded that there were changes in favour of banks over Skils between 1980 to 1983. With regard to institution evaluation again there were changes in favour of banks over Skils between 1980 to 1983. The overall conclusion was that consumers' preference for banks over Skils remained unchanged.	Older consumers were reluctant to use the modern banking technology and their were conservative towards bank innovations. Also, they preferred banking services provided to them in their local area.	There were significant differences between frequent and infrequent users in terms of the importance they attached to some independent variables related to the location of the bank and the services it provides.
Qualitative research	Fercentages & Chi-square test	Simple ranking of importance according to percentages	The T-Test
Examining the rol- and contribution of market research had played in the development and launch of savings accounts for children.	Investigating the competition between banks and gkLIS for savings and the criteria for consumers to select between them between 1980 to 1983.	Providing some insights into the banking behaviour of consumers in terms of their age focusing on older consumers.	Extending the application of usage rate of a service as a method for segmenting the market (frequent vs. infrequent users).
acudge E Green (1986) The UK	Langrehr (1986) The USA	Alessandra, Knierim & Yavas (1986) The USA	Kynak k Yavas (1986) Canada

Exhibit 3.1 (continued)

RINTATIONS	The study lacks using statistical methods of data analysis to provide statistically tested results.	Little information was given about the life style of non-savers and infrequent savers. The sample consisted of males, i.e., females were excluded from the analysis. The study lacks more reliable statistical methods of analysis.	The response rate was very low 31% of the original sample. Factor congruencies were calculated manually due to lack of computer programme for calculating them at that time.
MAIN FINDINOS	Convenience of location was the prime reason for maintaining a checking account with a bank. Convenience of location and having a checking account at the bank were the prime reason for maintaining a savings account at the bank. Favourable interest on savings was the prime reason for maintaining a savings account at S&LIS and Credit Unions.	The life style of frequent savers is different in comparison with non-savers and infrequent savers. The importance of life style analysis in providing better understanding for bank consumers' behaviour was underlined.	The findings indicated that seven common factors influencing the choice of banks and S&L institutions these are: dependability of institution, convenience, variety of services, accessibility, convenience of non- person banking, availability of loans and interest competitive.
МВТНОРОГОЭУ	Percentage a	Percentages	Factor Analysis & Factor Congruency
FOCUS OF STUDY	Determining consumers' reasons for maintaining checking accounts with banks and savings accounts with banks, S&LIS and Credit Unions.	Investigating the life style of three groups of consumers: non-savers infrequent savers frequent savers.	<pre>Kxamining a number of factors which are important in the choice of a financial institution (banks &amp; S&amp;LIS).</pre>
RESEARCHER	Mason & Mayer (1974) The USA	Reymolds & Wells (1978) The USA	Arora, Cavusgil & Nevin (1985) The USA

Exhibit 3.1 (continued)

Exhibit 3.1: Rummary of Previous Research Publications Related to Consumers' Saving Behaviour.

RESEARCHER	FOCUS OF STUDY	METHODOLOGY	SENIGNIA NIVW	LIMITATIONS
Claycamp (1965) The USA	Identifying the differences between savers with Commercial Banks/ Savings and Loan Institutions/ with both.	Discriminant Analysis	Bocio-economic variables (except age) were of little value in discriminating between groups. Differences exited between the groups in terms of their psychological characteristics. Savings motives were not important as discriminators.	Later studies indicated that both socio- economic characteristics and motivations were of significant importance in discriminating between different groups. The sample contained only holders of savings accounts.
Martin (1967) The USA	Searching for predictive factors related to savings inflow in S&L institutions.	Regression Analysis	The independent variables no. of competitors, consumers aged 65 and over and population growth were positively related to annual savings inflow. Reconmic and demographic variables of consumers were positively related to annual savings inflow.	The study did not investigate the influence of consumers' perceptions and motivations on annual savings inflow. The study did not explain the relationship between annual savings inflow and offices' area economic and demographic attributes.
Allaire (1972) Canada	Investigating the reasons behind the selection of a financial institution to deposit at it.	Percentages	Convenience in terms of location and time was the prime reason for the selection of a bank and received 48.8% of the rate of importance. Other variables such as higher rate of interest, security and familiarity with the bank received also high rates of importance.	The study lacks using statistical methods of data analysis to provide statistically tested results.